Technische Universität München

TUM School of Engineering and Design

### Assessment of the Added Value of Bidirectionally Chargeable Electric Vehicles for the User and the Energy System

Timo Kern

Vollständiger Abdruck der von der TUM School of Engineering and Design der Technischen Universität München zur Erlangung des akademischen Grades eines

### Doktors der Ingenieurwissenschaften (Dr.-Ing.)

genehmigten Dissertation.

Vorsitz:	Prof. Dr. rer. pol. Christoph Goebel
Prüfer der Dissertation:	1. Prof. DrIng. Ulrich Wagner
	2. Prof. Dr. Reinhard Haas

Die Dissertation wurde am 17.10.2022 bei der Technischen Universität München eingereicht und durch die TUM School of Engineering and Design am 29.01.2023 angenommen.

# Abstract

To limit the anthropogenic greenhouse gas effect, a variety of adjustments are required in the final energy sectors in addition to transforming the energy sector toward a climateneutral energy supply. For example, the integration of electric vehicles (EVs) enables the displacement of fossil fuels in the transport sector and leads to a stronger coupling of the transport and supply sectors. EVs can potentially provide flexibility as storage in this process through the new technology of bidirectional charging and discharging. Therefore, the objective of this cumulative dissertation is to quantify the added value of bidirectional EVs for both the user and the energy system. The developed modeling of bidirectional EVs has specificities depending on the point of view (system or user view) and the use case.

To evaluate the added value of bidirectional EVs from the user perspective, the optimization model eFlame is further developed to evaluate the use cases of photovoltaic (PV) self-consumption optimization and arbitrage trading via the spot markets. Due to the small power flows in the household, there is a requirement to model PV self-consumption optimization with variable charging and discharging efficiencies of the EV. Without this, revenue potentials are overestimated by 30%. For optimized trading in the spot market, consecutive trading in the day-ahead and intraday markets is modeled so that countertrading between the markets is possible. The modeled use cases show a high variability of revenue potentials depending on EV characteristics, user behavior, level and volatility of electricity prices, PV feed-in tariff, and numerous other influencing factors. Even taking into account the additional costs of bidirectional charging, however, both use cases become economical for an average user when investing in 2025.

For the evaluation of the added value of bidirectional charging from an energy system perspective, the energy system model ISAaR is further developed in order to be able to analyze two systemically oriented vehicle-to-grid use cases: Arbitrage trading via the spot markets, and congestion management provision. For this purpose, bidirectional EVs are modeled via aggregated profiles per market area or per grid node in the transmission grid in order to keep the complexity as low as possible. Despite the additional investment costs, numerous bidirectional EVs are integrated in the modeled, cost-optimal future energy system. These, in turn, lead to improved integration of PV energy and reduce the necessary capacities of both stationary storage and thermal power plants, significantly lowering the overall costs of the energy system in turn. The provision of congestion management in the transmission grid can also be partially provided by bidirectional EVs, reducing energy system costs and greenhouse gas emissions by displacing thermal power plants.

The use case of arbitrage trading via the spot markets can lead to added economic value for both the user and the energy system. Therefore, the regulatory framework should enable this use case. The other use cases considered can also be economical for either the users or the energy system, leading to the conclusion that the technology of bidirectional charging can be classified as an important part in the energy system of the future.

### Kurzfassung

Zur Einschränkung des anthropogenen Treibhausgaseffekts bedarf es zusätzlich zur Transformation des Energiesektors hin zu einer klimaneutralen Energieversorgung auch in den Endenergiesektoren vielfältiger Anpassungen. So ermöglicht die Integration von Elektrofahrzeugen (EVs) die Verdrängung von fossilen Brennstoffen im Verkehrssektor und führt zu einer stärkeren Kopplung zwischen Verkehrs- und Bereitstellungssektor. EVs können durch die neue Technologie des bidirektionalen Ladens und Entladens dabei potenziell als Speicher Flexibilität bereitstellen. Ziel dieser kumulativen Dissertation ist daher die Quantifizierung des Mehrwerts bidirektionaler EVs sowohl für den Nutzer als auch das Energiesystem. Die entwickelten Modellierungen bidirektionaler EVs weisen Spezifika in Abhängigkeit der Betrachtungsweise (System oder Akteurssicht) und des Use Cases auf.

Zur Bewertung des Mehrwerts bidirektionaler EVs aus Nutzersicht wird das Optimierungsmodell eFlame weiterentwickelt, um die Use Cases der Photovoltaik (PV)-Eigenverbrauchsoptimierung und des zeitlichen Arbitrage-Handels am Spotmarkt bewerten zu können. Aufgrund der kleinen Leistungsflüsse im Haushalt ergibt sich die Anforderung, die PV-Eigenverbrauchsoptimierung mit variablen Ladeund Entladewirkungsgrad des EVs zu modellieren, da sich ansonsten um 30 % überschätzte Erlöspotenziale ergeben. Für den optimierten Handel am Spotmarkt wird das konsekutive Handeln am Day-Ahead- und am Intraday-Markt modelliert, so dass ein Countertrading zwischen den Märkten möglich ist. Die modellierten Use Cases weisen eine hohe Variabilität der Erlöspotenziale in Abhängigkeit der EV-Charakteristik, des Nutzerverhaltens, des Niveaus und der Volatilität der Strompreise, der PV-Einspeisevergütung, sowie zahlreicher weiterer Einflussfaktoren auf. Auch unter Beachtung der zusätzlichen Kosten durch das bidirektionale Laden werden beide Use Cases aber für einen durchschnittlichen Nutzer bei einer Investition im Jahr 2025 wirtschaftlich.

Für die Bewertung des systemischen Mehrwerts des bidirektionalen Ladens wird das Energiesystemmodell ISAaR weiterentwickelt, um die Vehicle-to-Grid Use Cases zeitliche Arbitrage durch den Handel am Spotmarkt und Engpassmanagement- Bereitstellung aus Energiesystemsicht analysieren zu können. Bidirektionale EVs werden dafür über aggregierte Profile je Marktgebiet bzw. je Netzknoten im Übertragungsnetz modelliert, um die Komplexität möglichst gering zu halten. Trotz der zusätzlichen Investitionskosten werden im modellierten, kostenoptimalen zukünftigen Energiesystem zahlreiche bidirektionale EVs integriert. Diese führen wiederum zu einer verbesserten Integration der PV-Energie und zu verringerten notwendigen Kapazitäten von stationären Speichern sowie thermischen Kraftwerken, so dass die Gesamtkosten des Energiesystems signifikant sinken. Auch die Bereitstellung des Engpassmanagements im Übertragungsnetz kann teilweise durch bidirektionale EVs erbracht werden, wodurch Kosten und Treibhausgasemissionen des Energiesystems durch Verdrängung von thermischen Kraftwerken verringert werden.

Der Use Case zeitliche Arbitrage durch Handel am Spotmarkt kann sowohl auf Nutzer- als auch auf Energiesystemseite zu einem ökonomischen Mehrwert führen, so dass der regulatorische Rahmen diesen Use Case ermöglichen sollte. Aber auch die anderen betrachteten Use Cases können für die Nutzer oder das Energiesystem wirtschaftlich sein, so dass die Technologie des bidirektionalen Ladens als ein wichtiger Baustein im Energiesystem der Zukunft einzustufen ist.

### **Table of Contents**

ABSTRACT	III
KURZFASSUNG	V
TABLE OF CONTENTS	VII
ABBREVIATIONS	IX
LIST OF FIGURES	XI
LIST OF TABLES	XIII
LIST OF PUBLICATIONS	xv
1 INTRODUCTION	1
1.1 Literature Review and Research Gaps	2
1.2 Research Questions	4
1.3 Methodology	5
1.4 Dissertation and Publication Structure	7
2 BIDIRECTIONALLY CHARGEABLE ELECTRIC VEHICLES FROM THE USER PERSPECTIVE	11
2.1 Methods	11
2.1.1 Modeling of Charging and Discharging Efficiencies	12
2.1.2 Modeled Optimization Horizons	14
2.1.3 Use Case Dependent Mathematic Formulation	15
2.2 REVENUE POTENTIALS AND THEIR MOST IMPORTANT INFLUENCING FACTORS	17
2.2.1 Photovoltaic Self-Consumption Optimization	17
2.2.2 Arbitrage Trading	19
2.2.3 Combined Photovoltaic Self-Consumption Optimization and Arbitrage Trading	22
2.3 Profitability of Bidirectional Charging	23
2.3.1 Additional Costs of Bidirectional Charging	23
2.3.2 Evaluation of Profitability	25
3 BIDIRECTIONALLY CHARGEABLE ELECTRIC VEHICLES FROM THE ENERGY SYSTEM	20
PERSPECTIVE	29
3.1 Methods	29
3.1.1 Modeling of Electricity Market Integration	29
3.1.2 Modeling of Congestion Management Provision	32
3.2 SCENARIO DEFINITION	34 25
3.4 Impact on the Future Energy System	دد ۵۲
3.41 Other Assets in the Energy System	رد ۶۹
3.4.2 Total System Costs and Electricity Prices	41
3.4.3 Regional Residual Load Characteristics	42
3.4.4 Congestion Management	47
4 COMBINING THE USER AND THE ENERGY SYSTEM PERSPECTIVE ON BIDIRECTIONALLY	
CHARGEABLE ELECTRIC VEHICLES	49
4.1 COMPARISON OF MODELING APPROACHES DEPENDING ON USE CASE AND PERSPECTIVE	49
4.2 Possibility of a Common Added Value	52

5	С	ONCLUSION, DISCUSSION AND OUTLOOK	55
	5.1 5.2	Answers to Research Questions Critical Reflection, Concluding Comments and Further Research	. 55 . 58
6	В	IBLIOGRAPHY	61
7	Т	HESIS SUPERVISED BY THE AUTHOR	69
8	Ρ	UBLICATIONS OF THE AUTHOR	70
	Pubi Mar	lication 1 (Pub1): Integrating Bidirectionally Chargeable Electric Vehicles into the Electrici exets	тү 70
	Риві то-с Риві Futi Риві	LICATION 2 (PUB2): REVENUE OPPORTUNITIES BY INTEGRATING COMBINED VEHICLE-TO-HOME AND VEHIC GRID APPLICATIONS IN SMART HOMES LICATION 3 (PUB3): MODELING AND EVALUATING BIDIRECTIONALLY CHARGEABLE ELECTRIC VEHICLES IN JRE EUROPEAN ENERGY SYSTEM LICATION 4 (PUB4): ADDED VALUE OF PROVIDING TRANSMISSION GRID CONGESTION MANAGEMENT VIA	2LE- 108 THE 130 A
	RIDIE	RECTIONALLY CHARGEABLE ELECTRIC VEHICLES	152

# Abbreviations

BCM	Bidirectional Charging Management
Bidirectional EV	Bidirectionally Chargeable EV
EFC	Equivalent Full Cycles
eFlame	Electric Flexibility Assessment Modeling Environment
EV	Electric Vehicle
EVSE	Electric Vehicle Supply Equipment
GCP	Grid Connection Point
HP	Heat Pump
ISAaR	Integrated Simulation Model for Unit Dispatch and Expansion with Regionalization
LP	Linear Programming
MID	Mobility in Germany (Survey on Mobility)
MILP	Mixed Integer Linear Programming
NPV	Net Present Value
NUTS	Nomenclature of Territorial Units for Statistics
OH	Operating Hours
PTDF	Power Transfer Distribution Factors
PV	Photovoltaic
ResOpt	Residential Optimizer
RQ	Research Question
SBS	Stationary Battery Storage
Smart EV	Smart unidirectional charging EVs
SoC	State of Charge
V2B	Vehicle-to-Business
V2G	Vehicle-to-Grid
V2H	Vehicle-to-Home

# List of Figures

Figure 1-1:	Research questions (RQ) at user, energy system and combined perspective	5
Figure 1-2:	Methodology for assessing the added value of bidirectional EVs for the	
	user and the energy system	7
Figure 1-3:	Structure and publications of this dissertation assigned to research	
	questions at user, energy system and combined perspective	9
Figure 2-1:	Schematic representation of eFlame and ResOpt for evaluation of use	
	cases of bidirectional EVs based on [Pub2]	12
Figure 2-2:	(a): Comparison of real and modeled charging losses depending on the	
	AC charging power; (b): Annual duration curve of residual load of five	
	households (hh1 to hh5)	13
Figure 2-3:	Schematic representation of the rolling optimization runs with limited	
	foresight of different market prices based on [Pub1]	14
Figure 2-4:	(a): Revenues through PV self-consumption optimization of smart and	
	bidirectional EVs depending on household configuration and user type as	
	presented in [Pub2]; (b): PV self-consumption depending on EV charging	10
<b>Figure 2 F</b>	strategy, household configuration, and user type as presented in [Pub2]	18
Figure 2-5:	Impact of most important influencing factors on revenues of PV self-	
	based on [Pub2]	10
Figure 2 6:	Daseu OII [PUD2] Povonuo potontials of arbitrago trading in different electricity markets for	19
Figure 2-0.	different EV and user types in the year 2019 in Cormany as presented in	
		20
Figure 2-7	[rubi] Revenue potentials depending on the regulatory framework (a) and on the	20
	electricity price volatility (b) based on [Pub1]	22
Figure 2-8 <sup>.</sup>	Daily share of households using V2H and V2G (bottom figure) depending	
	on weekly standard deviation of electricity prices and weekly PV	
	generation (upper figure) based on [Pub2]	23
Figure 2-9:	Profitability of V2H PV self-consumption optimization depending on the	
5	investment year of bidirectional EV and EVSE	26
Figure 2-10:	Profitability of V2G arbitrage trading depending on the investment year of	
3	bidirectional EV and EVSE	28
Figure 3-1:	Different availability of aggregated and discrete EV profiles for an	
	exemplary week as presented in [Pub3]	32
Figure 3-2:	Schematic modeling of the restricted storage state of charge at the start	
	and the end of a time slice as presented in [Pub4]	34
Figure 3-3:	Number of EVs per charging strategy for scenarios BCM, Sen1 and Sen2	
	based on [Pub3] and on further evaluations	36
Figure 3-4:	Dependance of shares and numbers of EVs per charging strategy and	
	country for the BCM (a) and Sen2 (b) scenarios in 2050 on the full load	
	hours of PV generation based on [Pub3] and on further evaluations	37
Figure 3-5:	Dependance of shares of bidirectional EVs in total EVs on various	
	influencing factors	38
Figure 3-6:	Installed capacities of variable renewable energies (a) and thermal power	
	plants (b) for the scenarios Ref, BCM and Sen2 for the years 2030, 2040	
	and 2050 in Europe based on [Pub3] and on further evaluations	40

Figure 3-7:	Installed capacities of mobile (a) and stationary (b) storages for the scenarios Ref. BCM and Sen2 for the years 2030, 2040 and 2050 in Europe	
	based on [Pub3] and on further evaluations	40
Figure 3-8:	Annual hourly duration curves of European residual load for the years	
	2030, 2040 and 2050 for both scenarios Ref and BCM and the scenario	
	BCM without bidirectional EVs	43
Figure 3-9:	Factors indicating change in peak load (a) and generation surpluses (b)	
	per NUTS 3 region in Europe in 2030 in scenario BCM compared to Ref	45
Figure 3-10:	Factors indicating change in grid dimensioning per NUTS 3 region in	
	Europe in 2030 (a) and 2050 (b) in scenario BCM compared to Ref	46
Figure 3-11:	Negative (a) and positive (b) congestion management per grid node split	
	up in different technologies for the BCM grid run	48

# List of Tables

Table 2-1:	Additional investment cost range for V2H, and V2G in $\epsilon_{2021}$ /EV compared	
	to unmanaged charging based on [43]	24
Table 2-2:	Additional yearly cost range for V2H, and V2G in $\epsilon_{2021}$ /EV/a compared to	
	unmanaged charging based on [43]	24
Table 2-3:	V2H Revenues based on household type and PV feed-in tariff for the years	
	2020, 2025 and 2030	25
Table 2-4:	V2G revenues based on regulatory framework, electricity price	
	characteristic, as well as EV and EVSE parameterization for the years 2020	
	to 2040	27
Table 3-1:	Modeling approaches of bidirectional EVs in a large-scale energy system	
	model and their advantages and disadvantages	31
Table 3-2:	Technical parameters of bidirectional and smart EVs	35
Table 3-3:	Additional investment costs of bidirectional and smart EVs compared to	
	unmanaged charging EVs for base and sensitivity (SenX) scenarios	35
Table 3-4:	Total energy system costs and mean European electricity price for the	
	years 2025 to 2050 for scenarios Ref, BCM and Sen2	42
Table 4-1:	Comparison of modeling approaches of bidirectional EVs presented in this	
	dissertation	51
Table 4-2:	Main added values for the user and the energy system by considered use	
	cases in this dissertation	53

# **List of Publications**

In this cumulative dissertation, based on four publications, the added value of bidirectionally chargeable electric vehicles for the user and the energy system is assessed. The four publications are briefly summarized on the following pages. The complete publications can be found in the appendix of this dissertation. In the main part of this dissertation (Chapters 2 to 5), the publications are linked together and supplemented with further analysis, methodology, and elaboration.

In science, and thus also at the Research Center for Energy Economics (FfE), publications by only one author are the exception, since work is often done on complex models with a wide variety of interfaces. Therefore, all persons involved in the implementation of the described model and the processing of the input data used are listed as authors on publications. However, in all publications listed below, the first author developed the basic concept as well as the methodological approach to answer the respective research questions, and the first author was responsible for the interpretation of the model results.

This cumulative dissertation is based on the following publications:

- Kern, T., Dossow, P., von Roon, S. Integrating Bidirectionally Chargeable Electric Vehicles into the Electricity Markets. Energies 2020, 13, 5812. https://doi.org/10.3390/en13215812
- Kern, T., Dossow, P., & Morlock, E. (2022). Revenue opportunities by integrating combined vehicle-to-home and vehicle-to-grid applications in smart homes. Applied Energy, 307. https://doi.org/10.1016/j.apenergy.2021.118187
- **Kern, T.,** Kigle, S. (2022). Modeling and Evaluating Bidirectionally Chargeable Electric Vehicles in the Future European Energy System. Energy Reports. (Accepted, not yet published)
- Kern, T., Wendlinger, C. (2022). Added Value of Providing Transmission Grid Congestion Management via Bidirectionally Chargeable Electric Vehicles. 2022 18th International Conference on the European Energy Market (EEM), 2022. (Accepted, not yet published)

The following four pages present a summary of the publications. In Chapter 8 the complete publications are attached. The detailed contributions of the first author are listed in the Section 'Own contribution'. The citation keys [Pub1] to [Pub4] refer to the publications.

**Publication 1 (Pub1):** Integrating Bidirectionally Chargeable Electric Vehicles into the Electricity Markets

**Summary:** The energy system transformation entails structural changes in all sectors. While the electricity supply sector in Germany has already been subjected to massive adjustments due to the expansion of renewable energies, the final energy sector of transport has been very slow to switch to more climate-friendly technologies. Replacing traditional internal combustion engine vehicles with electric vehicles (EVs) proves to be challenging for the transport sector, particularly due to the higher initial investment costs. Since EVs could be more profitable by participating in the electricity markets, the aim of this paper is to investigate revenue potentials when marketing bidirectionally chargeable EVs in the spot market.

To simulate a realistic marketing behavior of electric vehicles, a mixed integer linear, rolling horizon optimization model is formulated considering real trading times in the day-ahead and intraday market. The objective of the optimization problem is the charging at minimum costs while discharging at maximum costs. A single day is separated into eight time steps, in each of which an optimization of the following 48-72 hours is performed to enable an optimal adjustment of the schedule based on the current traded electricity market prices. By considering consecutive trading possibilities in the day-ahead market and the intraday trading, trades can be counter traded to generate revenues by price differences in the day-ahead and intraday markets. In addition to the in-depth modeling of the trading in electricity markets, a detailed modeling of the electric vehicle and vehicle user is also introduced. The state of charge (SoC) of the EV is constrained by a minimum SoC of the battery and a minimum SoC at departure of the EV that can be parameterized by the EV user. The developed optimization model allows a flexible adjustment of the scenarios, whereby a change of important influencing factors can be analyzed.

Results suggest that revenue potentials are strongly dependent on the EV pool, the user behavior, and the regulatory framework. Modeled potential revenues of EVs of current average size marketed with 2019 German day-ahead prices are found to be at around 200  $\notin$ /EV/a, which is comparable to other findings in literature, and go up to 500  $\notin$ /EV/a for consecutive trading in German day-ahead and intraday markets. For future EVs with larger batteries and higher efficiencies, potential revenues for current market prices can reach up to 1300  $\notin$ /EV/a. This study finds that revenues differ widely for different European countries and future perspectives. The identified revenues give EV owners a clear incentive to participate in vehicle-to-grid use cases, thereby increasing much needed flexibility for the energy system of the future.

**Own contribution:** Conceptualization, Data curation, Formal analysis, Investigation, Methodology, Software, Validation, Visualization, Writing - original draft and review & editing.

**Citation:** Kern, T.; Dossow, P.; von Roon, S. Integrating Bidirectionally Chargeable Electric Vehicles into the Electricity Markets. Energies 2020, 13, 5812. https://doi.org/10.3390/en13215812

**Publication 2 (Pub2):** Revenue opportunities by integrating combined vehicle-to-home and vehicle-to-grid applications in smart homes

**Summary:** A smart integration of electric vehicles (EVs) in the future energy system will be crucial in decarbonizing the energy sector. A survey by the German Association of Energy and Water Industries found that the high investment costs for an EV are the main argument against switching to electromobility. If the economic viability of EVs could be increased, it would provide an additional incentive for users to purchase an EV. Bidirectional EVs can provide flexibility for the system and generate revenues for the user through multiple use cases. Vehicle-to-home (V2H) use cases offer the benefit of optimized use of locally generated renewable energy while also providing revenue opportunities. In vehicle-to-grid (V2G) applications, bidirectionally chargeable EVs can contribute to grid stability and market liquidity in a system serving manner while offering economic benefits for EV owners.

V2H use cases are modeled with and without varying charging and discharging efficiencies. This novel evaluation of more realistic modeling of V2H as mixed-integer linear programming (MILP) with varying efficiencies versus modeling of V2H as linear programming (LP) with fixed efficiencies is necessary to assess whether MILP with varying efficiencies is beneficial or even necessary for V2H analyses. For this purpose, an optimization model is developed that optimizes the electric power flows of a household with the objective of minimizing electricity costs while taking technical restrictions into account. Three different charging strategies are compared regarding household energy flows: an unmanaged charging strategy, a smart charging strategy and a bidirectional charging strategy. Finally, a combined application of V2H and V2G arbitrage trading is modeled to show differences and benefits on the electricity costs of a household. This combined, novel modeling of V2H and V2G can realistically combine the seasonally different revenue opportunities of the use cases and shows that a separate modeling of a V2H or V2G use case can underestimate the revenue opportunities.

For a typical German household using a bidirectional EV for optimizing PV selfconsumption, revenues for V2H exclusively are about 310  $\notin$ /a, mostly generated during the summer. The revenue potentials are highly dependent on the influencing factors of PV size and PV feed-in tariff, household size, as well as household components, like heat pumps or stationary storages. Arbitrage trading well complements this V2H use case in the winter months, resulting in revenues up to 530  $\notin$ /a. These significant revenue potentials can lead to more profitable and interactive EVs incentivizing users to change from internal combustion vehicles to electric mobility.

**Own contribution:** Conceptualization, Methodology, Investigation, Formal analysis, Software, Validation, Data curation, Writing – original draft, Writing – review & editing, Supervision.

**Citation:** Kern, T., Dossow, P., & Morlock, E. (2022). Revenue opportunities by integrating combined vehicle-to-home and vehicle-to-grid applications in smart homes. Applied Energy, 307. https://doi.org/10.1016/j.apenergy.2021.118187.

**Publication 3 (Pub3):** Modeling and Evaluating Bidirectionally Chargeable Electric Vehicles in the Future European Energy System

**Summary:** In addition to a massive expansion of renewable energies, a successful change towards a decarbonized energy system requires the flexibilization of consumers and the integration of storage and sector coupling technologies. Bidirectionally chargeable electric vehicles (EVs) represent such a consumer flexibility. They are able to charge when there is an electricity generation surplus and to discharge when there is a shortage in electricity generation. Therefore, they can act as a storage from the perspective of the energy system. This paper analyzes different modeling approaches of bidirectionally chargeable EVs in large-scale energy systems and evaluates the impact of bidirectionally chargeable EVs on the future European energy system.

The modeling of discrete EV profiles as a reference is compared to the modeling of clustered EV profiles as well as an aggregated EV profile with simplified constraints for bidirectionally chargeable EVs. The approach of clustered EV profiles limits the number of discrete EV profiles to lower the computation times. This approach results in EV profiles that do not represent the EV characteristics well. Aggregation of EV profiles per country leads to significantly lower computation times, while still representing EV behavior well. This approach also leads to energy system results, e.g., power plant investment and operation, that are close to the reference case, although the EV constraints are simplified.

The number of bidirectionally chargeable EVs in a cost optimal future European energy system increases from 6 million EVs in 2025 to over 60 million EVs in 2050. We show that bidirectionally chargeable EVs support the integration of Photovoltaic generation. This is of great importance for the future energy system, since wind generation, especially onshore wind generation, deals with various acceptance issues across Europe. Furthermore, bidirectionally chargeable EVs lower installed capacities of gas- and hydrogen-fired power plants as well as stationary battery storages. Replacing the need for power plant capacities that operate with very few full load hours through bidirectionally chargeable EVs means contributing to covering the peak load in the energy system. Bidirectionally chargeable EVs also lead to decreasing electricity prices and total European energy system costs that are up to 9 billion €/a lower compared to the reference scenario.

**Own contribution:** Conceptualization, Methodology, Investigation, Formal analysis, Software, Validation, Data curation, Writing – original draft, Writing – review & editing, Supervision.

**Citation:** Kern, T., Kigle, S. (2022). Modeling and Evaluating Bidirectionally Chargeable Electric Vehicles in the Future European Energy System. Energy Reports. (Accepted, not yet published)

**Publication 4 (Pub4):** Added Value of Providing Transmission Grid Congestion Management via Bidirectionally Chargeable Electric Vehicles

**Summary:** Congestion management in the European transmission grid is currently mostly provided by conventional power plants. As electricity generation in the future will be increasingly characterized by volatile renewable energies, new technologies need to be integrated into congestion management provision. This paper develops a methodology for modeling storages in congestion management and investigates the potential for bidirectionally chargeable electric vehicles (bidirectional EVs) in 2030.

Congestion management in the energy system model ISAaR (Integrated Simulation Model for Unit Dispatch and Expansion with Regionalization) is modeled in a two-step approach. First, a multi-energy market optimization run is conducted, in which one market area is represented by one node. Energy transfers between market areas are allowed up to maximum net transfer capacities. In the following step, the resulting dispatch of generation, load and storage instances is taken as a base for the transmission grid optimization run, in which the costs of congestion management (curtailment of renewable energies and provision of redispatch services) are minimized. A direct current load flow via Power Transfer Distribution Factors (PTDF) is used to model the load flow in the transmission grid. Since the modeling involves great complexity due to time-coupling constraints of storages, different modeling approaches are investigated. The length of the time slice that is optimized coupled is set to 168 hours so as not to overly constrain the flexibility of bidirectional EVs while maintaining computability.

The results show that bidirectional EVs can take over a significant part of the congestion management due to their decentralized distribution. A total of 26 TWh of positive congestion management of conventional power plants in 2030 can be replaced by the optimized use of bidirectional EVs. This leads to a reduction of emissions of 12 million tons CO<sub>2</sub>. The high absolute volumes of congestion management that appear in the simulations are classified and possible reasons for the deviation from current volumes are elaborated.

**Own contribution:** Conceptualization, Methodology, Investigation, Formal analysis, Validation, Data curation, Writing – original draft, Writing – review & editing, Supervision.

**Citation:** Kern, T., Wendlinger, C. (2022). Added Value of Providing Transmission Grid Congestion Management via Bidirectionally Chargeable Electric Vehicles. 2022 18th International Conference on the European Energy Market (EEM), 2022. (Accepted, not yet published)

### 1 Introduction

To minimize the anthropogenic greenhouse effect and thus mitigate climate change, the EU and Germany have set themselves ambitious climate targets. At the European level, the energy system is to be completely climate-neutral by 2050 [1]; in Germany, the aim is to decarbonize the energy system by 2045 [2]. The turnaround in energy policy that has been initiated is leading to challenges at a wide variety of levels. In the final energy sectors of private households, the tertiary sector, industry and transport, applications and processes are being adapted through efficiency measures, fuel substitutions, flexibilization and electrification [3]. The transport sector in particular has been slow to reduce greenhouse gases in recent years. In Germany, in 2021 the emissions of just under 150 million t CO<sub>2</sub> increased compared to in 2020 and were above the emissions permitted by the Climate Action Law [4]. However, with various subsidies for electric vehicles (EVs) [5] and the increase in fuel and CO<sub>2</sub> prices as a result of the Covid-19 crisis [6], 2020 and 2021 saw a dynamic increase in EV registrations [7].

In the energy sector, the massive expansion of variable renewable energies and the associated substitution of conventional nuclear, coal and gas-fired power plants will lead to a completely new characteristic of electricity supply. Due to the volatility of the supply from wind and sun, electricity from these renewable energies will not be generated in line with the static demand from the final energy sectors in the future. Flexibilities in the energy system are therefore needed to ensure that the balance between electricity generation and consumption is maintained [3].

Currently, almost all EVs in Germany are charged unmanaged, although there are numerous projects working on smart charging of EVs based on grid status, electricity prices or electricity generation of renewable energies [8]. Bidirectionally chargeable electric vehicles (bidirectional EVs) represent a new technology that can address major challenges in the transportation final energy sector and the energy sector alike by a stronger coupling of these sectors. Bidirectional EVs can not only charge electricity, but also discharge via bidirectional electric vehicle supply equipment (EVSE) [9]. This enables a variety of use cases that can bring potential added value for the user and the energy system. If the user receives an ecological or economic added value, this also promotes the integration of EVs, as these then become more profitable in terms of overall costs.

This cumulative dissertation therefore evaluates the added value of bidirectional EVs for the user and the energy system. In this introduction, Section 1.1 first identifies the current state of research and existing research gaps. From this, the research questions of this dissertation are derived in Section 1.2. Based on this, Section 1.3 presents the methodology to answer these research questions. Finally, the structure of this cumulative dissertation and classification of the publications is added in Section 1.4.

#### 1.1 Literature Review and Research Gaps

Bidirectional EVs can execute a wide variety of energy-related use cases, leading to increased coupling of the transportation sector with the energy sector [10]. In the 'Bidirectional Charging Management' (BCM) project, numerous use cases of bidirectional EVs were defined, which were divided into the groups Vehicle-to-Home (V2H), Vehicle-to-Grid (V2G) and Vehicle-to-Business (V2B) [11]. V2H includes the photovoltaic (PV) self-consumption optimization and the electricity tariff optimized charging and discharging. V2G implies arbitrage trading in the day-ahead and intraday market, CO<sub>2</sub>-optimized charging and discharging as well as different distribution and transmission grid-oriented use cases. V2B mainly deals with peak load shaving in companies in order to save grid fees. This dissertation focuses on the use cases PV self-consumption optimization, arbitrage trading and the provision of the ancillary service congestion management as a transmission grid-oriented use case. These selected use cases are prioritized in the BCM project, implemented in the field test, and show promising values from a user and system perspective. The main region of analysis from the user perspective is Germany, whereas the systemic analysis focuses on the European coupled energy system. However, the developed models can also be applied to other regions worldwide by adapting the parameters. In this section, this dissertation is classified within the existing research based on relevant studies, upon which the research questions are posed in the following section.

The economic and ecological added values from the user's perspective on bidirectional EVs are of great relevance because ultimately the user makes the investment decision in a bidirectional EV and EVSE. Numerous studies model smart unidirectional charging EVs (smart EVs) and bidirectional EVs dealing with their revenue potentials for arbitrage trading. While investigations in studies of Illing et al. [12] and Shafiullah et al. [13] are still limited to unidirectional smart charging, the work of Bessa et al. [14], Rominger et al. [15], Schmidt et al. [16], Peterson et al. [17], and Pelzer et al. [18] model bidirectional EVs participating in the electricity spot markets. These studies show revenue potentials for bidirectional EVs in different spot markets worldwide, but do not focus on the modeling of a realistic user behavior and its impact on the revenues, and furthermore only model the trading on single electricity markets in a simplified way without the possibility of consecutive trading in different spot markets. PV self-consumption optimization has also been the subject of numerous studies. Publications of Salpakari et al. [19], Chen et al. [20], Erdinc et al. [21], Kataoka et al. [22], Wickert et al. [23] and Keiner et al. [24] model V2H with bidirectional EVs in a household and examine their revenue potentials in the countries Sweden, China, Portugal, Japan, and Germany. When comparing the studies, it becomes clear that the regulatory framework, such as the composition of the household electricity price, is a key influencing factor. Again, in these studies, user behavior is only inaccurately modeled, so that the effects of different parameterizations cannot be highlighted. In addition, a fixed charging and discharging efficiency of the EV is always assumed in order to keep the optimization problem linear. Englberger et al. [25] also already deal with a combination of different use cases for bidirectional EVs by segmenting the battery for frontof-the-meter and behind-the-meter applications. Behind-the-meter applications refer to

use cases that happen on the energy user's side of the meter. Front-of-the-meter applications represent use cases that interact with the electricity grid off the energy user's side of the meter. However, this interesting modeling approach of Englberger et al. seems hardly feasible in terms of measurement technology, so that the need for more practice-oriented modeling arises here.

While V2B and V2H optimize energy flows behind the meter to enable, for example, savings in electricity costs, V2G applications are designed to meet the needs of the energy system. By integrating bidirectional EVs into the electricity market, studies of Hanemann et al. [26], Rodríguez et al. [27], and Huang et al. [28] describe a smoothing effect on electricity demand that also smoothes resulting electricity prices [26]. The publication of Wie et al. [10] also shows a positive ecological and economic effect of bidirectional EVs on integrated multi-energy systems. However, these demonstrated positive effects of bidirectional EVs refer to small- and medium-scale energy systems and are shown with a highly simplified modeling of the EVs. Furthermore, hardly any effects on other flexibilities, such as stationary battery storages and sector-coupling elements, are investigated. The work of Child et al. [29] analyzes this feedback of bidirectional EVs on the energy system, by indicating that reduced capacities of generation and storage are needed, but again only for a small-scale energy system, the Åland Islands near Finland. The integration of bidirectional EVs into a large European energy system model implies the challenge of no longer modeling each EV discretely, but rather performing such simplifications as clustering or aggregation of EV profiles.

This challenge of realistic simplifications arises even more when modeling the provision of congestion management (curtailment of renewable energies and redispatch services). Redispatch services refer to short-term adjustments of the schedule of generation or consumption assets to lower congestions in the transmission grid [30]. The integration of bidirectional EVs or other storage assets leads to much more complex simulations of the European transmission grid. For example, the work of Böing [31] excludes storages in the congestion management run because of their time-coupled constraints, which introduces great complexity and considerable modeling challenges. Kotzur et. al analyze possibilities to handle the complexity of time coupling in energy system models, like temporal decomposition of the optimization problem, but point out that these possibilities are usually associated with complex implementations and numerous assumptions [66]. However, in the literature, some studies already deal with the integration of storages in the provision of ancillary services. The publications of Meyer-Huebner et al. [32], Xiong et al. [33], Eickmann et al. [34], Müller et al. [35] and Gutermuth et al. [36] model stationary storages or other time-coupling elements, such as power-to-gas units with virtual gas storages, providing redispatch services. These studies focus only on stationary and not mobile storages and limit the optimization to small-scale test systems, only few timesteps or to very simplified EV modeling. The integration of bidirectional EVs into the redispatch process is modeled in the work of Staudt et al. [37] and Thormann et al. [38] but with simplified heuristic modeling or without considering the operation of EVs and the system needs of redispatch services. None of the existing approaches in the literature explore the transferability of their modeling approaches to a large-scale European energy system,

which causes a very different complexity. Thus, the added values of bidirectional EVs can only be determined inadequately from a system perspective.

In general, none of the studies mentioned in the previous two paragraphs address the question of how high the share of bidirectional EVs in the total EVs is optimal in a future multi energy system. This requires the more complex modeling of bidirectional EVs as an element that can be added endogenously optimized by the energy system model. However, answering this question is of great importance so that stakeholders in the field of energy can design their business models, policy makers their support mechanisms, and scientists their energy system models accordingly. Furthermore, the question arises to what extent bidirectional EVs can bring a combined added value from the user and energy system perspective, which this dissertation aims to answer.

### 1.2 Research Questions

The previously analyzed research gaps are transferred into the research questions of this dissertation. The definition of the research questions (RQ) is divided into three sections as shown in Figure 1-1.

The first section of the research questions refers to the added value of bidirectional EVs for the user:

- 1. What modeling specifications are required for the evaluation of revenue potentials of vehicle-to-home and vehicle-to-grid use cases?
- 2. What are the revenue potentials and their most important influencing factors for vehicle-to-home and vehicle-to-grid use cases?
- 3. What is the economic profitability of vehicle-to-home and vehicle-to-grid use cases considering the additional investment costs?

The second section of the research questions deals with the added value of bidirectional EVs for the energy system:

- 4. How do modeling specifications of bidirectional EVs need to be adapted to evaluate the European energy system perspective compared to modeling discrete EVs?
- 5. How high is the optimal share of bidirectional EVs in the total EVs from a system cost perspective?
- 6. What is the added value of bidirectional EVs integrated in the electricity markets and in congestion management in the transmission grid from an energy system perspective?

The third section of the research questions brings together the previous sections and discusses the possibility of a combined added value for the user and the energy system.

7. How can use cases of bidirectional EVs create combined added value from the user and the energy system perspective?

Figure 1-1 depicts the relationships between the research questions in the three sections and shows the sequential structure of this dissertation. It can be observed that the research questions within the sections are not considered completely separately but build on each other. The analysis of the revenue potentials and profitability of vehicle-to-grid and vehicle-to-home use cases from the user perspective in RQ2 and RQ3 depend on the approach to modeling bidirectional EVs in RQ1. The assessment of the systemic cost-optimal share of bidirectional EVs in total EVs and its impact on the energy system in RQ5 and RQ6 depend on the developed modeling in RQ4. RQ7, in turn, builds on all the previously named RQs. There are also interdependencies between the analyses from the user perspective and from the system perspective. The detailed modeling to answer the research questions is described in the following section.



Figure 1-1: Research questions (RQ) at user, energy system and combined perspective

#### 1.3 Methodology

Based on the research questions established in the previous section, Figure 1-2 presents the methodology of this dissertation. The methodology is divided into the sub-areas of bidirectional charging from the user's point of view and from the energy system's point of view.

First, the eFlame model (electric Flexibility Assessment Modeling Environment) with the optimization module ResOpt (Residential Optimizer) [Pub2] is further developed by a methodology for modeling V2H PV self-consumption optimization and V2G arbitrage trading from the user's perspective. For V2H, minimization of household electricity costs is modeled to allow evaluation of the revenue potentials of bidirectional EVs. The charging and discharging efficiencies are modeled to be variable, so that the revenue potentials of

bidirectional EVs are not overestimated. For V2G arbitrage trading, minimizing the charging costs while maximizing the revenue from discharging an EV is implemented. For this purpose, a rolling optimization is developed that allows trading in all consecutive spot markets. Countertrades can be used to take advantage of price differences in the consecutive spot markets. The next step merges V2H and V2G modeling to evaluate a combination option. Based on various sensitivity calculations, the revenue potentials and their most important influencing factors of the use cases are then analyzed. By evaluating the additional investment costs of bidirectional charging, the economic profitability of the different use cases can be assessed in a final step.

Second, the energy system model ISAaR (Integrated Simulation Model for Unit Dispatch and Expansion with Regionalization) [39] is further developed by a methodology for modeling bidirectional EVs in the energy system and in congestion management provision in the transmission grid. To evaluate V2G arbitrage trading from an energy system perspective, a minimization of the total economic costs of the European energy system is performed. Different approaches to model bidirectional EVs are investigated in order to keep the complexity of the model as low as possible while representing bidirectional EVs realistically. Based on a scenario for the future European energy system, cost-optimal shares of bidirectional EVs in the total EVs are then determined for future years in the different European countries. For the evaluation of V2G congestion management provision, a minimization of congestion management costs in the European energy system is performed considering bidirectional EVs. For this purpose, different time slice lengths of the rolling optimization are analyzed, since the optimization problem has a very high complexity due to the time coupling constraints of bidirectional EVs. Finally, the impact of bidirectional EVs on the future energy system and the provision of congestion management in the transmission grid is assessed. This includes, for example, the impact of bidirectional EVs on the capacities of thermal power plants, renewable energies, and stationary battery storages as well as the impact on the total system costs.

The methodology considers the interdependencies between the user and the energy system side through user behavior constraints for discrete EVs, mean investment costs for bidirectional charging, and future electricity prices. User behavior constraints are first analyzed for discrete EVs in use case arbitrage trading and then, in a simplified way, also modeled in the system perspective. The future mean investment costs of bidirectional EVs are passed to the energy system model ISAaR. ISAaR decides endogenously within the optimization, considering the additional costs of bidirectional charging, how high the cost-optimal share of bidirectional EVs is for the future European energy system model to determine future revenue potentials for various users.

In a consolidating step, the modeling approaches of bidirectional EVs from the user and energy system perspective are compared and evaluated. Finally, the highlighted added values of bidirectional EVs are summarized and a combination of added values from user and energy system perspectives is discussed.



Figure 1-2: Methodology for assessing the added value of bidirectional EVs for the user and the energy system

#### 1.4 Dissertation and Publication Structure

The structure of this dissertation builds on the previous sections. Chapter 2 evaluates bidirectional charging from the user's perspective with regard to various influencing factors. Section 2.1 starts with the modeling developed for V2H and V2G use cases. Based on this, revenue potentials are determined for different user types in Section 2.2 with the highlighting of the revenues' most important influencing factors. Section 2.3 compares the revenue potentials to the additional investment costs for the technology of bidirectional charging in order to ultimately be able to evaluate the economic profitability for the user. Chapter 3 switches the perspective on bidirectional EVs from the user to the energy system.

Consequently, the focus here is on V2G use cases that interact with the energy system. Section 3.1 presents the approaches to model bidirectional EVs integrated in the electricity markets and providing congestion management. Following the scenario definition in Section 3.2, Section 3.3 assesses the cost-optimal penetration rates of bidirectional EVs in the future energy system. Finally, Section 3.4 analyzes the impact of bidirectional EVs on the costs and characteristics of the future energy system.

Chapter 4 then brings together the findings of the previous chapters and assesses bidirectional EVs combined from a user perspective and from an energy system perspective. To this end, Section 4.1 first compares the different modeling approaches and highlights

the differences in modeling the user and system views. Section 4.2 analyzes which use cases combine well and when to obtain added value for the EV user and the energy system.

Finally, Chapter 5 concludes by answering the research questions raised, discussing the key findings, and summarizing them. Based on this, an Outlook identifies the need for further research.

Figure 1-3 summarizes the structure of this dissertation and assigns the research questions to the publications carried out. Each publication addresses several research questions. Publications 1 and 2 investigate bidirectional EVs from the user's perspective and answer the research questions RQ1 and RQ2 on modeling, economic efficiency, and the most important influencing factors for V2H and V2G use cases, respectively. The main part of this dissertation, considering the investment costs of bidirectional charging, complements these analyses by evaluating the profitability of bidirectional EVs today and in the future. Publications 3 and 4 evaluate the energy system view on bidirectional EVs, answering RQ4 to RQ6. Publication 3 evaluates the effect of bidirectional EVs on the electricity market and other components in the future energy system. Publication 4 identifies the potential of bidirectional EVs to provide congestion management in the transmission grid and thus prevent renewable energy curtailment and reduce the operation of thermal power plants. The main part of this dissertation complements these publications by examining two sensitivities with greatly reduced investment costs of smart EVs to draw conclusions about the robustness of the results. Furthermore, additional investigations on the impact of bidirectional EVs on the regional residual load (load minus electricity generation of variable renewable energies) show indications for the grid loads in the electricity grid.

Further, the main part of this dissertation addresses and answers RQ7 by summarizing and comparing the findings of the Chapters 2 and 3.



Figure 1-3: Structure and publications of this dissertation assigned to research questions at user, energy system and combined perspective

# 2 Bidirectionally Chargeable Electric Vehicles from the User Perspective

Based on the publications [Pub1] and [Pub2] and further research, this chapter analyzes the profitability of PV self-consumption optimization (V2H) and arbitrage trading (V2G) by bidirectional EVs. For this purpose, Section 2.1 first addresses the use case dependent requirements of the modeling, upon which the revenue potentials and most important influencing factors of the use cases are determined in Section 2.2. Section 2.3 then presents the economic profitability of bidirectional EVs by including the additional costs of bidirectional charging.

### 2.1 Methods

For the detailed evaluation of bidirectional EVs from the user's point of view, the model environment eFlame with the optimization module ResOpt has been further developed. The model environment eFlame allows a variable scenario creation for different use cases of bidirectional EVs or other energy assets, like stationary battery storages (SBS). In ResOpt, the mathematical optimization problem is formulated depending on the parameterization made in eFlame. The optimization problem can be formulated as linear programming (LP) or mixed integer linear programming (MILP). ResOpt is a module that is also integrated into other models, such as the distribution grid model GridSim of FfE Munich [40].

Figure 2-1 represents the structure of eFlame including ResOpt with the main decision and input variables. The model environment eFlame formulates scenarios in the database, which are read into Matlab and passed to ResOpt. The variable scenario definition in eFlame allows the calculation of numerous sensitivities, whereby the most important influencing factors of use cases can be determined. For each scenario created, the EV is modeled with three different charging strategies:

- **unmanaged charging:** the EV charges directly after arriving at a charging location.
- **smart charging:** the EV can postpone charging at a charging location within its flexibility limits.
- **bidirectional charging:** the EV can charge and discharge at a charging location at any time within its flexibility limits.

The differential electricity costs of the charging strategies can be used to determine revenues of a bidirectional or smart EV versus an unmanaged charging EV.

Depending on the energy assets modeled in a use case, ResOpt introduces different decision variables: EV charging  $P_{EV,c'}$  EV discharging  $P_{EV,d}$  and potentially standby losses of the EV and EVSE  $P_{EV,l,s'}$  SBS charging  $P_{SBS,c}$  and discharging  $P_{SBS,d'}$  PV curtailment  $P_{PV,curt}$ , heat pump (HP) demand  $P_{HP,el}$ , as well as power from grid  $P_{GCP,in}$  and

power to grid  $P_{GCP,out}$ . In addition, public charging energy  $E_{EV,pub,c}$  and driving energy  $E_{EV,drive}$  of the EV are considered as fixed input. For the combined modeling of V2H and V2G use cases, the decision variables  $P_{GCP,in,v2g}$  and  $P_{GCP,out,v2g}$  are integrated. A fixed thermal  $P_{HH,th}$  and electrical  $P_{HH,el}$  household demand, an electrical industry demand  $P_{Ind,el}$  as well as a fixed PV generation  $P_{PV}$  can also be provided, whereas industrial demand is not used in this dissertation. For the modeling of trading in consecutive spot markets, the decision variables  $P_{EV,cou,buy}$  and  $P_{EV,cou,sell}$  that represent countertrades are integrated.  $P_{sched,AC}$  represents scheduled powers from previous optimization runs. A more detailed description of eFlame and integrated model ResOpt can be found in [Pub1] and [Pub2].



Figure 2-1: Schematic representation of eFlame and ResOpt for evaluation of use cases of bidirectional EVs based on [Pub2]

The use cases PV self-consumption optimization and arbitrage trading and a combined modeling of these use cases have different modeling requirements due to their fundamentally different functionality. Sections 2.1.1 and 2.1.2 address specifics in the modeling of the use cases, after which the basic mathematical formulation is derived in Section 2.1.3.

### 2.1.1 Modeling of Charging and Discharging Efficiencies

A very important aspect is the different modeling of the charging and discharging efficiency since this significantly influences the complexity of the optimization. [Pub2] shows that power losses of an inverter consist of a constant self-consumption, voltage losses at diodes and transistors that are proportional to the output power, and quadratic power-dependent losses caused by ohmic loss resistances. Figure 2-2 (a) shows the comparison of the real charging efficiency and different modeled charging efficiencies in a linear programming.

Modeling a fixed charging efficiency (line 'fix', red) leads to proportionally increasing charging losses to the AC charging power. A comparison with the real losses (yellow line) shows that the losses are greatly underestimated, especially for small charging powers. Modeling charging losses as a linear function ('Interval 1', light blue) or a piecewise linear function ('Interval 2', dark blue) results in a much better representation of the real losses. The modeling results in a mixed-integer linear programming with higher complexity, because in both cases constant losses depending on the Boolean operation variable (operating or not operating) are included [Pub2].

Since real losses during charging and discharging differ from modeled losses with a fixed efficiency, especially at low powers, the typical charging and discharging power of a use case is important for choosing the modeling approach. For this purpose, Figure 2-2 (b) classifies the EV charging and discharging powers for the PV self-consumption optimization use case by showing the annual duration curve of the residual load of five different medium households from [Pub2] with an annual electrical demand of 3,800 kWh and a 5.5 kWp PV system. The residual load for the household in this case is calculated by the load subtracted by the PV generation. For these exemplary medium households, the absolute residual load is less than 2 kW in 85% of the time. Due to these numerous time points with low power at the household connection point, a linear modeling would lead to a strong underestimation of the losses. Accordingly, [Pub2] reports revenues for V2H for linear modeling that are 30% higher than revenues with mixed-integer linear modeling.





In contrast, for exclusively marketing of EVs in the electricity market in [Pub1], the charging and discharging powers are usually high. Provided that the price spreads between sold and purchased electricity are adequate, the EV will discharge and charge at full power if it has sufficient flexibility to maximize revenues. Further, the marketing in the electricity market will be done through a pool of EVs in which the EVs can be intelligently managed so that only high charging and discharging powers are called for a single EV. For this reason, the modeling of the use case arbitrage trading with fixed charging and discharging efficiencies is valid and yields reliable results.

### 2.1.2 Modeled Optimization Horizons

Participating in European electricity markets can include consecutive day-ahead trading, trading in the intraday auction followed by continuous intraday trading. In Germany, bids for the day-ahead trading have to be submitted by 12 noon [41]. For the German intraday auction, bids must be submitted by 3 pm [41]. Thereupon, continuous intraday trading starts. The modeling of consecutive marketing on spot markets with real trading hours leads to the requirement to formulate the model as a rolling optimization model.

In [Pub1], the developed rolling optimization model is introduced. With each optimization step, the starting point of the optimization period is set three hours forward. A limited forecast of two to three days is modeled. Figure 2-3 illustrates that depending on the start time for the optimization period, different prices are set from the day-ahead and intraday market. For the start time 12 noon, for example, prices from continuous intraday trading are set for the first 12 hours, prices from the day-ahead market for the following day (d+1) and a day-ahead price forecast for one more following day (d+2).

Consecutive optimization runs lead to the possibility of countertrades for a considered period. Purchased or sold energy in the day-ahead market can be countertraded in the intraday auction or in the continuous intraday trading. A more detailed description of the consecutive rolling optimization can be found in [Pub1].



Figure 2-3: Schematic representation of the rolling optimization runs with limited foresight of different market prices based on [Pub1]

For the modeling of the PV self-consumption optimization, a perfect forecast for a whole year is set. Since there are no consecutive actions for a period, like countertrading in

different electricity markets, and there is generally a good foresight for PV feed-in, this simplified perfect foresight is acceptable here [Pub2].

#### 2.1.3 Use Case Dependent Mathematic Formulation

In publications [Pub1] and [Pub2], the mathematic formulation for arbitrage trading with consecutive spot markets [Pub1] and for PV self-consumption optimization [Pub2] is presented. Further, in [Pub2] the modeling of combined PV self-consumption optimization and arbitrage trading is shown. In this section, the most important constraints, as well as the objective function of the different use cases are compared. For a more detailed derivation of the equations, [Pub1] and [Pub2] can be consulted.

First, Equations (2-1), (2-2), and (2-3) show the different objective functions of the use case dependent optimization problem. Variables  $p_{el,buy}$ ,  $p_{el,buy,v2g}$ ,  $p_{el,sell}$ , and  $p_{el,sell,v2g}$  refer to electricity market prices, household prices or feed-in tariffs. These prices are multiplied by the power at the grid connection point  $P_{GCP,in}$ ,  $P_{GCP,out}$ ,  $P_{GCP,in,v2g}$ , and  $P_{GCP,out,v2g}$  or, in the case of V2G, by the powers of the EVs  $P_{EV,C'}$ ,  $P_{EV,d'}$ ,  $P_{EV,cou,buy'}$ , and  $P_{EV,cou,sell}$ . Minimization is performed over all time steps t of a period T. Equation (2-1) focuses on minimizing the household electricity costs  $p_{el,buy}(t) \cdot P_{GCP,in}(t) \cdot \Delta t$  minus the feed-in revenues from PV electricity  $p_{el,sell}(t) \cdot P_{GCP,out}(t) \cdot \Delta t$  for the PV self-consumption optimization (V2H). Equation (2-2) minimizes the costs of purchasing electricity for the EV  $p_{el,buy}(t) \cdot (P_{EV,c}(t) + P_{EV,cou,buy}) \cdot \Delta t$ , while maximizing revenues of sold electricity  $p_{el.sell}(t) \cdot (P_{EV,d}(t) + P_{EV,cou.sell}) \cdot \Delta t$  in consideration of countertrades for purchasing  $P_{EV,cou,buy}$  and selling  $P_{EV,cou,sell}$  electricity in consecutive spot markets (V2G) independently of household electricity demand. For the combination of V2H and V2G (Comb), Equation (2-3) merges these optimizations without considering countertrades for complexity reasons. In Equation (2-3), V2G prices  $p_{el,buy}$  and  $p_{el,sell}$  are transferred to  $p_{el,buy,v2g}$  and  $p_{el,sell,v2g}$  to allow different pricing for arbitrage trading compared to household pricing. The charging and discharging powers of V2G  $P_{EV,c}$  and  $P_{EV,d}$  are transferred to  $P_{GCP,in,v2g}$  and  $P_{GCP,out,v2g}$  to separate the power flows for V2G from the household power flows. Since the prices in Equations (2-1) to (2-3) refer to energies, all power variables are multiplied by the corresponding timeframe  $\Delta t$ .

$$V2H \qquad min\left(\sum_{t=1}^{T} \left[p_{el,buy}(t) \cdot P_{GCP,in}(t) \cdot \Delta t - p_{el,sell}(t) \cdot P_{GCP,out}(t) \cdot \Delta t\right]\right) \qquad (2-1)$$

$$min\left(\sum_{t=1}^{T} \left[p_{el,buy}(t) \cdot \left(P_{EV,c}(t) + P_{EV,cou,buy}\right) \cdot \Delta t\right] - p_{el,sell}(t) \cdot \left(P_{EV,d}(t) + P_{EV,cou,sell}\right) \cdot \Delta t\right]\right) \qquad (2-1)$$

Objective functions

$$\min \left( \sum_{t=1}^{T} \left[ p_{el,buy}(t) \cdot P_{GCP,in}(t) \cdot \Delta t - p_{el,sell}(t) \cdot P_{GCP,out}(t) \cdot \Delta t + p_{el,buy,v2g}(t) \cdot P_{GCP,in,v2g}(t) \cdot \Delta t - p_{el,sell,v2g}(t) \cdot P_{GCP,out,v2g}(t) \cdot \Delta t \right] \right)$$

$$(2-3)$$

15

Second, Equations (2-4), (2-5), and (2-6) constrain the power flows at the grid connection point (GCP). Equation (2-4) includes the power flows of all modeled assets of a household for V2H. Equation (2-5) referring to the arbitrage trading constrains the power flows of the EV equal to the power flows of the GCP. Standby losses of the EV and EVSE  $P_{EV,l,s}$  are neglected in [Pub1] for V2G, as the operating hours are significantly higher than for V2H.  $P_{sched,AC}$  represents scheduled powers from previous optimization runs.  $P_{EV,cou,buy}$  and  $P_{EV,cou,sell}$  describe countertraded powers in current optimization run. For regulatory reasons, a separate grid connection point for the EV has initially been modeled here [Pub1]. Equation (2-6) combines V2H and V2G by adding to Equation (2-4) a separate modeled power consumption  $P_{GCP,in,v2g}(t)$  and supply  $P_{GCP,out,v2g}(t)$  for the V2G use case.

V2H 
$$P_{GCP,in}(t) - P_{GCP,out}(t) = P_{HH,el}(t) + P_{HP,el}(t) - P_{PV}(t) + P_{PV,curt}(t) + P_{SBS,c}(t) - P_{SBS,d}(t) + P_{EV,c}(t) - P_{EV,d}(t) + P_{EV,l,s}(t)$$
(2-4)

Power  
constraints  
GCP  
$$V2G \qquad P_{GCP,in}(t) - P_{GCP,out}(t) = P_{EV,c}(t) - P_{EV,d}(t) + P_{EV,cou,buy}(t) - P_{EV,cou,sell}(t) + P_{sched,AC}(t) \qquad (2-5)$$
  
$$P_{GCP,in}(t) - P_{GCP,out}(t) + P_{GCP,in,v2g}(t) - P_{GCP,out,v2g}(t) + P_{HH,el}(t) - P_{PV}(t) + P_{PV,curt}(t) + P_{SBS,c}(t) - P_{SBS,d}(t) + P_{EV,c}(t) - P_{EV,d}(t) + P_{EV,l,s}(t) \qquad (2-6)$$

Another elementary constraint of the use cases is the energy conservation of the EV storage level  $E_{EV}$ . Equation (2-7) formulates this constraint for V2H and Comb under consideration of constant charging and discharging losses of the EV  $P_{EV,l,const,c/d}$  depending on the charging and discharging decision  $b_{EV,c/d}$  and linear charging and discharging losses of the EV and EVSE  $P_{EV,l,c/d}$  depending on the charging and discharging power. Equation (2-8) constrains the EV storage energy conservation under consideration of countertrades  $P_{EV,cou,sell}$  and  $P_{EV,cou,buy}$ , while modeling losses by a fixed charging and discharging and discharging and discharging and discharging and charging and charging losses is stored in the variable  $P_{sched}$ . Public charging  $E_{EV,pub,c}$  and driving energy  $E_{EV,drive}$  also affect the storage level.

$$E_{EV}(t) = E_{EV}(t-1) + \left[P_{EV,c}(t) - P_{EV,l,c}(t)\right] \cdot \Delta t$$
Storage
energy
conservation
EV
$$E_{EV}(t) = E_{EV}(t-1) + P_{EV,l,const,c/d} \cdot \left[b_{EV,c}(t) + b_{EV,d}(t)\right] \cdot \Delta t \quad (2-7)$$

$$+E_{EV,pub,c}(t) - E_{EV,drive}(t)$$

$$E_{EV}(t) = E_{EV}(t-1) + P_{EV,c}(t) \cdot \mu_{EV,c} \cdot \Delta t$$

$$+P_{eV,cou,sell}(t) \cdot \mu_{EV,c} \cdot \Delta t - \frac{P_{EV,cou,buy}(t)}{\mu_{EV,d}} \cdot \Delta t + \frac{P_{EV,cou,buy}(t)}{\mu_{EV,d}} \cdot \Delta t \quad (2-8)$$

$$+P_{sched} \cdot \Delta t + E_{EV,pub,c}(t) - E_{EV,drive}(t)$$

For all use cases considered, Equations (2-9), and (2-10) further constrain the EV's battery storage level. Equation (2-9) sets the minimum battery storage level in dependance of the maximum EV battery capacity  $E_{EV,max}$ , and a parameterized safety state of charge (SoC)  $SoC_{safe}$  for all time steps  $c_{EV,connected}$ , in which the EV is connected to the EVSE. This limitation on storage capacity is introduced so that the user can make at least one trip to the hospital with the EV at any time for safety reasons. Equation (2-10) further constrains the minimum battery storage level depending on the user
parameterization  $SoC_{dep}$  and the EV battery capacity  $E_{EV,max}$  for all time steps  $c_{EV,dep}$ , in which the EV is scheduled to depart. This is again due to realistic modeling of the user, who can specify a desired minimum SoC when the EV departs.

Safety			
Soc	V2H/	$E_{EV}(t) \ge SoC_{safe} \cdot E_{EV,max} \cdot c_{EV,connected}(t)$	(2-9)
500	V2G/		
Departure SoC	Comb	$E_{EV}(t) \ge SoC_{dep} \cdot E_{EV,max} \cdot c_{EV,dep}(t)$	(2-10)

The mathematical formulation of the use cases shown here reveals fundamental differences in modeling. In [Pub1] and [Pub2], numerous other constraints are introduced to model a realistic design of the use cases. For example, the simultaneous combination of V2G and V2H is restricted for regulatory reasons, and the modeling of countertrades is described in more detail for V2G. More details of the modeling can therefore be taken from the publications.

## 2.2 Revenue Potentials and Their Most Important Influencing Factors

In the following, Section 2.2.1 presents revenue potentials for PV self-consumption optimization as a V2H use case based on [Pub2]. Section 2.2.2 shows revenue potentials for arbitrage trading as a V2G use case on the basis of [Pub1]. Then, Section 2.2.3 combines these use cases for an evaluation of added revenues by a use case combination based on [Pub2]. More detailed descriptions of parameter assumptions and revenue assessments can be found in [Pub1] and [Pub2].

#### 2.2.1 Photovoltaic Self-Consumption Optimization

For the evaluation of revenue potentials from PV self-consumption optimization, a medium household (average annual electricity demand of 3,800 kWh) and user type (not regularly commuting to work) is defined, for which 20 different instances are modeled. The average revenues for such a medium household are around 210  $\notin$ /a for a smart EV compared to an unmanaged charging EV and around 310  $\notin$ /a for a bidirectional EV compared to an unmanaged charging EV.

Based on this medium household and user, Figure 2-4 (a) analyzes numerous sensitivities to this basic configuration (scenario *Base*). First, as already mentioned in Section 2.1.1, linear programming with fixed charging and discharging efficiencies (*Base linear*) results in 30% overestimated revenues for a bidirectional EV. From this, it was concluded in [Pub2] that modeling of fixed charging and discharging efficiencies is not valid for V2H. Second, a user who regularly commutes to work (*COM*) has 30% decreased revenues for a bidirectional EV, but only 5% decreased revenues for a smart EV. Since commuting EVs on average have a higher annual consumption and thus greater potential for smart charging, this compensates for lower availability. Integrating other smart components into the home, such as heat pumps (*HP*), stationary battery storage (*SBS*), or a combination of those (*HP SBS*), reduces the revenues of a smart or bidirectional EV significantly. These other smart components also use the cheaper PV energy and thus compete with the EV.

The correlation of revenues to PV self-consumption is evident in Figure 2-4 (b). In the scenarios, where the PV self-consumption can be increased more, the revenues also increase more.





There are numerous other quantitative influencing factors that affect the revenue potentials of smart and bidirectional EVs, as shown in Figure 2-5. Starting from the household and user scenario Base (represented by the central point in the chart), the figure shows how an increase or decrease of the influencing factors affect the revenues. The steeper the curve in the chart, the greater is the impact of the influencing factor. The PV parameters peak power and feed-in tariff affect the revenues significantly. A lower feed-in tariff leads to greater revenues since the spread between household electricity price and feed-in tariff increases and thus the self-consumption of PV energy becomes more profitable. In contrast, the parameters of the EV have only a minor impact on revenues. Since the power flows in a household are small, a smaller or larger power of the EVSE as well as a smaller or larger EV battery capacity has little impact on revenues. In the course of the fuel crisis and the strong increase of electricity prices in Europe since the end of 2021 [63], there was a sharp rise in electricity stock exchange prices in Europe and consequently also in household electricity prices. Therefore, the household electricity price was added to the influencing factors analyzed in [Pub2]. It is evident that the revenue changes for changes in the household electricity price show an even higher absolute gradient than for the PV feed-in tariff. This is because the absolute difference between the household electricity price and the PV feedin tariff is decisive for the revenue potentials. A relative increase of the feed-in tariff by 70% and a relative reduction of the household electricity price by one third equally lead to a

reduced spread between household electricity price and PV feed-in tariff of 10 ct/kWh. Revenues are reduced by 65% in both cases. In a *Max* scenario with the most attractive characteristics of the influencing factors in Figure 2-5, the revenues go up to 1,300 €/a for the bidirectional EV and 750 €/a for the smart EV, which shows the high revenue potential of the use case.





Overall, the use case of PV self-consumption optimization through smart or bidirectional EVs is very sensitive. To estimate revenues, it is necessary to know the exact configuration of the household and user. The most important factor influencing revenue potentials is the spread between household electricity prices and PV feed-in tariff. The higher is the spread, the higher are the revenue potentials.

#### 2.2.2 Arbitrage Trading

For the evaluation of arbitrage trading in the German electricity markets in 2019, Figure 2-6 presents the revenues of smart and bidirectional EVs dependent on the EV and EVSE configuration, the user type, and the considered electricity market. Revenues again refer to the differential revenues of smart and bidirectional EVs compared to unmanaged charging EVs. For this initial revenue analysis, an exemption from taxes, levies and surcharges is assumed for charged electricity that is temporarily stored and thus discharged later.

Various influencing factors can be identified that have a strong impact on revenues of bidirectional EVs. Marketing in the intraday market can generate 50% to 100% higher revenues than marketing in the day-ahead market due to the higher price volatility in the intraday trading. Consecutive marketing on the day-ahead and intraday market can increase the revenue potential by another 10% compared to intraday trading. Furthermore, a higher battery capacity of the EV and a higher charging and discharging power of the EVSE have a strong revenue increasing effect. Non-commuters can generate on average about 15% higher revenues that are only 5% to 25% of bidirectional EVs. These revenues depend only slightly on the EV and EVSE configuration, but more on the considered electricity market. However, revenue potentials of smart EVs are comparably low so that smart charging is far less relevant for the use case arbitrage trading than for the use case of PV self-consumption optimization.





Two other very important factors influencing revenue potentials are the regulation and thus the taxes, levies and surcharges that must be paid on charged electricity, and the electricity price volatility, as shown in Figure 2-7. For these studies, a non-commuting EV with battery capacity of 100 kWh and a charging and discharging power of 11 kW has been assumed.

Figure 2-7 (a) shows that the revenues (here exemplified in the intraday market) decrease sharply with increasing taxes, levies, and charges to be paid on charged electricity. In Germany, storage facilities are eligible for exemption from taxes, levies, and surcharges on charged electricity that is later discharged. The BCM project has published a policy paper on this subject that details the possibilities for exemption from taxes, levies, and surcharges [42]. In the case of a regulatory classification of bidirectional EVs as storage systems, reduced taxes, levies, and surcharges on temporarily stored electricity will be incurred. There is currently no exemption option for the 'StromNEV' levy (depending on the

annual energy throughput in the storage facility) and the 'AbLaV' levy. A home storage system must also pay a concession fee to the city or municipality on charged electricity, which does not necessarily apply to a large battery storage system or a pumped storage system [42]. The exemptions from taxes, levies, and surcharges result in the reduced revenues shown in Figure 2-7 (a) for a regulatory classification as pumped storage hydropower, large-scale stationary battery storage, or home storage system. Since even with the intuitive regulatory classification of a bidirectional EV as a home storage system, the revenue potentials drop by almost 70%, further exemptions from taxes, levies, and surcharges should be created to make arbitrage trading economically more feasible.

Figure 2-7 (b) shows the dependence of the revenue potentials of bidirectional EVs on the characteristics of electricity prices, which changes significantly in the years 2020 to 2050. The underlying electricity prices are modeled prices using the energy system model ISAaR [39], which is described in more detail in Chapter 3, without taxes, levies, and surcharges. Electricity price volatility, shown in the figure by the average daily standard deviation of electricity prices over one year, increases sharply in the considered scenario in future years. Due to the strong expansion of variable renewable energies that have a high simultaneity in electricity generation, there are numerous times with very low electricity prices. Additionally, there are increasingly times with high electricity prices due to higher fuel costs and CO<sub>2</sub>-prices, which increase the marginal costs of thermal power plants. Overall, this leads to a strongly increasing mean daily standard deviation of the electricity price. Revenue potentials correlate very strongly with the daily standard deviation of the electricity price, as bidirectional EVs often act as daily storage. The influence of the electricity price volatility is also shown in [Pub1] by modeling the revenue potentials of bidirectional EVs in 28 different European countries for the year 2019. Revenues for modeled noncommuting EVs trading in the day-ahead market vary from 50 €/EV/a in Norway to 700 €/EV/a in Hungary. In the course of the energy crisis in 2021, the electricity price volatility has increased a lot, so that in 2021 the average daily standard deviation of the dayahead price of 24.5 €/MWh was almost 3 times as high as in 2019 [63]. Accordingly, the revenue potentials have also increased by about three times.

In summary, the revenue potentials of the arbitrage trading use case are also very sensitive, like those of the PV self-consumption optimization. In contrast to the PV self-consumption optimization, the influencing factors of electricity price volatility, regulatory framework, as well as the EV and EVSE characteristics are of great importance. Other influencing factors, including, for example, user behavior and liquidity of the electricity spot markets, are discussed in [Pub1].





## 2.2.3 Combined Photovoltaic Self-Consumption Optimization and Arbitrage Trading

In the combined modeling of PV self-consumption optimization and arbitrage trading, arbitrage trading is limited to the day-ahead market for reasons of complexity. From a regulatory point of view, a simultaneous implementation of V2H and V2G is challenging, since the PV electricity could then be fed into the grid via the EV and thus possibly pay reduced taxes, levies and surcharges for electricity purchased later. Therefore, daily usage is limited to either V2H or V2G so that seasonal characteristics can also be analyzed. For trading in the electricity market, an exemption from taxes, levies and surcharges is assumed for this study.

Figure 2-8 shows in the upper diagram first the weekly standard deviation of electricity prices and the weekly PV generation of the medium household (see Section 2.2.1). In the lower part of Figure 2-8 the share of V2G and V2H used by the 20 modeled households is shown. It is evident that V2H PV self-consumption optimization is clearly preferred in the summer months from April to September. When comparing the diagrams, a correlation with the PV feed-in can be identified. In contrast, V2G is used more in the winter months, when there are slightly higher spreads of electricity prices in the spot market, expressed by the higher standard deviation. It can be deduced from this that V2H and V2G can in principle be combined very well due to their seasonally different revenue potentials.

From a revenue perspective, the combination of the use cases PV self-consumption optimization and arbitrage trading brings significant added value. Thus, the revenues of the medium household can be increased by 70% from 310  $\epsilon$ /a to 530  $\epsilon$ /a. Interestingly, the actual V2H revenues are only slightly reduced, since V2H is hardly carried out in the winter months due to small PV feed-in quantities. Even with a modeled limitation of the operating

hours (OHs) and the equivalent full cycles of the battery (EFCs) to 5 OHs/d and 130 EFCs/a, respectively, significant additional revenues of 100  $\notin$ /a can still be generated by this use case combination.





## 2.3 Profitability of Bidirectional Charging

To evaluate the economic profitability of bidirectional EVs performing the PV selfconsumption optimization or arbitrage trading use cases, the additional costs of bidirectional charging must be considered in addition to the revenues highlighted in Section 2.2.

#### 2.3.1 Additional Costs of Bidirectional Charging

Based on research and discussions with experts in the BCM project, the following additional cost components for bidirectional charging of EVs were identified:

- purchase of bidirectional EVSE
- installation of bidirectional EVSE
- operation of bidirectional EVSE
- installation and operation of additional metering equipment
- additional hardware/software
- purchase of bidirectional EV
- operation of bidirectional EV
- additional processes of registration, permits and contracts

Detailed information on the additional costs can be found in [43]. For the purchase and installation of the bidirectional EVSE, the installation and operation of additional metering equipment and additional hardware, costs and cost predictions were determined within the BCM project. The operation costs of the bidirectional EVSE through additional losses are included in the revenue modeling. The additional purchase and operation costs of the bidirectional EV could not be determined. Since Volkswagen, as an example of an EV manufacturer, sees bidirectional charging as a standard in the future [44], there could also be no additional investment costs for a bidirectional EV, at least for upper-class models. The costs for additional processes of registration, permits and contracts are set to zero since it is uncertain whether there will be additional costs for it.

Table 2-1 shows the projected development of additional investment costs of bidirectional charging compared to unmanaged charging. It can be noticed that especially in 2020 the additional investment costs are driven by the purchase of the bidirectional EVSE, which will show a significant cost degression in the future. The bidirectional EVSE installation is more expensive than the installation of an unmanaged EVSE due to higher working time costs and potential laid empty conduits with wall openings for network connection. Additional hardware includes either an optocoupler or an additional smart energy meter. In future years, an existing smart energy meter might be sufficient, such that additional hardware costs could be reduced to zero."

	2020	2025	2030	2035	2040
EVSE purchase	5,300-5,700	2,000-2,100	1,400-1,500	1,100-1,200	800-900
EVSE installation	830-880	60-350	60-350	60-350	60-350
Additional hardware	100-450	100-450	0-450	0-450	0-450
Total	6,230-7,030	2,160-2,900	1,460-2,300	1,160-2,000	860-1,700

Table 2-1:Additional investment cost range for V2H, and V2G in €2021/EV compared to<br/>unmanaged charging based on [43]

Table 2-2 shows the additional yearly costs due to additional modern measuring devices and smart meter gateways (SMGWs). SMGWs are mandatory in Germany under certain circumstances, e.g., for consumers with an electricity consumption of more than 6,000 kWh per year [45]. Since the modeled medium household with the EV electricity demand has a higher electricity consumption, the SMGW is mandatory regardless of whether it is an unmanaged charging EV or a bidirectional EV. Therefore, only one additional modern measuring device with costs of 20  $\in_{2021}/a$  is needed. These costs are assumed to be constant in real terms.

Table 2-2:Additional yearly cost range for V2H, and V2G in €2021/EV/a compared to<br/>unmanaged charging based on [43]

	2020	2025	2030	2035	2040
Metering equipment	20	20	20	20	20
Total	20	20	20	20	20

#### 2.3.2 Evaluation of Profitability

For final evaluation of the economic profitability, the findings on the revenue potentials of bidirectional EVs from Section 2.2 and the additional costs of bidirectional EVs versus unmanaged charging EVs from Section 2.3.1 are brought together in this section. The three investment dates 2020, 2025 and 2030 for a bidirectional EV are considered. The one-time and yearly additional costs for bidirectional charging compared to unmanaged charging are obtained from Table 2-1 and Table 2-2.

The revenues for the use case PV self-consumption optimization are calculated based on sensitivity analyses for a differing PV feed-in tariff for a medium household. Table 2-3 presents the revenues as well as the underlying parametrization for the years 2020 to 2030. The feed-in tariff of the PV electricity generation for 2020 is taken from [67] for a PV plant with a peak power smaller 10 kW constructed in July 2020. For the years 2025 (5.5 ct/kWh) and 2030 (4 ct/kWh) a further reduction of the fixed PV feed-in tariff has been assumed according to the trend of the last years. In April 2022, for example, the feed-in tariff had already been reduced to 6.5 ct/kWh [67]. All other parameters, such as PV peak power (5.5 kW), household electricity price (29.9 ct/kWh) and EV characteristics are left constant according to the baseline household scenario described in Section 2.2.1. The household electricity price is kept constant in nominal terms, which corresponds to a reduction in real terms. The actual future development of household electricity prices shows great uncertainties here, especially in the context of the current energy crisis. Increasing household electricity prices would mean increasing revenue potentials from PV self-consumption optimization, as shown in Section 2.2.1.

	2020	2025	2030
Household type	medium	medium	medium
PV feed-in tariff	9.0 ct/kWh	5.5 ct/kWh	4 ct/kWh
EV/EVSE parameterization	60 kWh battery ca powe	apacity, 11 kW chargi r, non-commuting u	ng/discharging ser
Revenues	380 €/a	470 €/a	530 €/a

Table 2-3:	V2H Revenues based on household type and PV feed-in tariff for the years 2020,
	2025 and 2030

For future years after the investment year, the revenues are generally assumed to be constant. The parameters of the household do not change, the fixed feed-in tariff is guaranteed to be paid for a period of 20 years [68], and the household electricity price is assumed to be constant. Since the reference year for costs is 2021, all revenues must consequently still be discounted. For this purpose, an average future inflation rate of 1.4% is used based on the average inflation rate in Germany from 2012 to 2021 [46]. In addition, a real interest rate of 1.6%, reflecting risk premium rate, is used, resulting in a nominal interest rate of 3%. This nominal interest rate  $i_n$  of 3% is used for discounting the revenues  $Rev_t$  of a time step t to the discounted revenues  $Rev_0$  by following Equation (2-11). The discounting formula is based on [69].

$$Rev_0 = \frac{Rev_t}{(1+i_n)^t}$$
(2-11)

The net present value *NPV* is thus calculated as the sum of all revenues less the initial investment costs  $costs_{initial}$  and the sum of the annual costs  $costs_{yearly}$  in real terms based on [70] in Equation (2-12).

$$NPV = \sum_{0}^{t} \frac{Rev_t}{(1+i_n)^t} - costs_{initial} - \sum_{0}^{t} costs_{yearly}$$
(2-12)

The yearly cash flows as well as the NPV for the three scenarios with a V2H investment in 2020, 2025 and 2030 are shown in Figure 2-9. Due to the high investment costs of a bidirectional EVSE, an investment in V2H is not economical in 2020. However, with an investment in V2H in 2025, a positive NPV can already be achieved after seven years, which is 2,300  $\in_{2021}$  at the end of the EVSE's lifetime of 15 years. An investment in 2030 achieves a positive NPV after a little more than five years, which rises to 2,800  $\in_{2021}$  at the end of the EVSE's lifetime.





The revenues for the profitability calculation of the arbitrage trading use case are determined using the parameterization of Table 2-4. For better comparability, the same EV and EVSE parameterization (including the same EV profiles) is assumed as in the profitability evaluation of the V2H use case, but with a fixed charging and discharging efficiency. For future years, a gradual exemption of the charged, temporarily stored electricity from taxes, levies and surcharges is assumed. In 2025, the bidirectional EV is classified as a home storage system (see Section 2.2.2), and from 2030 there is a complete exemption from taxes, levies, and surcharges.

Marketing is limited to trading in the day-ahead market. For the year 2020, historical dayahead electricity prices are used [47]. For future years, modeled electricity prices are provided from the energy system model ISAaR based on simulations of the future European energy system in Section 3.4.2 with bidirectional EVs already integrated (*BCM* scenario). The electricity price time series have the mean daily standard deviations shown in Table 2-4. [Pub3] explains the future electricity price characteristics in more detail.

	2020	2025	2030	2035	2040	
Regulatory framework	All taxes, levies, and surcharges must be paid	Classification as home storage system: 2,1 ct/kWh	full exemption from taxes, levies, and surcharges	full exemption from taxes, levies, and surcharges	full exemption from taxes, levies, and surcharges	
Daily standard deviation of electricity prices in €/MWh	9.4	15.8	18.0	17.6	16.8	
EV/EVSE parameterization	60 kWh battery capacity, 11 kW charging/discharging power, 86% roundtrip efficiency, non-commuting user					
Revenues in €/a	60	120	360	340	280	

Table 2-4:V2G revenues based on regulatory framework, electricity price characteristic, as<br/>well as EV and EVSE parameterization for the years 2020 to 2040

Unlike V2H revenues, V2G revenues are not left constant after the investment decision of a bidirectional EV but are adjusted based on Table 2-4, since a regulatory change and a change of the electricity price characteristics also apply to earlier investments. The revenues are deliberately not interpolated here since the adjustment of the regulatory framework is a discrete decision. V2G revenues are only discounted by the risk premium rate of 1.6% since prices in the energy system model ISAaR are modeled in real terms (inflation-adjusted) and further related to the base year 2021.

Figure 2-10 shows the resulting profitability of V2G arbitrage trading depending on the investment year. For an investment in a bidirectional EV and EVSE in 2020, the NPV is strongly negative since revenues in the years 2020 to 2030 are low mainly due to regulatory issues. Investing in a bidirectional EV and EVSE in 2025 leads to a positive NPV after 12 years. Only a later investment in 2030 results in a short amortization period of around 7 years. Thus, with an investment in a bidirectional EV and EVSE in 2030, an NPV of around 1,650  $\in_{2021}$  can be achieved after 15 years.



Figure 2-10: Profitability of V2G arbitrage trading depending on the investment year of bidirectional EV and EVSE

In summary, for the investigated parameterization of household, EV and EVSE, user behavior, regulatory framework, and electricity price characteristics, both use cases can become profitable in future years. V2H PV self-consumption optimization is the more secure economic use case, which depends less on the regulatory framework. However, the basic requirement for this use case is the ownership of a PV system, so that not all users can rely on it. In contrast, V2G arbitrage trading through, for example, participation in the intraday market, has similarly diverse opportunities to improve revenue potential. Finally, it is important to note that both use cases are extremely sensitive, as explained in Section 2.2, so that the final economic analysis must always be carried out with the real parameterization of a considered scenario.

## 3 Bidirectionally Chargeable Electric Vehicles from the Energy System Perspective

Due to ambitious European and national climate targets, the European energy system is facing major structural changes. Based on [Pub3] and [Pub4] and further research, this chapter analyzes the added value of bidirectional EVs for the European energy system. For this purpose, the focus is on systemically oriented V2G use cases. Section 3.1 analyzes the different modeling types of bidirectional EVs in a large-scale energy system model for the integration of EVs into the electricity market (use case arbitrage trading) and for the provision of system services by bidirectional EVs (use case congestion management). To evaluate the impact of bidirectional EVs on the energy system, Section 3.2 first defines scenarios. Based on the scenario definition and the modeling approaches, Section 3.3 analyzes the cost-optimal penetration rates of bidirectional EVs and their influencing factors. Finally, Section 3.4 examines the quantitative impact of bidirectional EVs on the European energy system.

#### 3.1 Methods

For the consideration of the added value of bidirectional EVs from a system perspective, the energy system model ISAaR is further developed. Detailed information on ISAaR can be found in [31] and [39]. Section 3.1.1 focuses on modeling the integration of bidirectional EVs in the electricity market. Section 3.1.2 builds on the market integration and extends the subsequent transmission grid run by bidirectional EVs that can provide congestion management. In addition to the mathematical formulation in this section, the underlying data model, which contains a detailed description of the techno-economic parameters of all components of the European energy system, is also of great significance. The parameterization of the components is described in the following Section 3.2.

#### 3.1.1 Modeling of Electricity Market Integration

The energy system model ISAaR forms a linear optimization problem with the objective function to minimize the European energy system costs. The linearized equations describe techno-economic relations within an energy system. The objective function is formulated in Equation (3-1). The costs *C* result from operation and investment costs. The operation costs are formed by specific costs  $f_{t,n,i}$  and the utilization of the optimization variables  $x_{t,n,i}$  for all time steps *t*, nodes *n* and instances *i*. Variables  $x_{t,n,i}$  include generation, consumption, import and exports of all modeled energy carriers. Investment costs and other not time-dependent costs are calculated using specific costs  $f_{n,i}$  and the utilization of the optimization variables  $x_{n,i}$  include various kind of assets, e.g. storages, renewable energies or hydrogen-fired power

plants. In the market optimization run, nodes *n* refer to market areas in the European energy system.

$$C = \min\left(\sum_{t \in T} \sum_{n \in N} \sum_{i \in I} (f_{t,n,i} \cdot x_{t,n,i}) + \sum_{n \in N} \sum_{i \in I} (f_{n,i} \cdot x_{n,i})\right)$$
(3-1)

From the perspective of the ISAaR energy system model, bidirectional EVs represent an expansion option that replace unmanaged charging EVs. This expansion option is accompanied by investment costs that depend on necessary additional infrastructure and software as well as additional operational costs. These costs are included in the objective function of ISAaR. In addition to bidirectional EVs, the model can also replace unmanaged charging EVs with smart EVs that only charge unidirectionally and have lower investment costs.

The charging and discharging powers of bidirectional EVs are included in the power balances. According to Equation (3-2), the final energy demand  $P_{demand}$  is equal to generation  $P_{gen}$  and imports  $P_{import}$  minus consumption  $P_{cons}$  and exports  $P_{export}$  for every timestep t, nodes n and energy carrier c over all instances i. The charging of an EV is included in the system power balance as consumption and the discharging as generation.

$$P_{demand}(t, n, c) = \sum_{i} P_{gen}(t, n, c) - \sum_{i} P_{cons}(t, n, c) + \sum_{i} P_{import}(t, n, c) - \sum_{i} P_{export}(t, n, c)$$

$$(3-2)$$

The modeling of bidirectional EVs integrated into the European electricity market and thus of an energy system-optimized use of the vehicles corresponds to the use case of arbitrage trading from the user's perspective. In a system-optimal operation, bidirectional EVs are charged at times of low electricity prices and discharged at times of expensive electricity prices. In this way, they can displace the use of thermal power plants, which produce at expensive marginal costs when fuel and CO<sub>2</sub> prices are high. Compared to the modeling of the use case of arbitrage trading from the user perspective discussed in Chapter 2, however, the focus is now on the integration of the EVs in the day-ahead markets, since large energy system models do not typically model the short-term intraday markets. Price formation on intraday markets is based primarily on shorter product duration (in Germany quarter-hourly, half-hourly and hourly products), lower market liquidity and short-term forecast errors [48], neither of which is represented in large energy system models. Further, through cost-optimal integration of bidirectional EVs, the impact on electricity day-ahead prices is endogenously included in ISAaR.

Based on Equation (2-8), the storage energy conservation equation of EVs is simplified so that no counter trades are represented in the intraday market in Equation (3-3). The EV's storage level  $E_{EV}$  in a timestep t is equal to the storage level of the previous time step added by the charged energy  $P_{EV,c}(t) \cdot \mu_{EV,c} \cdot \Delta t$  and the charged energy in public  $E_{EV,pub,c}(t)$  subtracted by the discharged energy  $\frac{P_{EV,d}(t)}{\mu_{EV,d}} \cdot \Delta t$  and the consumed energy by driving  $E_{EV,drive}(t)$ . The charging and discharging efficiencies  $\mu_{EV,c}$  and  $\mu_{EV,d}$ again are modeled constant due to predominantly high charging and discharging powers (compare Section 2.1.1).

$$E_{EV}(t) = E_{EV}(t-1) + P_{EV,c}(t) \cdot \mu_{EV,c} \cdot \Delta t - \frac{P_{EV,d}(t)}{\mu_{EV,d}} \cdot \Delta t + E_{EV,pub,c}(t) - E_{EV,drive}(t)$$
(3-3)

By using Equations (2-9) and (2-10) for the modeling of a safety SoC and a departure SoC that constrain the battery storage level, a realistic user behavior is also provided.

Discrete modeling of all individual EVs in the European energy system would lead to a great number of modeled storage instances, which would introduce an enormous complexity. In [Pub3] three approaches for modeling bidirectional EVs in large-scale energy system models are presented to solve this problem. All approaches have the goal of modeling entire vehicle fleets in countries while keeping the complexity of the model as low as possible. Table 3-1 sums up the approaches and their advantages and disadvantages.

	1,000 Discrete EV Profiles per Market Area	Clustered, Discrete EV Profiles per Market Area	Aggregated EV Profile per Market Area			
Modeling Approach	• Modeling of 1,000 discrete EV instances to represent EV fleet	• Modeling of clusters with 5 to 50 discrete EV instances to represent EV fleet	• Modeling of one aggregated EV instance to represent EV fleet			
Advantages	Best representation of EV fleet and its impact on energy system	<ul> <li>Exact representation of EV constraints</li> <li>Good representation of EV fleet and its impact on energy system</li> </ul>	<ul> <li>Lowest modeling complexity</li> <li>Good representation of EV fleet and its impact on energy system</li> </ul>			
Dis- advantages	<ul> <li>High complexity</li> <li>Not computable for optimized expansion of bidirectional EVs</li> </ul>	High complexity	No exact     representation of EV     constraints			

Table 3-1:Modeling approaches of bidirectional EVs in a large-scale energy system model<br/>and their advantages and disadvantages

The modeling approach via aggregation of all EV profiles softens the EV constraints that restrict the SoC. Figure 3-1 shows the aggregated availability of EVs at the location at home compared to a single EV's availability for an exemplary week. Discrete EV profiles have time dependencies that are neglected when modeling an aggregated EV profile. A single EV departing with a SoC resulting from the system-optimal operation of the vehicle will return with a SoC that dependency, but still accounts for the overall SoC constraints of the EV pool due to the aggregated minimum availability. Since the modeling approach via aggregation of all EV profiles still leads to a good representation of the EV fleet and its impact on the European energy system, while exhibiting the lowest modeling complexity, it is chosen for further investigation. A more detailed description of the comparison of the modeling approaches can be found in [Pub3]. Further, smart EVs that only charge unidirectional can also be modelled by the same method. The only difference is their discharge power that is set to zero [Pub3].



Figure 3-1: Different availability of aggregated and discrete EV profiles for an exemplary week as presented in [Pub3]

#### 3.1.2 Modeling of Congestion Management Provision

Modeling of congestion management in ISAaR is based on a two-stage optimization:

- First run: multi-energy market optimization (described in Section 3.1.1)
- Second run: transmission grid optimization (described in this section)

The transmission grid optimization run fixes the investment in new assets based on the results of the multi-energy market optimization. Further, the dispatch of assets in the multi-energy market optimization represents the baseline for their operation in the transmission grid optimization, which can only deviate from the baseline with additional costs. For these reasons, the objective function of the transmission grid optimization differs from the market run according to Equation (3-4). It is limited to operation costs and costs incurred by the provision of congestion management for all time steps t, nodes n and instances i dependent on specific costs  $f_{t,n,i}$  and the utilization of the optimization variables  $x_{t,n,i}$ . In the transmission grid optimization run, nodes n refer to transmission grid nodes in the European energy system. Generators, loads and flexibilities in the distribution grid are aggregated for complexity reasons and assigned to the nodes of the transmission grid.

$$C = \min\left(\sum_{t \in T} \sum_{n \in N} \sum_{i \in I} (f_{t,n,i} \cdot x_{t,n,i})\right)$$
(3-4)

Congestion management costs (increase of dispatch) and revenues (decrease of dispatch) of thermal power plants are set dependent on their operational costs. Revenues for curtailment of renewable energies are set to zero to ensure their curtailment as a last option. Costs and revenues for storages that provide congestion management are set low. Their main costs arise from the subsequent compensation of schedule deviations. This energy compensation results in costs that are dependent on congestion management costs of other assets that provide the compensation.

The power balance in the transmission grid run is formulated per transmission grid node and thus has a significantly higher regional resolution than the market run. For the line flow, a DC (direct current) load flow based on PTDF (Power Transfer Distribution Factors) that represents a linearized approximation of the non-linear alternating current (AC) load flow is applied. This simplified modeling of the load flow is permissible under the assumption that there are no voltage drops (voltage amplitude is equal for all nodes), reactive power is neglected, line losses are neglected, and the voltage angle differences are small [49]. Modeling the DC load flow based on PTDF, the load flow of AC lines  $P_{line,AC}$  is determined by the *PTDF* matrix multiplied by the injections and withdrawals in a grid node  $P_{node}$  according to Equation (3-5). The load flow of DC lines  $P_{line,DC}$  can be fully controlled due to converter stations at the end and at the beginning of a DC line and are therefore modeled via a transport model from node *x* to *y* according to Equation (3-6).

$$P_{line,AC} = PTDF \cdot P_{node} \tag{3-5}$$

$$P_{line,DC} = P_{x \to y} \tag{3-6}$$

A detailed derivation of the PTDF equations has already been described numerous times and can be found for example in [50], [49], [31] as well as in [Pub4].

Injections and withdrawals in a grid node are the result of generation and consumption of all assets in the market run as well as their congestion management optimized in the transmission grid run. The output of power plants and renewable energies  $P_o$  is equal to the market result  $P_M$  added by the increased dispatch  $P_p$  and subtracted by the decreased dispatch  $P_n$  for all timesteps *t* according to Equation (3-7). Increasing dispatch represents positive congestion management and decreasing dispatch means negative congestion management. Variable renewable energies operate at their maximum output in the market run, considering a possible market-related curtailment. Therefore, their variable  $P_p$  is set to zero. Storages and thus also bidirectional EVs have the possibility to consume electricity  $P_i$ in addition to the possibility to generate electricity  $P_o$  from the system's perspective. For this reason, Equation (3-7) is expanded to Equation (3-8).

$$P_o(t) = P_M(t) + P_p(t) - P_n(t)$$
(3-7)

$$P_o(t) - P_i(t) = P_M(t) + P_p(t) - P_n(t)$$
(3-8)

Modeling of storages providing congestion management is more complex due to their time-coupling energy conservation constraint formulated in Equation (3-3) for bidirectional EVs. The temporal coupling further leads to the fact that time steps cannot be arbitrarily decomposed and parallelized in order to reduce complexity of the simulations. Figure 3-2 schematically illustrates the time-dependence of storages that compensate their adjusted schedule in the transmission grid run. Positive and negative congestion management is compensated by the opposite congestion management in a different time step. The storage level at the end of a time slice is fixed to the market run output and passed to the next time slice. Thus, the time slices are coupled only by the market run and can be simulated in parallel. The length of the self-contained optimized time slices can be selected variably. On the one hand, a longer time slice leads to greater flexibility to compensate for an adjusted schedule of a storage facility. On the other hand, it leads to a greater complexity of the optimization problem. [Pub4] analyzes the effect of different time slice lengths in detail. The results presented in Section 3.4.3 are based on a time slice length of one week.



Figure 3-2: Schematic modeling of the restricted storage state of charge at the start and the end of a time slice as presented in [Pub4]

### 3.2 Scenario Definition

For an evaluation of the cost-optimal integration of bidirectional EVs into the future energy system, a scenario is set up that forms the framework for this study. This is based on the climate protection scenario solidEU from the eXtremOS project, which aims a 95% reduction in greenhouse gas emissions in Europe by 2050 with a strong electrification of the final energy sectors. Detailed information on the scenario and the techno-economic parameters can be found in [3]. For Germany, this scenario was updated to include targeted capacities for renewable energies and power plants in the year 2030 planned by the German government [51]:

- Installed capacity of PV plants: 200 GW
- Installed capacity of wind onshore plants: 100 GW
- Installed capacity of wind offshore plants: 30 GW
- Coal phase-out by 2030

The solidEU scenario did not provide the opportunity to integrate smart or bidirectional EVs in the future energy system. Therefore, the solidEU scenario with previously addressed adjustments serves as a reference scenario *Ref* to highlight the added value of an energy system with bidirectional EVs compared to a system without bidirectional EVs.

Based on the modeling with an aggregated EV profile per country (see Section 3.1.1), the model is endogenously allowed to integrate smart or bidirectional EVs into the system. For this purpose, Table 3-1 defines the parameterization of the EVs (including their EVSEs). The EVs are parameterized by a medium passenger car battery capacity of 50 kWh [52]. The other parameters are based on discussions within the research project BCM [Pub3].

Battery capacity	Charging/ discharging power	Charging/ discharging efficiency	Minimum safety SoC	Minimum SoC at departure	Location of bidirectional EVSE
50 kWh	11 kW	94%	30%	70%	At home

 Table 3-2:
 Technical parameters of bidirectional and smart EVs

In addition to the technical parameters of the EVs shown in Table 3-2, the economic parameters are also relevant. Table 3-3 shows the additional investment costs for bidirectional and smart EVs compared to unmanaged charging EVs. The investment costs are based on the costs presented in Section 2.3.1 and are also discussed in [Pub3]. They include additional costs for hardware, software, and installation of EV and EVSE. In addition, sensitivities for lower costs of smart EVs (*Sen1* and *Sen2*) were added to assess the impact of smart EVs on bidirectional EVs more precisely. Investment costs are annualized by an expected lifetime of EV and EVSE of 15 years [53] and an interest rate from the energy system's perspective of 3.5% [54].

Table 3-3:Additional investment costs of bidirectional and smart EVs compared to<br/>unmanaged charging EVs for base and sensitivity (SenX) scenarios

		2025	2030	2035	2040	2045	2050
Additional	<b>Bidirectional EVs</b>	2840	2190	1890	1590	1590	1590
investment costs for EV and EVSE in €	Smart EVs Base	960	760	760	760	760	760
	Smart EVs Sen1	480	380	380	380	380	380
	Smart EVs Sen2	200	160	160	160	160	160

Since smart and bidirectional EVs are integrated as independent elements in the ISAaR energy system model, the maximum share of the absolute number of EVs is set to 50% in each case. This ensures that the number of smart and bidirectional EVs does not exceed the absolute number of EVs. Overall, the following study scenarios result:

- **Ref:** Reference scenario with no smart and bidirectional EVs
- **BCM:** Based on reference scenario with option to integrate bidirectional EVs and smart EVs with base investment costs in Table 3-3
- **Sen1:** Based on reference scenario with option to integrate bidirectional EVs and smart EVs with Sen1 investment costs in Table 3-3
- **Sen2:** Based on reference scenario with option to integrate bidirectional EVs and smart EVs with Sen2 investment costs in Table 3-3

## 3.3 Future Cost-Optimal Penetration Rates

Building on the scenarios established in Section 3.2, a cost optimization of the future European energy system is performed for the *BCM* scenario and the sensitivities *Sen1* and *Sen2* with reduced investment costs for smart EVs. An important finding in [Pub3] was that, given the investment costs in the *BCM* scenario, the option of smart EVs is hardly

added endogenously by the model. The number of bidirectional EVs in the *BCM* scenario in Europe goes up from 19 million EVs in 2030 to 62 million EVs in 2050. This corresponds to a share of bidirectional EVs in total EVs of 25% in 2030 and of 30% in 2050. The share of bidirectional EVs in the total number of EVs is highest in 2040, at around 35%. The added value of bidirectional EVs for the European energy system compared to smart EVs is so much higher that their higher investment costs are compensated. On the one hand, this is due to the high availability of the EVs at home, so that the higher flexibility of bidirectional EVs through charging and discharging leads to added values for the energy system. On the other hand, smart EVs can only adapt their charging process, which limits their energy flexibility potential to their driving consumption.

With a reduction in investment costs for smart EVs in the *Sen1* and *Sen2* scenarios, significantly more smart EVs are endogenously integrated by the energy system model, as shown in Figure 3-3. Especially in scenario *Sen2*, the share of smart EVs in the absolute number of EVs is close to the maximum of 50% in 2030 and 2040. Consequently, the option of smart EVs leads to a sufficiently large added value for the energy system so that the additional investment costs can be compensated. However, compared to the BCM scenario, these additional smart EVs are used less to replace bidirectional EVs than previously unmanaged charging EVs. This shows that the flexibility option of smart EVs only slightly influences the integration of bidirectional EVs. Even in scenario *Sen2* with lowest costs for smart EVs, the number of bidirectional EV increases still robust, predicted to rise from 14 million EVs in 2030 to 54 million in 2050.





In addition to the absolute number of EVs per charging strategy, an analysis of the regional distribution of EVs is also informative in order to obtain indicators for factors influencing a high share of bidirectional EVs. For this purpose, Figure 3-4 presents the dependance of shares and numbers of EVs per charging strategy and country for the *BCM* and *Sen2* 

scenarios in 2050 on the full load hours of PV generation. This complements the analysis in [Pub3], which compares 2030 and 2050 for the *BCM* scenario. Various characteristics are recognizable. In Scandinavia, no bidirectional EVs are integrated in either scenario and only a quarter of smart EVs are integrated in the *Sen2* scenario. The shares of bidirectional EVs are higher in Southern Europe, especially in the *BCM* scenario, for example in Greece, Italy, Spain and Bulgaria. In the *Sen2* scenario, smart EVs displace a little part of the bidirectional EVs in almost all countries. Only France and Portugal stand out with a higher share of bidirectional EVs, although a large number of smart EVs are also integrated. This is due to a further significant increase in PV capacity. The European countries on the map are each colored based on the full load hours of PV systems. Here, a correlation can already be determined visually. A higher number of full load hours of PV plants tends to lead to a higher share of bidirectional EVs in total EVs.



Figure 3-4: Dependance of shares and numbers of EVs per charging strategy and country for the BCM (a) and Sen2 (b) scenarios in 2050 on the full load hours of PV generation based on [Pub3] and on further evaluations

To further investigate the influencing factor of PV energy and other influencing factors, Figure 3-5 presents dependencies of the share of bidirectional EVs on various influencing factors for all countries with a total of more than one million EVs. This evaluation refers to the *BCM* scenario in 2050. Each point in the diagram represents one country. In addition, a linear regression curve and the coefficient of determination R<sup>2</sup> of the linear regression were determined. The coefficient of determination, which always lies in the interval between 0 and 1, indicates the quality of the regression. The closer the coefficient of determination is to 1, the better the fit of the determined regression line. With a coefficient of determination of 1, all residuals are 0 [55].

The upper left plot in Figure 3-5 shows the dependence of the share of bidirectional EVs on the average full load hours of PV systems in a country to reinforce or refute the visual indication from Figure 3-4. A coefficient of determination R<sup>2</sup> of 0.518 is obtained, which indicates a certain linear correlation of the share of bidirectional EVs with the average full load hours of PV plants. Since bidirectional EVs act as storage in the energy system, they will often use the PV peaks that are above the load peaks to integrate this otherwise curtailed energy into the system. Therefore, the upper right diagram in Figure 3-5 shows the dependance of the share of bidirectional EVs to the ratio of PV peak power to peak load. This results in a coefficient of determination R<sup>2</sup> of 0.718, which indicates a good linear relationship. Another possible assumption is that bidirectional EVs tend to be more integrated when peak wind power in a country is significantly above peak load. Therefore, the diagram in the lower left of Figure 3-5 shows the relationship between the share of bidirectional EVs and the ratio of peak wind power to peak load. Here, however, the coefficient of determination R<sup>2</sup> is close to 0, which means no linear correlation. The bottom right diagram in Figure 3-5 brings together wind energy and PV energy in a final study and plots the relationship of the share of bidirectional EVs to the ratio of the peak of the summed wind and PV power to the peak load. The coefficient of determination R<sup>2</sup> of 0.237 indicates only a slight linear correlation. It is also significantly lower here than when considering the ratio of peak PV power to peak load in the second plot.



Figure 3-5: Dependance of shares of bidirectional EVs in total EVs on various influencing factors

Wind turbines have a less regular feed-in characteristic than PV systems, which always have their daily peak feed-in times around noon. This regular feed-in characteristic is well suited for daytime storages, which can charge during the day, store the energy and discharge at night. Even if the availability of bidirectional EVs is more limited during the day [56], at least 50% of all vehicles in a country are always at home. Bidirectional EVs therefore often

act as daytime storage and are thus a very good complement to PV energy. This and other added values will be taken up and discussed in the following Section 3.4.

## 3.4 Impact on the Future Energy System

In the previous section, it was already shown that bidirectional EVs with underlying investment and operating costs are integrated in a cost-optimal future energy system. In this section, the added values and their impact on the energy system are evaluated more concretely based on [Pub3], [Pub4] and further evaluations.

### 3.4.1 Other Assets in the Energy System

The evaluations listed below complement the evaluations in [Pub3] by comparing the reference scenario *Ref* without smart and bidirectional EVs with the *BCM* scenario and the *Sen2* scenario defined in Section 3.2. In this way, the influences of smart and bidirectional EVs on the energy system and other elements in the energy system can be quantified.

Figure 3-6 shows the installed capacities of variable renewable energies and thermal power plants for the scenarios *Ref, BCM* and *Sen2* for the years 2030, 2040 and 2050 in Europe. As already pointed out in [Pub3], the installed capacity of PV increases significantly in the *BCM* scenario compared to the *Ref* scenario, whereas the capacities of onshore wind and offshore wind decrease. A comparison of the *BCM* and *Sen2* scenarios reveals only minor differences. The *Sen2* scenario tends to integrate slightly less PV and wind offshore capacity, while the installed capacity of wind onshore increases slightly. The relative curtailment of PV energy decreases by 2 to 3 percentage points from 9% to 6% in 2030 and 10% to 8% in 2050 in the *BCM* scenario compared to the *Ref* scenario. The relative wind energy curtailment, on the contrary, remains at a similar level at 2% in 2030 and 4% in 2050 in both scenarios.

The differences in the installed capacity of thermal power plants in the *BCM* and *Sen2* scenarios are similarly small. In *Sen2*, slightly lower capacities of gas-fired and hydrogen-fired power plants are required compared to *BCM*. Therefore, in both scenarios, the bidirectional EVs in particular lead to significantly lower capacities of gas- and hydrogen-fired power plants than in the *Ref* scenario.

The significant cost reduction of smart EVs and the associated high integration of these therefore only has a minor impact on the expansion of variable renewable energies and thermal power plants. As already discussed in Section 3.3, the share of bidirectional EVs in the *Sen2* scenario is similar to the *BCM* scenario resulting in the same characteristics of the energy system. Very cheap smart EVs cause slightly reduced required capacities of thermal power plants, but the impact compared to bidirectional EVs is small.



Figure 3-6: Installed capacities of variable renewable energies (a) and thermal power plants (b) for the scenarios Ref, BCM and Sen2 for the years 2030, 2040 and 2050 in Europe based on [Pub3] and on further evaluations

Going further, Figure 3-7 shows the installed capacities of mobile and stationary storages for the scenarios *Ref, BCM* and *Sen2* for the years 2030, 2040 and 2050 in Europe. The storage capacities of the smart and bidirectional EVs result directly from the number of EVs multiplied by a storage capacity of 50 kWh. Since the storage capacity cannot be fully utilized due to unavailability of the vehicles and limitations of the SoC, it represents a theoretical storage capacity. The storage capacities of stationary battery storages and pumped storage hydropower plants do not differ in the *BCM* and *Sen2* scenarios. Pumped storage hydropower plants are fixed and not modeled as endogenous expansion assets. Stationary battery storages are decreased to a minimum, exogenously specified capacity in both the *BCM* and *Sen2* scenarios. Therefore, the integration of very low-cost smart EVs in the *Sen2* scenario does not lead to any changes beyond the reduction of stationary battery storages in the *BCM* scenario.





Installed capacities of mobile (a) and stationary (b) storages for the scenarios Ref, BCM and Sen2 for the years 2030, 2040 and 2050 in Europe based on [Pub3] and on further evaluations

#### 3.4.2 Total System Costs and Electricity Prices

The structural changes in the energy system in the *BCM* and *Sen2* scenarios also result in different total system costs and mean European electricity prices, which are summarized in Table 3-4 for the years 2025 to 2050. The *Sen2* scenario is only modeled for the years 2030, 2040 and 2050. In the *BCM* scenario, total system costs decrease by  $\in$ 8 billion/a in 2050 compared to the *Ref* scenario, although the integration of bidirectional EVs is modeled by additional investment costs. In the *Sen2* scenario, total system costs can be reduced by a further  $\in$ 3 billion/a in 2050 compared with the *Ref* scenario. More than  $\in$ 1 billion/a of the  $\in$ 3 billion/a cost reduction in the *Sen2* scenario compared to the *BCM* scenario can be attributed to the 8 million EVs reduction in bidirectional EVs. The average, consumption-weighted European electricity price on the wholesale market is 2-5  $\notin$ /MWh lower in both scenarios *BCM* and *Sen2* than in the *Ref* scenario. The *Sen2* scenario does not cause any significant change in electricity prices compared to the *BCM* scenario, since the total system costs consist of the infrastructure costs and the supply costs across all energy carriers, there is no direct correlation to the electricity prices.

As a complementary evaluation to the price analysis in [Pub3], Table 3-4 also presents the mean European daily standard deviation of the electricity prices in the three scenarios. It is calculated by the mean daily standard deviation of the electricity prices per country, which is then used to determine the consumption-weighted European mean. The impact of bidirectional EVs on the electricity price characteristics is evident. In the *BCM* scenario, the electricity price is significantly smoother, resulting in a daily standard deviation reduced by 6-11  $\notin$ /MWh compared to the *Ref* scenario. In the *Sen2* scenario, the daily standard deviation of electricity prices no more changes significantly compared to the *BCM* scenario. In Section 2.3.2, electricity prices for the German market area from the *BCM* scenario are used as simulated future electricity prices to estimate revenue potentials of specific users of bidirectional EVs. For Germany in the *BCM* scenario, a slightly higher price volatility with a mean daily standard deviation of 15.8  $\notin$ /MWh in 2025, 18  $\notin$ /MWh in 2030, 17.6  $\notin$ /MWh in 2030, and 16.8  $\notin$ /MWh in 2040 is observed.

Year	Overall energy system costs in billion €/a		M eleo	Mean European electricity price in €/MWh			Mean European daily standard deviation of electricity prices in €/MWh		
	Ref	BCM	Sen2	Ref	BCM	Sen2	Ref	BCM	Sen2
2025	432.4	431.7	-	44.6	42.6	-	21.2	15.3	-
2030	414.6	412.4	411.4	42.5	39.4	39.7	23.4	15.4	15
2035	354.9	350.1	-	43.1	39.1	-	26.8	16	-
2040	332.2	325.7	324.1	44.6	39.4	39.3	25.2	15.3	15.3
2045	323.5	314.4	-	44.4	41.1	-	25.5	17.7	-
2050	353.1	345.0	342.0	41.6	38.8	38.9	22.4	15.3	15.9

Table 3-4:Total energy system costs and mean European electricity price for the years 2025<br/>to 2050 for scenarios Ref, BCM and Sen2

#### 3.4.3 Regional Residual Load Characteristics

A much-discussed topic in the scientific community is the question of to what extent bidirectional EVs bring an additional load to the power grids, i.e., transmission grids and distribution grids [71], [57]. For the grid dimensioning of distribution grids in low voltage and medium voltage [58] as well as transmission grids in high and ultra-high voltage [72], both the peak load at low generation and the peak generation at low load can be relevant. Since the future energy system will be strongly characterized by variable renewable energies on the generation side, the maximum and minimum regional residual load, meaning the regional load minus the regional feed-in from renewable energies, is a meaningful indicator for the load of electricity grids in addition to detailed grid simulations. An increase of the maximum residual load peaks (peak load) or a decrease of the minimum negative residual loads (generation surpluses) thus signals a required higher grid utilization. This is only to be taken as an indicator, as the actual grid situation must be analyzed in each individual case. Nonetheless, smoothing the regional residual load and thus reducing load and generation peaks represents a reduction in the maximum loads and generation for which a grid must be designed. For this reason, this section supplements the evaluations in [Pub3] by an analysis of the impact of bidirectional EVs on regional residual load.

Figure 3-8 takes up the question of whether the residual load is smoothed by bidirectional EVs and first analyzes the Europe-wide residual load for the modeled years 2030, 2040, and 2050 for both scenarios *Ref* and *BCM* and the scenario *BCM without EVs*. The *BCM* scenario adds charging power and subtracts discharging power of bidirectional EVs to the residual load, which are excluded in the *BCM without EVs* scenario. The residual load is calculated by the static electricity load of the final energy sectors subtracted by the variable renewable energies generation. The European residual load is shown as an annual duration curve to better analyze the scenarios.

Over the years 2030 to 2050, the maximum residual load increases significantly, and the minimum residual load decreases considerably. On the one hand, this is due to increasing

electrification in the final energy sectors, which increases the electrical load. On the other hand, variable renewable energies are increasingly being integrated on the generation side, which in some cases cause large generation surpluses due to high simultaneities. When comparing the *Ref* scenario with *BCM without EVs*, it becomes apparent that significantly higher generation surpluses occur in the *BCM without EVs* scenario. The bidirectional EVs bring more flexibility into the system, which means that in the *BCM without EVs* scenario, more PV power with a high feed-in simultaneity is integrated, so that the peaks of the generation surpluses increase. This effect is not only compensated by bidirectional EVs, but the residual load is also even smoother in the *BCM* scenario than in the *Ref* scenario. Costoptimally operated bidirectional EVs charge at favorable electricity prices (often during generation surpluses) and discharge at high electricity prices (often during peak loads) and thus have a smoothing effect on the residual load.





At the European level, the smoothing of the residual load is a fundamentally positive effect, as less flexibility is needed to keep load and generation in balance. However, analyzing the European residual load does not allow any conclusions on grid loads. To show the change of the regional residual load, all European market areas are regionalized into NUTS 3 (Nomenclature of territorial units for statistics) regions [59]. NUTS regions are a classification of territorial units for statistics. The regionalization of the final energy demand as well as the PV and wind power plants is based on research project eXtremOS. The regionalization of bidirectional EVs is consistently based on the distribution of passenger cars in the final energy sector of transport. Detailed information on the regionalization can be found in [60] and [3]. This regionalization into NUTS 3 regions divides Germany, for example, into about 400 regions. As of 2022, there are 569 substations in Germany for transforming electricity from the transmission grid level (380/220 kV) to the high voltage level (110 kV) based on data from 50Hertz (62 substations [73]), Amprion (234 substations [74]), TenneT (189 substations [75]), and TransnetBW (84 substations [76]). This

means that a NUTS 3 region contains on average 1 to 1.5 substations. The analysis of the regional residual load under the implemented regionalization thus only allows for grid dimensioning indications on the transmission grid and partly on the high-voltage grid level since the high voltage grid is used for distribution as well as transmission [77]. Therefore, an indication for an evaluation of dimensioning lower voltage levels is not possible.

For each NUTS 3 region, the annual hourly residual load duration curve is created for both scenarios Ref and BCM. For an analysis of the change in peak load, the 100 hours with the maximum residual load of a region are extracted and the average of these values is calculated. In the same way, for an evaluation of the generation surpluses based on the 100 hours with the lowest residual load, the average value is calculated. Since the residual load can be negative or positive, but a grid load exists in both directions, the absolute value is taken in each case. To compare the peak load of both BCM and Ref scenarios, Equation (3-9) calculates a factor factor peak load by taking the average of the absolute maximum residual loads in the BCM scenario  $|resload_{BCM,max100h}|$  minus the absolute maximum residual loads in the Ref scenario resload<sub>Ref.max100h</sub>. In the same way, in Equation (3-10), a factor *factor<sub>gen\_surplus</sub>* is formed to evaluate the change in generation surpluses. It is important to note that there are also regions that have only positive residual loads throughout the year, e.g., urban regions. Consequently, these regions have no generation surpluses. In these cases, the minimum residual load refers to the minimum grid loads. However, due to the difference in the absolute values of the residual loads of the two scenarios, positive and negative minimum residual loads can also be compared. Consequently, if the factors are negative, it implies a reduction in grid loads due to peak loads or generation surpluses in the BCM scenario compared to the Ref scenario. If the factors are positive, the loads are lower in the *Ref* scenario.

$$\overline{|resload_{BCM,max100h}|} - \overline{|resload_{Ref,max100h}|} = factor_{peak\_load}$$
(3-9)

$$\overline{resload_{BCM,min100h}|} - \overline{|resload_{Ref,min100h}|} = factor_{gen\_surplus}$$
(3-10)

The factors per NUTS 3 region are visualized in Figure 3-9. The left map plots the change in peak load per NUTS 3 region. Green tones indicate a decrease and red tones an increase of the peak load in the *BCM* scenario. It can be observed that the regional peak load in many regions in Europe is reduced by the integration of bidirectional EVs. However, especially in urban regions such as Madrid, Paris or Berlin, the peak load increases. This is because a high number of bidirectional EVs are integrated there due to the high population, which overlays the load characteristics of the regions in times of high charging power. The right map of Figure 3-9 shows the change in generation surpluses. Here, a much more heterogeneous picture emerges, which in some regions is exactly opposite to the evaluation of the peak loads. For urban regions, this can be explained by the fact that the residual load is often positive throughout the year. Bidirectional EVs overlay this load characteristic and cause a reduction of the absolute residual load in times with high discharge powers. In many other regions, e.g., in Spain and France, higher negative residual loads and thus generation surpluses occur. These are often regions where significantly more PV is integrated in the *BCM* scenario compared to the *Ref* Scenario, so that PV peak generation

increases. In contrast, relatively few bidirectional EVs are integrated in these often-rural regions, so that a compensation of the higher PV generation by charging the bidirectional EVs cannot take place. In other regions, however, such as southern Germany, the highest generation surpluses correlate with high charging powers of bidirectional EVs, reducing generation surpluses. In Germany, it should also be noted that the PV expansion for the *Ref* and *BCM* scenarios in 2030 was fixed based on the targets of the German government.





Since both generation peaks and load peaks are relevant for the design of electricity grids, the two factors are combined in Equation (3-11). The factor *factor*<sub>grid\_dim</sub> is calculated by the maximum of the absolute maximum and minimum residual loads in the *BCM* scenario minus the maximum of the absolute maximum and minimum residual loads in the *Ref* scenario. Consequently, the factor reflects the load on the distribution grids, regardless of whether they are generation or load peaks.

$$factor_{grid\_dim} = \max\left(\overline{|resload_{BCM,max100h}|}, \overline{|resload_{BCM,min100h}|}\right) - \max\left(\overline{|resload_{Ref,max100h}|}, \overline{|resload_{Ref,min100h}|}\right)$$
(3-11)

Figure 3-10 shows these factors indicating a change in grid dimensioning per NUTS 3 region in Europe in 2030 and 2050. When comparing the changes in generation surpluses and peak loads with the grid dimensioning factor for the year 2030, it is noticeable that in Southern Europe the change in generation surpluses accounts for the grid dimensioning in most regions. In Central and Northern Europe, the picture is heterogeneous with regions showing a reduced grid dimensioning factor and regions with higher grid loads. Overall, 770 regions have a negative grid dimensioning factor, and 610 regions have a positive grid dimensioning factor in 2030. It can be concluded that bidirectional EVs basically have a very different impact on regional residual load maxima and minima

depending on regional conditions. There is no clear direction on whether integrating bidirectional EVs will increase or decrease the grid load in transmission and high voltage grids.



Figure 3-10: Factors indicating change in grid dimensioning per NUTS 3 region in Europe in 2030 (a) and 2050 (b) in scenario BCM compared to Ref

The comparison of the grid dimensioning factor for the years 2030 and 2050 also shows interesting changes. In most regions in Southern Europe, the grid dimensioning factor changes from positive values to negative values. This means that the integration of bidirectional EVs has a relieving effect on grid utilization there. This is due to the PV expansion there, which is at the maximum potential limit in both scenarios BCM and Ref. Since no additional PV capacity is added in the BCM scenario compared to the Ref scenario, the bidirectional EVs have a mostly positive smoothing effect on the residual load. This is different in Poland, the Czech Republic and France, for example. In France, the PV capacity in the BCM scenario in 2050 is 420 GW compared to 290 GW in the Ref scenario. The PV peaks are not completely smoothed by the bidirectional EVs, so that in most regions the generation surpluses are significantly higher regionally and thus the grid load increases. In Germany and Benelux, the level of grid dimensioning factors varies a lot. There, very high installed capacities of wind offshore and wind onshore have been integrated by 2050, which also leads to generation surpluses in the market areas in many hours. Since this stimulates the charging of bidirectional EVs in exactly these hours, the bidirectional EVs in northern Germany and Benelux often have a grid-relieving effect. In other regions, where PV and load are more relevant to grid dimensioning, the impact of bidirectional EVs is more heterogeneous. Overall, 590 regions have a negative grid dimensioning factor, and 780 regions have a positive grid dimensioning factor in 2050. Consequently, even in 2050, no clear statement can be made as to whether bidirectional EVs will reduce or increase grid loads when integrated into electricity markets. In individual cases, the regional load and generation conditions must be analyzed for each region or grid area.

### 3.4.4 Congestion Management

After the investigations of regional residual load described in the previous section indicate that bidirectional EVs have partly positive and partly negative impacts on the electricity grids, the extent to which they can provide congestion management in the transmission grid is now investigated in more detail. Since the grid simulations with optimization of congestion management as described in Section 3.1.2 are very complex, the study area is restricted to the year 2030. The studies presented here are based on [Pub4], which also describes the scenario framework of the European transmission grid in more detail. A comparison of two transmission grid simulations is performed:

- **Ref grid run:** Transmission grid run with optimization of congestion management without bidirectional EVs
- **BCM grid run:** Transmission grid run with optimization of congestion management with bidirectional EVs

Both transmission grid optimizations are based on the market run of the BCM scenario.

Congestion management volumes in both runs increase sharply from today's volumes. In the *Ref grid run*, the volumes of positive congestion management in Europe for the year 2030 are 600 TWh compared to a total electricity production of 5,000 TWh. The German positive congestion management volume of 160 TWh in the *Ref grid run* indicates high amounts of congestion management that do not fit to the volumes simulated in the German transmission grid expansion plan of around 7 TWh in 2035 in the base scenario. Reasons for this are discussed in [Pub4]. They include the lack of modeling of overhead line monitoring, uncertain regionalization of added assets, and simplified modeling of a linearized load flow. Furthermore, the endogenously cost-optimized scenario as well as the parameterization of the assets in the energy system is also not the same as the scenario used in grid expansion plans.

Nevertheless, by comparing the *BCM grid run* and *Ref grid run*, conclusions can be drawn about the added value of bidirectional EVs in congestion management. The use of bidirectional EVs in congestion management reduces 26 TWh positive redispatch of thermal power plants, 9 TWh negative redispatch of thermal power plants and 23 TWh curtailment of renewable energies. This results in 17 TWh less electricity generated by thermal power plants, saving 12 million tons of CO<sub>2</sub>. Compared to total modeled emissions in the energy sector in 2030 of just under 500 million tons of CO<sub>2</sub>, this is a 2.5% reduction in emissions.

Even though the absolute reduced numbers of congestion management by power plants and renewable energies are favored by the high total volumes of congestion management, the positive effect of bidirectional EVs is evident. Figure 3-11 shows the regional distribution of positive and negative congestion management per technology per transmission grid node. The decentralized distribution of bidirectional EVs proves to be an advantage. Bidirectional EVs near nodes with high curtailments of renewable energies (for example in northern Germany) can reduce curtailments through negative congestion management. Positive congestion management then re-establishes their initial schedule within the time slice. In southern Germany, on the other hand, positive congestion management is often used first, thus displacing the use of thermal power plants. Here, the schedule must then be re-established at other times through negative congestion management.

In [Pub4], the regional distribution of congestion management per technology for the *Ref grid run* as well as the mean line usage of both runs are also shown and analyzed. Overall, many transmission grid lines have a very high utilization. However, especially in the case of wind offshore plants, the curtailment often takes place immediately after the grid connection point.



Figure 3-11: Negative (a) and positive (b) congestion management per grid node split up in different technologies for the BCM grid run

Overall, in addition to the added values of bidirectional EVs shown in the previous sections regarding the design of the future energy system, a potential added value for the provision of congestion management can also be shown. Ultimately, however, more in-depth analyses with a parameterization of the transmission grid that is suitable for the underlying market scenario are required.

## 4 Combining the User and the Energy System Perspective on Bidirectionally Chargeable Electric Vehicles

In a final step, the user and energy system views on bidirectional EVs are brought together. Section 4.1 compares the modeling approaches of bidirectional EVs developed in this dissertation. Section 4.2 discusses a common added value of bidirectional EVs from the user and the energy system perspective.

# 4.1 Comparison of Modeling Approaches Depending on Use Case and Perspective

As shown in the previous chapters, the modeling approaches of bidirectional EVs differ depending on the focus of investigation. For this purpose, Table 4-1 provides a clear comparison of the modeling approaches of bidirectional EVs presented in this dissertation.

All four presented types of modeling bidirectional EVs formulate an optimization problem. Depending on the perspective, the objective function focuses on minimizing the electricity costs of a user or the total systemic costs for energy provision or congestion management. For the user perspective, minimizing electricity costs could also be reformulated as maximizing revenues. The objective function values can become negative values and thus mean real revenues.

The basic modeling does not differ in all approaches. A bidirectional EV is modeled as a storage with limited availability and electrical consumption, which has typical storage constraints such as energy conservation, and EV user-specific constraints such as maintaining a minimum safety SoC and a departure SoC.

In addition, all modeling approaches presented have special features that are only relevant in the focus of investigation under consideration and thus lead to advantages and disadvantages of the respective approach:

- For the modeling of PV self-consumption optimization, variable charge and discharge efficiencies are modeled for a wide range of household types, since power flows in the household are small and losses would otherwise be greatly underestimated. The modeling approach leads to more realistic results but adds a higher complexity to the optimization problem, since it is no longer linear but mixed-integer linear.
- For the modeling of arbitrage trading from the user perspective, a rolling-horizon optimization with consideration of real trading hours is implemented to allow consecutive spot market trading in day-ahead and intraday markets. This consecutive market trading results in higher revenue potentials than single market trading and thus shows the potential revenues of the use case arbitrage trading.

The complexity of the optimization problem can be kept small by rolling optimization. However, no feedback effects on electricity prices are modeled, so that e.g., low liquidity of the intraday market indicates less robust revenue potentials.

- For the modeling of arbitrage trading from the energy system perspective, the aggregation of EV profiles and thus a non-discrete modeling of the constraints enables investigations on cost-optimal integration of bidirectional EVs with only a slight increase in the complexity of the optimization problem. The impact of the bidirectional EVs integrated into the electricity market on the day-ahead market prices result endogenously from the optimized operation of the EVs. The cost-optimal future penetration rates of bidirectional EVs can be used for policy recommendations.
- For the modeling of the provision of congestion management via bidirectional EVs, a rolling parallelized grid optimization run with simplified time-coupling of storage is performed based on a market run. In both runs, bidirectional EVs are modeled by aggregated EV profiles. Although the optimization problem of the transmission grid calculation is very complex, it is computable for a time slice length of one week by modeling of one aggregated instance per transmission grid node. This allows realistic potentials of congestion management to be determined.

Table 4-1 further shows the publications and sections associated with the modeling approaches in this cumulative dissertation. Overall, the various approaches to modeling bidirectional EVs have fundamental similarities, but also necessary differences. The specifics per use case and perspective can be transferred well to other use cases of bidirectional charging. Consequently, other modelers can benefit from the overview of the different approaches to modeling bidirectional EVs to adapt their modeling according to the use cases and perspective.

Table 4-1:	Comparison of dissertation	modeling approad	hes of bidirectional	EVs presented in this				
Use Case	Vehicle-to- Home: PV self- consumption optimization	Vehicle-to-Grid: Arbitrage trading	Vehicle-to-Grid: Arbitrage trading	Vehicle-to-Grid: Provision of congestion management				
Perspective	User	User	Energy system	Energy system				
Objective function	Minimization of household electricity costs	Minimization of EV's electricity costs	Minimization of total energy system costs	Minimization of congestion management costs				
Base modeling approach of	<ul> <li>Modeling of bid limited availabil</li> <li>Time-coupling</li> </ul>	directional EVs as a lity via equation for sto	storage with energy c	onsumption and tion				
bidirectional EVs	<ul> <li>Modeling of minimum safety SoC and minimum SoC at departure for realistic user behavior</li> </ul>							
	<ul> <li>Basic constraint absolute SoC</li> <li>EV profiles of E<sup>1</sup></li> </ul>	s, like maximum ch V consumption and	arging/discharging pc EV location	ower and maximum				
Modeling specifics	<ul> <li>Modeling of varying charging and discharging efficiencies</li> <li>Modeling of various household configurations</li> </ul>	<ul> <li>Modeling of consecutive spot market trading</li> <li>Rolling optimization with modeling of real trading hours</li> </ul>	<ul> <li>Modeling of cost- optimized integration of bidirectional EVs</li> <li>Aggregation of discrete EV profiles</li> </ul>	<ul> <li>Aggregation of discrete EV profiles</li> <li>Rolling grid optimization with simplified time-coupling of storage</li> </ul>				
Advantages of modeling specifics	<ul> <li>Realistic modeling of EV and EVSE losses leads to realistic estimation of revenue potentials</li> </ul>	<ul> <li>Revenue optimization by multiple market trading</li> <li>Limited complexity via rolling optimization</li> </ul>	<ul> <li>Only slight increase of model complexity</li> <li>Cost-optimal penetration rates of bidirectional EVs for policy recommendations</li> </ul>	<ul> <li>Optimization problem still computable despite high complexity</li> <li>Determination of realistic potentials of congestion management</li> </ul>				
Disadvantages of modeling specifics	<ul> <li>Increased complexity due to variable charging and</li> </ul>	<ul> <li>No feedback of arbitrage trading on electricity</li> </ul>	<ul> <li>No exact representation of EV constraints</li> <li>Limited on day</li> </ul>	<ul> <li>No exact representation of EV constraints</li> <li>Highly complex</li> </ul>				
	discharging	prices modeled	ahead market	optimization problem				
Sections in Dissertation	2.1.1 2.1.3	2.1.2 2.1.3	3.1.1	3.1.2				
Publication	[Pub1]	[Pub2]	[Pub3]	[Pub4]				

#### 4.2 Possibility of a Common Added Value

After multiple added values of bidirectional charging have been shown in this dissertation, the question arises as to what extent the added values for the user and the energy system can be combined.

The use case V2G arbitrage trading has been analyzed from a user and energy system perspective and shows promising added values for both perspectives. The user can generate significant revenues by participating in the day-ahead and intraday markets. These revenue potentials are above 1000 €/EV/a, but strongly depend on the EV and EVSE configuration, the user behavior and especially on the regulatory framework with regard to taxes, levies and grid charges to be paid. The energy system benefits from the integration of bidirectional EVs into the electricity markets on many levels. On the one hand, the flexibility of bidirectional EVs increases the market value of PV energy, allowing it to be better integrated. On the other hand, less capacity of thermal power plants and other storage technologies is needed. This ultimately decreases the total costs of the energy system thus becomes more efficient. The significant systemic benefit also generates the added value for the user of contributing to a cost-optimal energy system based on renewable energies. This is only an indirect, qualitative added value, but it can be relevant for the acceptance of the use case V2G arbitrage trading.

The added value from both the energy system and the user perspective is not directly evident for the V2H PV self-consumption optimization use case. The economic added value for the user was shown in Sections 2.2.1 and 2.3.2. Further, Section 2.2.1 also points out that the self-consumption of renewable PV generation and thus the household self-sufficiency increases. The profitability of the use case gives users the incentive to invest in a bidirectional EV and EVSE. The added value from the system perspective is open to question since PV self-consumption optimization is a local optimization without considering the effects on the German or European energy system. Modeling of local prosumer cells in the energy system performing PV self-consumption optimization by bidirectional EVs poses another research need here. Nevertheless, analysis of renewable energy curtailment can provide conclusions about a systemic effect of PV self-consumption optimization. In total, by the market integration of bidirectional EVs in the energy system in 2030, the relative PV curtailment in Europe reduces by 2 to 3 percentage points. Since PV self-consumption optimization also incentivizes charging of EVs at daytime, on the one hand, a reduction in PV curtailment and thus a positive systemic effect can be expected in summer. In winter, on the contrary, when there is an increased curtailment of wind energy instead of PV energy, PV self-consumption optimization may even have a negative effect.

The provision of congestion management primarily has a systemic benefit: thermal power plants can be operated less and renewable energies less curtailed. The additional revenues for the user through participation in congestion management are very difficult to quantify under the current design in Germany, as there is a cost-based remuneration [42]. The costs of bidirectional EVs to participate in congestion management consist of battery aging costs, aggregator costs, and other cost components, while the user's behavior may be restricted.
In addition, cost-based remuneration does not result in any real added value for the user. The introduction of market-based congestion management, which is being discussed in various ways [61], would open up new revenue opportunities for the use case. Without such market-based revenue opportunities, it is very unlikely that EV users would voluntarily participate in congestion management.

Table 4-2 ultimately summarizes these pointed out added values of bidirectional charging from the user and energy system perspectives. Overall, V2G arbitrage trading in particular can generate a wide range of systemic and user-related added value.

Use Case	Added Value User	Added Value Energy System		
V2H PV self- consumption optimization	<ul> <li>Revenues</li> <li>Increase in household self-sufficiency</li> <li>Increase in self-consumption of renewable PV generation</li> </ul>	<ul> <li>Indirect positive effects by decreasing PV curtailment in summer possible</li> <li>Unknown effect on renewable energy curtailment in winter</li> </ul>		
V2G arbitrage trading	<ul> <li>Revenues</li> <li>Contribution to cost- optimal energy system</li> </ul>	<ul> <li>Reduced capacities of power plants and battery storages</li> <li>Reduced energy system costs</li> <li>Improved market integration of PV</li> </ul>		
V2G congestion management provision	<ul> <li>In Germany only cost- based remuneration</li> <li>(×) Contribution to cost- optimal ancillary service provision</li> </ul>	<ul> <li>Less electricity generation of power plants needed</li> <li>Less curtailment of renewable energies</li> </ul>		

Table 4-2:Main added values for the user and the energy system by considered use cases<br/>in this dissertation

The combination of different use cases of bidirectional EVs can also add combined value from the user and system perspective. The possibility of combining V2H PV self-consumption optimization and V2G arbitrage trading has already been discussed in [Pub2] and in Section 2.2.3. PV self-consumption optimization exhibits strong seasonality. In the summer months from April to September, the user can expect significantly higher revenue potentials than in the winter months. Here, a simple seasonal switch of the use cases suggests operating V2G arbitrage trading in the winter months and V2H PV self-consumption optimization in the summer months. In the long term, a highly dynamic change of use cases every minute or hour will also be of interest in order to optimize the revenue potentials for the user and the added values for the energy system. In [Pub2], however, the high requirements for metering data for use case switching are already discussed.

In Germany, the 'Law on Immediate Measures for Accelerated Expansion of Renewable Energies and Further Measures in the Electricity Sector' was passed in the summer of 2022 [62]. Here, a balancing of the charging and discharging via a charging point is specified to enable V2G. The volumes of electricity drawn from the grid and simultaneously charged by an EV via the charging point are considered as consumption at the charging point. The volumes of electricity discharged by EVs via the charging point and simultaneously fed into the grid are regarded as generation at the charging point. This generation and consumption from the point of view of the charging point are netted out on a calendar year basis. The levies and surcharges are reduced on the balanced charged electricity. Consequently, measured values must be collected both at the grid connection point and at the charging point. This robust approach of annual netting the input and output volumes at the charging point would mean that even the simultaneous implementation of V2H and V2G would be possible. However, this would still allow the potential misuse addressed in [Pub2]. During the summer months, a user could initially charge PV energy into the EV and later discharge this electricity from the EV into the grid. In the winter months, the user could charge electricity from the grid into the EV. Netting would exempt this charged energy from levies and surcharges. This could result in higher revenues than PV energy fed directly into the grid during the summer months. This example shows that there are still open regulatory questions for a V2H and V2G use case combination regarding the interaction between the household with its various components and the bidirectional EV.

Since provision of congestion management for the transmission grid is a short-term application that takes place after day-ahead market and intraday auction trading, the use cases arbitrage trading and provision of congestion management could be combined very well. However, compared to continuous intraday trading, which is also short-term, a competitive situation arises that can prevent simultaneous participation in both use cases. This must also be considered when combining V2H PV self-consumption optimization and V2G congestion management provision. A change in the operation of bidirectional EVs through the provision of congestion management directly affects the PV self-consumption of a household and thus the associated revenues.

## 5 Conclusion, Discussion and Outlook

In this cumulative dissertation, different approaches for modeling bidirectional EVs in different use cases and perspectives were developed and their advantages and disadvantages were discussed. Based on this, the added values for the user and the energy system were shown. This chapter is divided into two parts: Section 5.1 answers the research questions raised in the introduction. Section 5.2 then reflects on the modeling and results, provides concluding remarks, and identifies further research needs.

#### 5.1 Answers to Research Questions

The following are concise answers to the research questions of this dissertation. Research questions 1 to 3 refer to the user perspective, whereas research questions 4 to 6 address the energy system perspective. Research question 7 deals with the combined perspective of the user and the energy system.

## 1. What modeling specifications are required for the evaluation of revenue potentials of vehicle-to-home and vehicle-to-grid use cases?

The modeling of the analyzed V2H and V2G use cases shows fundamental similarities but also significant differences in order to realistically evaluate the revenue potentials. All developed modeling approaches are based on a modeling of the bidirectional EV as a storage element with electrical consumption and limited availability. By comparing the objective function values with the objective function values of an unmanaged charging EV, the revenues for a user can be quantified.

For V2H PV self-consumption optimization, the condition of low-power load flows results in the requirement to set up the optimization problem as a MILP with variable charging and discharging efficiencies of the bidirectional EV. This leads to higher complexity but much more realistic revenues, which are otherwise overestimated by over 30%. Furthermore, the household is modeled with a wide variety of components, which leads to a large number of sensitivities. The modeling of V2G arbitrage trading is limited to the bidirectional EV and EVSE and their interaction with the electricity grid. Here, rolling optimization enables modeling of consecutive trading in the day-ahead and intraday markets with respect to real trading times.

## 2. What are the revenue potentials and their most important influencing factors for vehicle-to-home and vehicle-to-grid use cases?

The revenue potentials of V2H and V2G use cases are very sensitive depending on a variety of influencing factors. For V2H PV self-consumption optimization, typical revenues for a bidirectional EV compared to an unmanaged charging EV are around 300  $\in$ /a for a medium household. Two thirds of this revenue can already be generated by smart charging. The most important influencing factor is the difference between the household electricity price and the PV feed-in tariff. The higher the spread, the higher the revenue potential. In a scenario with a parameterization for maximum revenues, the revenues go up to 1,300 €/a. Other important influencing factors on revenues include PV system size and electrical household demand characteristic and level.

The revenues from V2G arbitrage trading are highly dependent on the levies and charges on charged, temporarily stored electricity. Typical revenue potentials with today's regulation in Germany and participation in the day-ahead market are around 200 €/year for a bidirectional EV compared to an unmanaged charging EV. However, increased price volatilities as a consequence of the energy crisis [63], a consecutive marketing on the day-ahead and intraday market, as well as a change of the EV and EVSE characteristics can lead to much higher revenues.

## 3. What is the economic profitability of vehicle-to-home and vehicle-to-grid use cases considering the additional investment costs?

The additional investment costs for bidirectional charging will see a strong cost degression in the coming years. Only then, V2H and V2G use cases can become profitable from the user's perspective. As with other new technologies, government subsidies may be necessary to support the ramp-up of the technology.

V2H PV self-consumption optimization turns out to be the more robust use case, which in a medium base scenario with an investment in 2025 has already amortized after seven years. V2G arbitrage trading achieves payback only after 12 years for an investment in 2025. Both use cases become more profitable in later years, as costs fall and revenues tend to rise, so that later investments pay off more quickly. With an investment in 2030, V2H amortizes after 5 years with a net present value after 15 years of 2,800  $\in$ , whereas V2G amortizes after 7 years with a net present value after 15 years of 1,650  $\in$ .

# 4. How do modeling specifications of bidirectional EVs need to be adapted to evaluate the European energy system perspective compared to modeling discrete EVs?

Complex European energy systems lead to the requirement of strongly clustering or aggregating bidirectional EVs rather than modeling them via discrete instances. The aggregation of all bidirectional EVs per country leads to a good representation of the vehicle fleet per country and its feedback on the energy system, while keeping complexity as low as possible. Here, the European market simulation with the objective function of minimizing the total costs leads to a similar operation of the bidirectional EVs, as in the V2G day-ahead market arbitrage trading use case from the user perspective.

Congestion management modeling in the European transmission grid requires even more the aggregation of all bidirectional EVs per transmission grid node. The combination of the time-coupled constraints of bidirectional EVs and the complex load flow calculations in the transmission grid strongly increases the complexity of the optimization problem compared to the European market simulation. A partition of the hourly annual calculation into weekly time slices leads to a sufficiently large flexibility of the bidirectional EVs while ensuring computability.

## 5. How high is the optimal share of bidirectional EVs in the total EVs from a system cost perspective?

Modeling the future energy system considering the additional investment costs of bidirectional EVs compared to unmanaged charging EVs results in system-cost-optimal penetration rates. The number of bidirectional EVs increases from just under 20 million EVs in 2030 to over 60 million EVs in 2050 in a baseline scenario. This corresponds to a share of bidirectional EVs in total EVs of 25% in 2030 and of 30% in 2050.

Provided that the additional investment costs of smart EVs compared to unmanaged charging EVs fall very sharply, smart EVs will also be integrated into the energy system in high numbers. However, this additional flexibility in the energy system has only a minor impact on the number of bidirectional EVs. Through these sensitivity considerations, it was found that bidirectional EVs are very robustly integrated into the cost-optimal future energy system.

# 6. What is the added value of bidirectional EVs integrated in the electricity markets and in congestion management in the transmission grid from an energy system perspective?

The integration of bidirectional EVs into the future energy system creates multiple added values:

- The energy system requires less capacities of thermal power plants (especially gas-fired and hydrogen-fired power plants) and other storage technologies. Bidirectional EVs provide a contribution to cover the electrical peak loads in the system.
- Bidirectional EVs often act as daytime storage, charging during the day and discharging at night. This increases the market value of PV and promotes PV integration. Since PV energy is more widely accepted and cheaper than wind energy in many countries, this encourages renewable energy expansion.
- The previous points result in significantly lower European energy system costs, by up to €9 billion/a.
- The provision of congestion management in the European transmission grid by bidirectional EVs leads to reduced use of thermal power plants and reduced curtailment of renewable energies. As a result, less CO<sub>2</sub> is emitted.

## 7. How can use cases of bidirectional EVs create combined added value from the user and the energy system perspective?

Use cases of bidirectional charging can generate simultaneous added value for the user and the energy system. By investigating the use case arbitrage trading from the user's and the system's point of view, this dissertation was able to show that the added value here is given for the user and the energy system in the future. PV self-consumption optimization primarily gives monetary and environmental added value to the user. The impact of this use case on the energy system can be positive and negative. The provision of congestion management in the transmission grid, on the contrary, currently generates hardly any added value for the user in Germany, as remuneration is cost-based. Here, the focus is on the systemic added value of reducing costs and CO<sub>2</sub> emissions.

Further, the combination of use cases can generate a combined added value for the user and the energy system. Especially V2H PV self-consumption optimization is very suitable for a combination with V2G use cases since it generates the revenues mainly in summer times and can be switched to other use cases in winter.

#### 5.2 Critical Reflection, Concluding Comments and Further Research

The technology of bidirectional charging of EVs is still a new technology that is yet to experience its market take-off. Currently in 2022, for example, there is still only one bidirectional EVSE available on the German market, which has very high investment costs [78]. Few automotive manufacturers currently offer bidirectional EVs, even though many have clearly positioned themselves positively toward the technology [44]. In addition to the technological side of EVSE and EV manufacturers, the regulatory side also plays an important role. Although regulation in Germany has taken important steps, many regulatory issues are still open and will need to be addressed by policymakers in the coming years. Consequently, bidirectional charging technology is still in its early stages. The analyses of this work thus offer an important piece to support the ramp-up of the technology and thus to create added value for the user and system.

Over a time period of four years, this dissertation has been created. During this period, many energy-economic changes and regulatory adjustments have occurred that have been directly referred to or referenced in this dissertation. These changes and adjustments in Germany have a direct impact on bidirectional charging of EVs:

• The first to be mentioned is the **energy crisis**, which has caused fuel and electricity prices to rise sharply worldwide. Since the end of 2021, fuel prices and subsequently also electricity prices have risen very sharply in the wake of Covid-19. This effect was further intensified by the Ukraine invasion in February 2022. The energy crisis has multiple implications for bidirectional EVs. In addition to potentially higher investment costs for users for bidirectional EVs and EVSEs due to increases in a wide range of commodity prices, there may also be multiple impacts on the use cases of bidirectional charging.

The high electricity price level causes a sharp increase in household electricity prices. This effect was directly addressed in Section 2.2.1, highlighting that this has a very positive impact on the revenue potentials of the PV self-consumption optimization use case. In addition to the electricity price level, the volatility of electricity prices has also risen very sharply. This was also directly addressed in Section 2.2.2. The increase in electricity price volatility results in significantly higher revenue potentials for the arbitrage trading use case. The current energy crisis has thus significantly increased the attractiveness of these two use cases.

- As a result of the energy crisis, the medium-term **climate targets** for the year 2030 were tightened in Europe and Germany. For example, Germany has set targets of 215 GW of PV energy and 115 GW of onshore wind energy for the addition of variable renewable energies [62]. As a result, the systemic need for flexibilities such as bidirectional EVs is increasing as electricity generation becomes more fluctuating. Furthermore, this transformation of the energy system leads to further increased electricity price volatility, which is also attractive for the use case arbitrage trading from a user perspective.
- The **regulatory framework** regarding bidirectional EVs has also evolved in recent years. The highlighting of dependencies of the added values of bidirectional charging for the user on the regulatory framework in [Pub1] and [Pub2] has given important impulses for regulatory adjustments. The German 'Law on Immediate Measures for an Accelerated Expansion of Renewable Energies and Further Measures in the Electricity Sector' defines the bidirectional charging point of EVs as a storage facility to allow an exemption from levies and charges as for a stationary storage facility. This first step does not include the power losses during the charging and discharging process, nor does it include all levies, surcharges, and taxes. Further regulatory steps must therefore be taken here to make the use case of arbitrage trading more attractive from the user's point of view.

Due to the transformation process of the energy system towards a climate-neutral system as well as the current Ukraine crisis, there are various short-term adjustments of fundamental energy-economic conditions. For example, the sharp rise in fuel prices and, in particular, the price of gas led to higher electricity prices and higher electricity price volatility [63]. Therefore, some of the scenarios presented in this dissertation are no longer valid. However, the fundamental assessments of bidirectional EVs still hold through the numerous sensitivity calculations.

From a scientific point of view, this dissertation shows further need for research regarding the technology of bidirectional charging. In the BCM project, a total of 14 use cases were defined that could have revenue potentials from the user's perspective. The approaches developed in this dissertation for modeling bidirectional EVs can be transferred for modeling other use cases, although the specifics may have to be adapted. Only by modeling and combining all these use cases, bidirectional EVs can be evaluated comprehensively from the user's point of view. Furthermore, a comparison of current control strategies in field tests with the mode of operation in the developed optimization models can show to what extent the revenue potentials can be achieved. In addition, modeling the user-optimized behind-the-meter use cases endogenously in an energy system model could directly quantify the impact of these use cases on the energy system. However, this step may involve a high level of complexity. Finally, the studies of congestion management provision by bidirectional EVs have shown initial positive potential but should be detailed by further scenarios. In this way, recommendations for action in the direction of market-based congestion management can ultimately be supported in order to better integrate decentralized technologies into congestion management.

The added values of bidirectional charging of EVs, which were investigated by models in this dissertation, have also been tested in pilot operation since 2021 as part of the BCM research project. For this purpose, 50 BMW i3 were equipped with the technology of bidirectional charging [64]. A comparison of the realized revenues for the user with the simulated revenue potentials at the end of the project can show here the extent to which there is still potential for improvement in, for example, the control system or user behavior.

In summary, this dissertation has demonstrated added values of bidirectional charging of EVs from a user and system perspective and developed different approaches to modeling for this purpose. Some of the results have already been integrated into a policy paper [65]. This shows recommendations for policymakers, users of EVs, and energy entrepreneurs to make well-informed decisions regarding the technology of bidirectional charging of EVs.

## 6 Bibliography

- [1] Regulation (EU) 2021/1119 of the European Parliament and of the Council of 30 June 2021 establishing the framework for achieving climate neutrality and amending Regulations (EC) No 401/2009 and (EU) 2018/1999 ('European Climate Law'). Brussels: Official Journal of the European Union, 2021.
- [2] Bundes-Klimaschutzgesetz (KSG). Ausgefertigt am 2019-12-12, Version vom 2021-07-07; Bonn: BMU, 2021.
- [3] Guminski, Andrej et al.: eXtremOS Summary Report Modeling Kit and Scenarios for Pathways Towards a Climate Neutral Europe. Munich: FfE, 2021.
- [4] Treibhausgasemissionen stiegen 2021 um 4,5 Prozent. Berlin: Bundesministerium für Wirtschaft und Klimaschutz (BMWK), 2022.
- [5] Förderung von elektrisch betriebenen Fahrzeugen Merkblatt für Anträge nach der Richtlinie zur Förderung des Absatzes von elektrisch betriebenen Fahrzeugen (Umweltbonus) vom 21. Oktober 2020 in der Fassung der Änderung vom 24. November 2021. Berlin: Bundesamt für Wirtschaft und Ausfuhrkontrolle, 2021.
- [6] Entwicklung der Energie- und CO2-Preise 2022. München: FfE München, 2022.
- [7] Neuzulassungen von Personenkraftwagen (Pkw) im Jahresverlauf 2021 nach Marken und alternativen Antrieben. Flensburg: Kraftfahrt-Bundesamt, 2022.
- [8] Johnsen, Doris et al.: Gesteuertes Laden von Elektrofahrzeugen über Preisanreize -Anwendungsbeispiele und Handlungsbedarf. Köln: TÜV Rheinland Consulting GmbH, 2020.
- [9] Gupta, Jitendra: A Bidirectional Home Charging Solution for an Electric Vehicle. In: IEEE International Conference on Environment and Electrical Engineering 2019. Delhi, New Delhi, India: Department of Electrical Engineering, Indian Institute of Technology, 2019.
- **[10]** Wei, Hongqian et al.: Planning integrated energy systems coupling V2G as a flexible storage. In: Energy Volume 239, Part B. Beijing, 100081, China: School of Mechanical Engineering, Beijing Institute of Technology, 2022.
- [11] Hinterstocker, Michael et al.: Bidirectional Charging Management Field Trial and Measurement Concept for Assessment of Novel Charging Strategies. München: Forschungsstelle für Energiewirtschaft e.V. (FfE), 2019.
- **[12]** Illing, Bjoern et al.: Achievable revenues for electric vehicles according to current and future energy market conditions. Porto: 2016 13th International Conference on the European Energy Market (EEM), 2016.
- **[13]** Shafiullah, Md et al.: Maximizing the profit of a load aggregator by optimal scheduling of day ahead load with EVs. Seville: 2015 IEEE International Conference on Industrial Technology (ICIT), 2015.

- [14] Bessa, Ricardo et al.: Optimized Bidding of a EV Aggregation Agent in the Electricity Market. New Jersey: IEEE Transactions on Smart Grid, VOL. 3, NO. 1, MARCH 2012, 2012.
- **[15]** Rominger, Julian et al.: Analysis of the German Continuous Intraday Market and the Revenue Potential for Flexibility Options. New Jersey, US: IEEE, 2019.
- [16] Schmidt, Robert et al.: Revenue-optimized Marketing of electric Vehicles' Flexibility Options. New Jersey, US: IEEE, 2019.
- [17] Peterson, Scott B. et al.: The economics of using plug-in hybrid electric vehicle battery packs for grid storage. London: RELX Group, 2009.
- **[18]** Pelzer, D. et al.: Energy arbitrage through smart scheduling of battery energy storage considering battery degradation and electricity price forecasts. Melbourne: 2016 IEEE Innovative Smart Grid Technologies Asia (ISGT-Asia), 2016.
- **[19]** Salpakari, Jyri et al.: Flexibility of electric vehicles and space heating in net zero energy houses: an optimal control model with thermal dynamics and battery degradation. In: Applied Energy 190/2017. Espoo, Finland: New Energy Technologies Group, Department of Applied Physics, School of Science, Aalto University, 2017.
- [20] Chen, Jianhong et al.: Strategic integration of vehicle-to-home system with home distributed photovoltaic power generation in Shanghai. In: Applied Energy 263/2020. Shanghai: Nanjing Branch of China Electric Power Research Institute, Shanghai Key Laboratory of Materials Protection and Advanced Materials in Electric Power, Shanghai Engineering Research Center of Electric Energy Conversion, Shanghai University of Electric Power, School of Electric Power, South China University of Technology, 2020.
- [21] Erdinc, Ozan et al.: Smart Household Operation Considering Bi-Directional EV and ESS Utilization by Real-Time Pricing-Based DR. In: IEEE TRANSACTIONS ON SMART GRID Vol. 6, No. 2, Mai/2015. ohne Ort: IEEE, 2015.
- [22] Kataoka, Ryosuke et al.: Comparison of the Economic and Environmental Performance of V2H and Residential Stationary Battery: Development of a Multi-Objective Optimization Method for Homes of EV Owners. In: World Electric Vehicle Journal 10/2019. Tokyo: Toyota Central R&D Labs., The University of Tokyo, 2019.
- [23] Wickert, Manuel et al.: Wissenschaftliche Unterstützung bei der Erstellung von fahrzeugbezogenen Analysen zur Netzintegration von Elektrofahrzeugen unter Nutzung erneuerbarer Energien - Endbericht zum Vorhaben FKZ UM 11 96 107. Kassel und Bremerhaven: Fraunhofer-Institut für Windenergie und Energiesystemtechnik, IWES, 2013.
- [24] International Journal of Sustainable Energy Planning and Management Cost and self-consumption optimised residential PV prosumer systems in Germany covering residential electricity, heat and mobility demand. Vol. 21 2019 35–58. Regensburg, Lappeenranta: OTH Regensburg, LUT University Lappeenranta, 2019.

- [25] Englberger, Stefan et al.: Electric vehicle multi-use: Optimizing multiple value streams using mobile storage systems in a vehicle-to-grid context. In: Applied Energy Volume 304. München: Institute for Electrical Energy Storage Technology, Technical University of Munich, The Mobility House GmbH, 2021.
- [26] Hanemann, Philipp et al.: Effects of electric vehicles on the spot market price. Amsterdam: Energy Volume 162, 1 November 2018, Pages 255-266, 2018.
- [27] Rodríguez, Cristian David Bedoya et al.: Evaluating the Impact of a V2G Scheme on the Demand Curve. New Jersey, US: IEEE, 2019.
- [28] Huang, Zhao et al.: Multi-objective optimization strategy for distribution network considering V2G-enabled electric vehicles in building integrated energy system. In: Protection and Control of Modern Power Systems volume 5, Article number: 7. Changsha, China: Hunan college of information, 2020.
- [29] Child, Michael et al.: The Impacts of High V2G Participation in a 100% Renewable Åland Energy System. In: Energies 11/2018. Lappeenranta, Turku: Lappeenranta University of Technology, Åbo Akademi, 2018.
- [30] Bjørndal, Endre et al.: Congestion management by dispatch or re-dispatch: flexibility costs and market power effects 10th International Conference on the European Energy Market (EEM). Stockholm: IEEE, 2013.
- [31] Felix, Böing: Cross-sector assessment of CO2 abatement measures and their impact on the transmission grid. Dissertation. Herausgegeben durch die TU München, geprüft von Wagner, Ulrich: München, 2020.
- [32] Meyer-Huebner, Nico et al.: N-1-Secure Optimal Generator Redispatch in Hybrid AC-DC Grids with Energy Storage. In: Innovative Smart Grid Technologies Conference Europe (ISGT-Europe) 2018 IEEE PES. Karlsruhe: Nico Meyer-Huebner, Michael Suriyah, Thomas Leibfried Institute of Electrical Energy Systems and High Voltage Technology, KIT, 2018.
- [33] Xiong, Bobby et al.: Spatial flexibility in redispatch: Supporting low carbon energy systems with Power-to-Gas. In: Applied Energy Volume 283, 1 February 2021, 116201. Trondheim: Department of Industrial Economics and Technology Management, 2021.
- [34] Eickmann, Jonas et al.: Optimizing Storages for Transmission System Operation. In: Energy Procedia 46 (2014) 13 – 21. Aachen: Institute of Power Systems and Power Economics (IAEW), RWTH Aachen, 2014.
- [35] Müller, Benjamin et al.: Coordinated Dynamic Use of Dispersed Flexibility to Maximize the Time-variant Aggregated Potential for Redispatch. In: IFAC-PapersOnLine Volume 53, Issue 2, 2020, Pages 12195-12200. Stuttgart: Institute for Combustion and Power Plant Technology (IFK), 2020.

- [36] Gutermuth, Georg et al.: Network operator owned storages as an option for congestion management. In: Innovative Smart Grid Technologies Europe (ISGT-Europe) 2020 IEEE PES. Ladenburg: ABB AG Corporate Research, 2020.
- [37] Staudt, Philipp et al.: Using vehicle-to-grid concepts to balance redispatch needs. In: Proceedings of the Ninth International Conference on Future Energy Systems e-Energy '18. Karlsruhe: Karlsruhe Institute of Technology, 2018.
- [38] Thormann, Bernd et al.: Evaluating the potential of future e-mobility use cases for providing grid ancilliary services. In: IEEE Xplore CIRED 2021 Conference. Leoben: Montanuniversitaet Leoben, Chair of Energy Network Technology, 2021.
- [39] Kigle, Stephan et al.: Greenhouse Gas Abatement in EUROPE—A Scenario-Based, Bottom-Up Analysis Showing the Effect of Deep Emission Mitigation on the European Energy System. In: Energies 2022, 15, 1334. Basel: MDPI, 2022. https://doi.org/10.3390/en15041334.
- [40] Müller, Mathias; Blume, Yannic; Reinhard, Janis: Impact of behind-the-meter optimised bidirectional electric vehicles on the distribution grid load. In: Energy 255, 124537. München: Forschungsstelle für Energiewirtschaft e.V. (FfE), 2022. DOI: https://doi.org/10.1016/j.energy.2022.124537
- [41] Weiss, Arnold: The Role and Importance of Power Exchanges. Vienna: EPEX SPOT SE, 2018.
- [42] Bereitstellung von Systemdienstleistungen aus Elektrofahrzeugen mit bidirektionalem Lademanagement - Ein BDL-Positionspapier zu Vehicle-to-Grid-Anwendungen. Ohne Ort: Bidirektionales Lademanagement – BDL, 2022.
- [43] Dossow, Patrick et al.: Profitability of V2X Under Uncertainty: Relevant Influencing Factors and Implications for Future Business Models. In: Energy Reports (Special Chapter of ICEEEP Proceedings). München: FfE, 2022.
- [44] Convenient, networked and sustainable: new solutions for charging electric Volkswagen models. Wolfsburg: Volkswagen, 2021.
- [45] Wohlschlager, Daniela et al.: Comparative environmental impact assessment of ICT for smart charging of electric vehicles in Germany. München: FfE, 2022.
- [46] Inflationsraten in Deutschland. In: https://www.finanztools.de/inflation/inflationsraten-deutschland. (Abruf 2022-01); Berlin: Statistisches Bundesamt (Destatis), 2021.
- [47] Day-ahead Prices. In: https://transparency.entsoe.eu/transmissiondomain/r2/dayAheadPrices/show. (Abruf am 2020-08-31); Brussels: ENTSO-E, 2020.
- [48] Martin, Henry et al.: German Intraday Electricity Market Analysis and Modeling Based on the Limit Order Book. Kassel, DE, Zurich, CH: ETH Zurich, 2018.
- [49] Hagspiel, Simon et al.: Cost-optimal power system extension under flow-based market coupling in: science direct energy. Amsterdam: Elsevier, 2014

- [50] Van Den Bergh, Kenneth; Delarue, Erik; D'haeseleer, William: DC power flow in unit commitment models in: TME Working Paper - Energy and Environment. Belgium: KU Leuven Energy Institute, 2014
- [51] Szenariorahmen zum Netzentwicklungsplan Strom 2037 mit Ausblick 2045, Version 2023 Entwurf der Übertragungsnetzbetreiber. Berlin, Dortmund, Bayreuth, Stuttgart: 50Hertz Transmission GmbH, 2022.
- [52] Ebner, Michael et al.: Kurzstudie Elektromobilität Modellierung für die Szenarienentwicklung des Netzentwicklungsplans. München: Forschungsstelle für Energiewirtschaft, 2019.
- [53] Borlaug, Brennan et al.: Levelized cost of charging electric vehicles in the United States. In: Joule 4 (7) (2020), pp. 1470-1485. Golden, CO, USA: National Renewable Energy Laboratory, 2020.
- **[54]** Ekins, Paul; Kesicki, Fabian; Smith, Andrew: Marginal Abatement Cost Curves A call for caution. London: University College London Energy Institute, 2011.
- [55] Chicco, Davide et al.: The coefficient of determination R-squared is more informative than SMAPE, MAE, MAPE, MSE and RMSE in regression analysis evaluation. In: PeerJ Computer Science 7:e623. Toronto: University of Toronto, 2021. https://doi.org/10.7717/peerj-cs.623.
- **[56]** Fattler, Steffen: Economic and Environmental Assessment of Electric Vehicle Charging Strategies. Dissertation. Herausgegeben durch die TU München, geprüft von Wagner, Ulrich und Wietschel, Martin: München, 2021.
- [57] Müller, Mathias et al.: Future grid load with bidirectional electric vehicles at home. Berlin: International ETG Congress 2021. VDE, 2021
- [58] Patzack, Sören; Erle, Niklas; Vennegeerts, Hendrik Dr.-Ing.; Moser, Albert Prof. Dr.-Ing.: Einfluss von auslegungsrelevanten Netznutzungsfällen auf die Netzdimensionierung. Aachen: Forschungsgemeinschaft für elektrische Anlagen und Stromwirtschaft e.V., 2016
- [59] NUTS Systematik der Gebietseinheiten für die Statistik. In: https://ec.europa.eu/eurostat/de/web/nuts/background. (Abruf am 2021-05-17); Luxemburg: Eurostat, 2021.
- [60] Fiedler, Claudia et al.: Modelling transformation pathways for EU27+3 final energy demand using temporally and spatially resolved sector models. In: Conference Proceedings Current and Future Challenges to Energy Security; 5th AIEE Energy Symposium, virtual conference, December 2020. München: Forschungsstelle für Energiewirtschaft e.V., 2020.
- [61] Hirth, Lion et al.: Kosten- oder Marktbasiert? Zukünftige Redispatch-Beschaffung in Deutschland Studie im Auftrag des BMWi. Berlin: Neon, 2019.

- [62] Gesetz zu Sofortmaßnahmen für einen beschleunigten Ausbau der erneuerbaren Energien und weiteren Maßnahmen im Stromsektor. Berlin: Deutsche Bundesregierung, 2022.
- [63] German Electricity Prices on the EPEX Spot Exchange in 2021. In https://www.ffe.de/en/publications/german-electricity-prices-on-the-epex-spot-exchange-in-2021/. (Abruf am 2022); München: FfE München, 2022.
- [64] V2G-Projekt BDL: Feldversuch mit 50 BMW i3 geht los. In https://www.electrive.net/2021/07/15/v2g-projekt-bdl-feldversuch-mit-50-bmw-i3geht-los/. (Abruf am 2022-09-03); Berlin: RABBIT PUBLISHING GmbH, 2022.
- **[65]** Zwischenspeicherung von Energie in bidirektional ladenden Elektrofahrzeugen zur Optimierung des Ladevorgangs am Intraday-Markt. München: Forschungsprojekt "Bidirektionales Lademanagement BDL", 2022.
- [66] Kotzur, Leander: A modeler's guide to handle complexity in energy systems optimization. In: Advances in Applied Energy 4 (2021) 100063. Jülich: Institute of Techno-economic Systems Analysis (IEK-3), Forschungszentrum Jülich GmbH, 2021.
- **[67]** Feste Einspeisevergütungen in Cent/kWh gemäß EEG 2021 und EEG 2017. Berlin: Bundesverband Solarwirtschaft, 2022.
- **[68]** Gesetz für den Ausbau erneuerbarer Energien (Erneuerbare- Energien-Gesetz EEG 2021). Berlin: Bundesregierung Deutschland, 2021.
- [69] Laitinen, Erkki K.: Discounted Cash Flow (DCF) as a Measure of Startup Financial Success. In: Theoretical Economics Letters 9, 2997-3020. Vaasa: The School of Accounting and Finance, University of Vaasa, 2019.
- [70] Knoke, Thomas: Use and misuse of the net present value in environmental studies. In: Ecological Economics Volume 174, August 2020, 106664. Munich: Institute of Forest Management, TUM School of Life Sciences Weihenstephan, Technical University of Munich, 2020.
- [71] Blume, Yannic et al.: Effects of variable grid fees on distribution grids with optimized bidirectional battery electric vehicles . In: 6th Grid Service Markets Symposium (GSM); Luzern: Hochschule Luzern, 2022.
- [72] Schaber, Katrin: Parametric study of variable renewable energy integration in Europe: Advantages and costs of transmission grid extensions. In: Energy Policy 42 (2012) 498–508. München: Lehrstuhl für Energiewirtschaft und Anwendungstechnik, Technical University Munich, 2012.
- [73] Das Übertragungsnetz von 50Hertz. In https://www.50hertz.com/de/Netz. (Abruf am 2022-08-28); Berlin: 50Hertz Transmission GmbH, 2022.
- [74]StrukturmerkmaledesElektrizitätsversorgungsnetzes.Inhttps://www.amprion.net/Netzkennzahlen/Strukturmerkmale/.(Abruf am 2022-08-28); Dortmund: Amprion GmbH, 2022.

# [75] Strukturmerkmale. In https://netztransparenz.tennet.eu/de/strommarkt/transparenz/transparenz-deutschland/netzkennzahlen/strukturmerkmale/. (Abruf am 2022-08-28); Bayreuth: TenneT TSO GmbH, 2022.

- [76] Strukturdaten. In https://www.transnetbw.de/de/transparenz/marktdaten/strukturdaten. (Abruf am 2022-08-28); Stuttgart: TransnetBW GmbH, 2022.
- [77] Müller, Ulf Philipp: The Role of the High Voltage Power Level in Future Power Systems and Their Modelling. In: International Conference on Smart Energy Systems and Technologies (SEST) 2018. Flensburg: Centre for Sustainable Energy Systems Flensburg University of Applied Sciences, 2018.
- [78] Wallbox Quasar Bidirektionale Ladestation. In https://besserladen.de/produkt/wallbox-quasar-bidirektionale-ladestation/. (Abruf am 2022-09-03); Dittelbrunn: Besserladen.de, 2022.

### 7 Thesis supervised by the author

- Rieping, Carl: Analysis of the Economic Benefits of Bidirectionally Chargeable Electric Vehicles for the Energy System of the Future. Masterarbeit. Herausgegeben durch die Technische Universität München, betreut durch die Forschungsgesellschaft für Energiewirtschaft mbH: München, 2021.
- Wendlinger, Christian: Berechnung und Analyse der Netzbelastung im zukünftigen europäischen Energiesystem mittels PTDF-Ansatz im Energiesystemmodell ISAaR.
   Masterarbeit. Herausgegeben durch die Hochschule Ansbach - Technische Hochschule Nürnberg - Hochschule Weihenstephan-Triesdorf, betreut durch die Forschungsstelle für Energiewirtschaft e. V.: Ansbach, 2021.
- Morlock, Elena Sophie: Entwicklung eines techno-ökonomischen Bewertungsmodells für Vehicle-to-Home-Systeme - Analyse der Erlöspotenziale von bidirektionalen Elektrofahrzeugen durch Eigenverbrauchserhöhung sowie durch tarifoptimiertes Laden und Entladen anhand eines Optimierungsmodells in MATLAB. Masterarbeit. Herausgegeben durch die Technische Universität Berlin, betreut durch die Forschungsgesellschaft für Energiewirtschaft. Berlin, 2020.
- Zimmermann, Fabian: Bidirektionales Laden von Elektrofahrzeugen -Modellbasierte Wirtschaftlichkeitsbewertung netzbezogener Anwendungsfälle -Bidirectional Charging of Electric Vehicles - Model-Based Economic Evaluation of Grid-Related Use Cases. Masterarbeit. Herausgegeben durch Technische Universität München - Lehrstuhl für Energiewirtschaft und Anwendungstechnik, betreut durch die Forschungsgesellschaft für Energiewirtschaft mbH, München, 2020.
- Haslbeck, Matthias: Auswirkungen von Power-to-Wasserstoff und Power-to-Methan auf den europäischen Gasmarkt - Impact of Power-to-Hydrogen and Power-to-Methane on the European Gas Market. Masterarbeit. Herausgegeben durch die Technische Universität München - Fakultät für Maschinenwesen. Betreut durch die Forschungsstelle für Energiewirtschaft e. V.: München, 2019.
- Buchwitz, Katharina: Energiesystem der Zukunft Sektorkopplung Strom und Gas -Detaillierung des Gasmarktmodells MInGa und Integration von Power-To-Gas. Masterarbeit. Herausgegeben durch RWTH Aachen - Forschungsgesellschaft für Energiewirtschaft mbH, betreut durch M. Sc. Kern, Timo; M. Sc. Deutz, Sarah: Aachen, 2018.
- Hübner, Tobias: Simulation von Referenzprozessen zur Bestimmung von Flexibilitätspotenzialen und der Auswirkungen von Effizienzmaßnahmen in der Grundstoffindustrie. Masterarbeit. Herausgegeben durch Technische Universität München, betreut durch die Forschungsgesellschaft für Energiewirtschaft mbH: München, 2017.

## 8 Publications of the Author

Kern, T.; Dossow, P.; von Roon, S. Integrating Bidirectionally Chargeable Electric Vehicles into the Electricity Markets. Energies 2020, 13, 5812. https://doi.org/10.3390/en13215812

©2020 by the authors. Published by MDPI.

\_\_\_\_\_

#### Publication 1 (Pub1): Integrating Bidirectionally Chargeable Electric Vehicles into the Electricity Markets

#### Timo Kern <sup>1,2,\*</sup>, Patrick Dossow <sup>1</sup> and Serafin von Roon <sup>1</sup>

- <sup>1</sup> Forschungsgesellschaft für Energiewirtschaft mbH (FfE), 80995 Munich, Germany; pdossow@ffe.de (P.D.); sroon@ffe.de (S.v.R.)
- Department of Electrical and Computer Engineering, Technical University of Munich (TUM), Arcisstraße
   21, 80333 München, Germany
- \* Correspondence: tkern@ffe.de; Tel.: +49-89-158121-35

Received: 31 August 2020; Accepted: 3 November 2020; Published: 5 November 2020

Abstract: Replacing traditional internal combustion engine vehicles with electric vehicles (EVs) proves to be challenging for the transport sector, particularly due to the higher initial investment. As EVs could be more profitable by participating in the electricity markets, the aim of this paper is to investigate revenue potentials when marketing bidirectionally chargeable electric vehicles in the spot market. To simulate a realistic marketing behavior of electric vehicles, a mixed integer linear, rolling horizon optimization model is formulated considering real trading times in the day-ahead and intraday market. Results suggest that revenue potentials are strongly dependent on the EV pool, the user behavior and the regulatory framework. Modeled potential revenues of EVs of current average size marketed with 2019 German day-ahead prices are found to be at around 200 €/EV/a, which is comparable to other findings in literature, and go up to 500 €/EV/a for consecutive trading in German day-ahead and intraday markets. For future EVs with larger batteries and higher efficiencies, potential revenues for current market prices can reach up to 1300 €/EV/a. This study finds that revenues differ widely for different European countries and future perspectives. The identified revenues give EV owners a clear incentive to participate in vehicle-to-grid use cases, thereby increasing much needed flexibility for the energy system of the future.

**Keywords:** V2G; bidirectionally chargeable electric vehicles; smart charging; unmanaged charging; spot markets; day-ahead market; intraday auction; continuous intraday trading; mixed integer linear optimization; revenues potentials of EVs

## 1. Introduction

The energy system transformation entails structural changes in all sectors. While the electricity supply sector in Germany has already been subjected to massive adjustments due to the expansion of renewable energies, the final energy sectors of private households, industry, transport, as well as small and medium enterprises have been very slow to switch to more climate-friendly technologies.

Emissions from the transport sector in Germany even increased since 1990 [1] and will, therefore, be unable to contribute to achieving the climate targets for 2020. Switching to electric vehicles (EVs) is one possible solution for lowering emissions. When comparing the carbon footprint of an EV to an internal combustion engine vehicle (ICEV), both the production of the battery as well as tailpipe emissions of the EV depend on the underlying energy system [2]. Due to a targeted reduction of CO<sub>2</sub>-emission of the European energy sector by up to 95% [3], at least operational emissions of EVs resulting mainly from charging electricity will decrease. Consequently, EVs are promising for lowering CO2-emissions.

However, due to higher initial investment costs compared to ICEVs, the integration of EVs proves to be challenging. Investment in EVs could become more profitable by participating in the electricity markets by smart or bidirectional charging. In this respect, the "Bidirectional Charge Management" (BCM) project, which was launched in May 2019, focuses on the analysis of revenue potentials of bidirectionally chargeable EVs in the different electricity markets [4]. The main region of our investigations is Germany since it is the largest electricity market in Europe that is characterized by a heterogenic generation portfolio of renewable energy sources as well as conventional power plants [5]. Due to the German energy transition, volatility in electricity generation will continue to increase thereby enhancing the potential benefits of bidirectional chargeable EVs. German spot markets include the day-ahead market (auction at 12 noon one day before delivery), the intraday auction (at 3 pm one day before delivery) [6]. In addition, the results compare revenue potentials of bidirectionally chargeable EVs in 28 European countries, where findings of the parameter analysis can be generalized to other regions.

The BCM project defined 13 different use cases for bidirectional charging management separated into the three revenue creation groups: "Vehicle-to-Grid" (V2G), "Vehicle-to-Business" (V2B) and "Vehicle-to-Home" (V2H) [4]. Two use cases of the V2G use cases group are examined in this paper, which refer to arbitrage trading in the day-ahead market and the intraday market. Since bidirectionally chargeable EVs that participate in the spot markets increase the flexibility of the markets and energy system, these two use cases can have an impact on a cost-effective integration of renewable energies while providing revenues to the EV owners. The aim of this paper is to show these revenue potentials for the marketing of EVs in an aggregated pool in the spot markets at the same time ensuring faster integration of e-mobility into the energy system of the future.

Several scientific publications have discussed the revenue potentials by participation of EVs in the spot markets [7]. Using smart (but unidirectional) charging, an aggregator can considerably reduce the costs of charging electric vehicles [8,9]. Such costs can be further reduced and revenues can be generated by bidirectional charging according to [10]. The studies mentioned above are limited to marketing in the day-ahead and reserve market. Rominger et al. [11] point out revenues for intraday trading, but only restricted on flexibilities with constant availability such as stationery storage. Schmidt et al. [12] consider prices in the day-ahead market and the intraday auction for EV charging, but is limited to a smart charging process without discharging to the grid. We are thus extending existing research with a more detailed representation of bidirectionally chargeable EVs participating in the spot markets by considering day-ahead and intraday trading in a rolling optimization with a limited time horizon. Another limitation of most studies mentioned is the focus on revenues of EVs by a fixed EV parameterization. As revenue potentials of bidirectionally chargeable EVs are strongly dependent on many influencing parameters, this paper points out how a variation of user parameters like the minimum safety state of charge (SoC), minimum SoCs at departure, plug-in probabilities and location of charging infrastructure change revenue potentials.

Peterson et al. deal with profits of bidirectionally chargeable EVs using arbitrage trading in three different US local markets [13]. Pelzer et al. find out that revenue potentials of bidirectionally chargeable EVs in the US and Singapore markets are highly dependent on spatial and temporal resolution of market prices [14]. These studies regard battery degradation costs, but are limited to an EV participation in only one electricity market and modeling without consideration of user behavior parameters.

Furthermore, several studies investigate the effects of V2G on the electricity markets. The participation of bidirectional vehicles in the spot market has a smoothing effect on electricity prices. In times of surplus feed-in by renewable energies, V2G is able to reduce price drops resulting in higher market values of renewable energies [15]. Rodríguez et al. [16] show a flattening impact of V2G applications on the demand curve that results in a higher load factor. These findings showing promising effects on the energy system are expanded by our study to determine whether the vehicle owner can also benefit by participating in the electricity market. Since the regulatory framework is of significant importance for the evaluation of revenues, the influence of additional charges on purchased energy is pointed out.

## 2. Methods

To determine V2G revenue potentials, we developed an aggregated storage optimization model that covers the use cases of arbitrage trading in the spot markets. The holistic implementation of all V2G use cases facilitates the applicability and avoids building several parallel models.

A simplified representation of the modeling process is displayed in Figure 1, where the model consists of four different parts. Based on a model of the EV's battery, optimized charging strategies can be developed for different scenarios depending on charging and discharging restrictions. An uncoupled aggregated EV pool can be modeled to participate in different markets, where marketing strategies are optimized with a rolling forecast of market prices. The optimized strategy depends on electricity prices of the respective energy markets and is based on historical and simulated future market prices. In addition to bidirectionally chargeable EVs, two reference scenarios are considered that cover smart, unidirectionally chargeable EVs and simple, directly chargeable EVs. The entire model is implemented in Matlab, where a CPLEX solver (optimization software package) is used for optimization.



**Figure 1.** Schematic representation of the developed optimization model for Vehicle-to-Grid (V2G) use cases of bidirectional charging.

#### 2.1. Modeling of Bidirectionally Chargeable Electric Vehicles

Modeling a bidirectionally chargeable EV consists mainly of a model of the electric battery of the vehicle similar to stationary electricity storage. The EV battery is modeled by the storage equation displayed below, which relates the state of charge (SoC) of the battery to the different amounts of electricity charged into or discharged out of the battery:

$$SoC(t) = SoC(t-1) + P_{charge}(t) \cdot \eta_{charge} \cdot \Delta t - \frac{P_{discharge}(t)}{\eta_{discharge}} \cdot \Delta t + \frac{P_{counter-purchase}(t)}{\eta_{discharge}} \cdot \Delta t - P_{counter-sale}(t) \cdot \eta_{charge} \cdot \Delta t$$

$$+ P_{schedule}(t) \cdot \Delta t + P_{fastcharge}(t) \cdot \Delta t \cdot \eta_{charge} - E_{consumption}(t)$$

$$(1)$$

here, t stands for the modeled point in time,  $\Delta t$  is the difference between two points in time and  $\eta$  represents efficiency, which differs for the charging and discharging process.  $P_{charge}(t)$  is the charging power and  $P_{discharge}(t)$  the discharging power at the modeled time t. The variables are considered on the alternating current side of the charging station and thus correspond to the amounts of energy traded in the market.  $P_{schedule}$  is the sum of purchases and sales already made at the modeled time t. Accordingly,  $P_{counter-purchase}(t)$  and  $P_{counter-sale}(t)$  represent the power which can be purchased or sold in the market to counteract transactions that have already taken place (countertrading). For example, purchased energy in the day-ahead market could be sold in the intraday markets resulting in a countertrade  $P_{fastcharge}(t)$  is the power, that can be used to rapidly charge the vehicle if necessary and  $E_{consumption}(t)$  is the EV's energy consumption by driving at time t. The limit values of all time-dependent variables are explained in the following section. Table A1 in the Appendix A provides an overview of all these variables and their limit values.

#### 2.1.1. State of Charge

SoC(t - 1) represents the battery's storage level at the point in time prior to the modeled point in time. The change in capacity of the battery storage corresponds to the difference from SoC(t) to SoC(t - 1). Due to the limited storage capacity of the electric vehicle and user requirements for a minimum storage level, the variable SoC(t) can assume a limited range of values.

The maximum storage level of the battery storage is limited by  $SoC_{max}$ . If a value of 100% is set for this parameter, the entire storage capacity is available for the charging process. The minimum value  $SoC_{min}$  varies depending on the vehicle's location status, which is known for each point in time. Equation (2) summarizes the values that  $SoC_{min}$  can assume:

$$SoC_{min,safe}$$
 for status = connected  
 $SoC_{min}(t) = SoC_{min,dep}$  for status = departure (2)  
 $SoC_{min \ disconnected}$  for status = not connected

If the vehicle is connected to the electric grid,  $SoC_{min}$  equals  $SoC_{min,safe}$ . The storage level must not fall below this value or must load onto it as quickly as possible. If the vehicle is connected to a charging station and is at the point of departure,  $SoC_{min}$  assumes the value  $SoC_{min,dep}$ . Before departure, the charging strategy is thus optimized in a manner such that the SoC at the time of departure at least corresponds to  $SoC_{min,dep}$ . If the vehicle is not connected to the electric grid, the value  $SoC_{min,disconnected}$  results in the minimum SoC.

To ensure that SoC(t) lies between the minimum and maximum possible storage level, Equation (3) is implemented:

$$SoC_{min}(t) \cdot C \leq SoC(t) + P_{supplement}(t) \cdot \Delta t \leq SoC_{max}(t) \cdot C,$$
 (3)

where C describes the storage capacity of the electric vehicle and  $P_{supplement}$  stands for additional, theoretical electric power to be charged.  $P_{supplement}(t)$  is incorporated to meet the storage level restriction in Equation (2) at any time. The possibility that the value of the storage level is below  $SoC_{min}$  exists even if the EV is connected to the electric grid. This might occur if the storage level is lower than the minimum SoC when the vehicle arrives at a charging station or if it is not possible to charge to the minimum SoC before departure because the vehicle has not been connected for long enough. In the case that the minimum storage level cannot be reached with the charging strategy,  $P_{supplement}(t)$  takes on a value greater than 0 to simulate a hypothetical charging process. The variable thus can be interpreted as penalty costs that arise from the driving behavior of a user who disregards the requirements for a minimal SoC. By introducing  $P_{supplement}$ (t), the model can optimize all driving profiles regardless of driving behavior or consumption, so that a selection of unsuitable driving profiles does not have to be made in advance.  $P_{supplement}$ (t) is not taken into account in the storage equation and does not change the actual storage level.

#### 2.1.2. Charging/ Discharging Power and Already Traded Energy

In the storage Equation (1),  $P_{charge}(t)$  and  $P_{discharge}(t)$  describe the purchase and sale of power and determine the change in storage capacity for each time step. Due to the limited power of any EV charging station, the variables are limited to the maximum charging and discharging power  $P_{charge,max}(t)$  and  $P_{discharge,max}(t)$  and to the minimum charging and discharging power  $P_{charge,min}(t)$  and  $P_{discharge,min}(t)$ . If the vehicle is connected to a charging station, the EV battery can be charged with  $P_{charge,max}(t)$  or discharged with  $P_{discharge,max}(t)$ . If the vehicle is not connected to the electric grid, both maximum and minimum charging and discharging power become 0.

The boolean variables  $b_{charge}$  and  $b_{discharge}$  describe the state of the battery during charging and discharging processes. If a charging process takes place,  $b_{charge}$  is *true*. Analogously, the variable  $b_{discharge}$  becomes *true* during discharging. Due to the fact that it is not possible to purchase and feed electricity into the electric grid at the same time, only one of the boolean variables can assume the value 1 (= *true*) at the modeled point in time Equation (4). A simultaneous purchase and sale in the market is, therefore, excluded.

$$b_{charge}(t) + b_{discharge}(t) \le 1$$
 (4)

The resulting constraints regarding these variables are shown in Equations (5) and (6):

$$P_{charge,min}(t) \cdot b_{charge}(t) \leq P_{charge}(t) \leq \left(P_{charge,max}(t) - P_{schedule,purchase}(t)\right) \cdot b_{charge}(t), \quad (5)$$

$$P_{discharge,min}(t) \cdot b_{discharge}(t) \leq P_{discharge}(t)$$

$$\leq \left(P_{discharge,max}(t) - P_{schedule,sale}(t)\right) \cdot b_{discharge}(t). \quad (6)$$

There is a possibility that the storage capacity of the electric vehicle has already been marketed through previous trading on the electricity markets, for example through consecutive trading on different spot markets. Such traded power must be taken into account in subsequent storage optimizations. In Equations (5) and (6),  $P_{schedule,purchase}$  and  $P_{schedule,sale}$  correspond to already made purchases or sales at the modeled time t. These amounts of electricity reduce the maximum charging or discharging power in such a way that only the capacity that has not yet been traded can be marketed.  $P_{schedule}$  can be defined as the difference between  $P_{schedule,purchase}$  and  $P_{schedule,sale}$  Equation (7) and is included in the storage Equation (1).

$$P_{schedule}(t) = P_{schedule,purchase}(t) \cdot \eta_{charge} - \frac{P_{schedule,sale}(t)}{\eta_{discharge}}$$
(7)

#### 2.1.3. Countertrades

Electricity spot markets in Germany include consecutive day-ahead and intraday trading resulting in the opportunity to countertrade day-ahead purchases or sells in the intraday market. The model not only accounts for already marketed storage capacities, it also includes the possibility of compensation transactions (countertrades) that compensate for the previous trade in the opposite direction. In this regard, the variable  $P_{counter-purchase}(t)$  corresponds to a buyback (counter purchase), the variable  $P_{counter-sale}(t)$  to a sellback (counter sale) of already traded storage capacities. The volume of a countertrade can at most assume the previously inversely traded amount of energy. Countertrades do not describe a physical loading or unloading process. The resulting constraints are shown in Equations (8) and (9):

$$P_{counter-purchase}(t) \leq P_{schedule,sale}(t) \cdot b_{counter-purchase'}$$
(8)

$$P_{counter-sale}(t) \le P_{schedule, purchase}(t) \cdot b_{counter-sale}.$$
(9)

Similar to the charging and discharging processes, the boolean variables  $b_{counter-purchase}$ and  $b_{counter-sale}$  describe the state of countertrades. If a counter-purchase takes place at time t ( $b_{counter-purchase} = 1$ ), the EV battery cannot be discharged at the same time. Conversely, no charging process can be conducted during a counter-sale ( $b_{counter-sale} = 1$ ). Equations (10) and (11) show these constraints:

$$b_{counter-purchase}(t) + b_{discharge}(t) \le 1,$$
 (10)

$$b_{counter-sale}(t) + b_{charge}(t) \le 1.$$
 (11)

#### 2.1.4. Electricity Consumption and Fast Charging

The battery of a vehicle has an electric energy consumption  $E_{consumption}(t)$  at the modeled time t, which is considered in the storage equation. Due to the foresight of the driving profiles (explained in Section 2.4), EV consumption is known at all times. Each driving phase of the vehicle results in a capacity reduction.

Depending on the user's driving behavior, it might occur that the electricity consumption of an EV is so high at one point in time that the current storage level is not sufficient to meet the energy demand, for example if the EV has not been connected to a charging station for too long. In this case, the fast charging power  $P_{fastcharge}(t)$  is utilized to comply with the restrictions of  $SoC_{min}(t)$  and to avoid a supposed negative storage level. The employment of the fast charging process is accompanied by an increase in storage capacity.  $P_{fastcharge}(t)$  represents the charging power at a public charging station rather than at a bidirectional charging station. The vehicle user is thus given the opportunity to charge the vehicle on the road.

#### 2.2. Formulation of Optimization Model

The developed model of a bidirectionally chargeable EVs allows for the implementation of different optimized charging and discharging strategies, which differ in particular in the

structure of the objective function. For the assessment of the use cases of arbitrage trading on the day-ahead market as well as on the intraday market, three different charging strategies are implemented: a strategy for bidirectional charging, a strategy for smart charging, and a strategy for unmanaged charging.

First, the bidirectional charging strategy, which allows for charging and discharging of the EV, is restricted to the storage equation and its aforementioned constraints. The objective of this charging strategy is to charge at minimum costs while discharging at maximum revenue. To do so, the objective function of the optimization model aims at minimizing all costs considered:

$$min\left(\sum_{t=1}^{T} p_{market,buy}(t) \cdot \left[P_{charge}(t) + P_{counter-purchase}(t)\right] - \sum_{t=1}^{T} p_{market,sell}(t) \cdot \left[P_{discharge}(t) + P_{counter-sale}(t)\right] + \sum_{t=1}^{T} p_{fastcharge}(t) \cdot P_{fastcharge}(t) + \sum_{t=1}^{T} p_{supplement}(t) \cdot P_{supplement}(t)\right)$$
(12)

where *T* corresponds to the number of time steps of the optimization. Depending on the respective market, traded energy quantities per time step as well as corresponding market prices  $p_{market,i}(t)$  are considered. In this regard,  $p_{market,buy}$  is the price at which electricity is bought and  $p_{market,sell}$  is the price at which electricity is sold, where both prices can include respective transaction costs and possibly additional electricity price components. Charged power corresponds to a purchase transaction and is associated with costs as is each counter purchase of power. In contrast, discharged power and counter sales are traded with corresponding revenues, which is why  $P_{discharge}$  and  $P_{counter-sale}$  are subtracted.

Fast charging power and supplement power are also included in the objective function. Both fast charging costs  $p_{fastcharge}(t)$  and penalty costs  $p_{supplement}(t)$  are fixed to be a relatively high value, so that only the minimum necessary power is charged to meet the requirements for minimum storage level. As  $P_{fastcharge}(t)$  should only be utilized to the extent that a negative *SoC* is avoided,  $p_{fastcharge}(t)$  should be selected sufficiently larger than  $p_{supplement}(t)$ . Thus, the following condition must also be fulfilled to guarantee a functioning bidirectional charging strategy:

$$p_{market,i}(t) \ll p_{supplement}(t) \ll p_{fastcharge}(t).$$
 (13)

Second, the smart charging strategy is implemented as a reference scenario to simulate already existing smart charging stations. The objective is to minimize electricity purchase costs by intelligent charging of the EV. Since discharging the EV battery is impossible in this scenario, Equation (14) is implemented. By eliminating the discharge power, the objective function already defined via Equation (12) for the bidirectional charging strategy can also be used for the smart charging strategy.

$$0 = P_{discharge,min} \le P_{discharge} \le P_{discharge,max} = 0$$
(14)

Third, the unmanaged charging strategy accounts for today's most commonly installed simple charging stations as a second reference scenario, where the EV battery is charged as soon as the vehicle is connected without an optimized charging control. As with the smart charging strategy, discharging is not possible Equation (14). The aim of the unmanaged charging strategy is, therefore, to maximize the storage level that is equal to minimizing the negative value of SoC(t) at all times. The battery is accordingly charged at maximum charging power until the battery's storage level corresponds to  $SoC_{max}(t)$  or until the EV leaves the location. In addition, as for the previously explained strategies, fast charging costs and penalty costs for an insufficient SoC with regard to the location-based limit values are included. The resulting objective function is expressed as follows:

$$min\left(\sum_{t=1}^{T} p_{fastcharge}(t) \cdot P_{fastcharge}(t) + \sum_{t=1}^{T} p_{supplement}(t) \cdot P_{supplement}(t) - \sum_{t=1}^{T} SoC(t)\right)$$
(15)

#### 2.3. Optimization with Limited Forecast in Consecutive Spot Markets

To investigate the influence of different characteristics and requirements of the considered markets on revenue potentials of bidirectional charging, optimized trading strategies based on price forecasts are simulated by a rolling optimization model, where realistic trading behavior results from a limited foresight of market prices.

Acting in the market under uncertainty is modeled in the same manner as described in [17], where each day is divided into 8 time slices of three hours each. The model regards real trading times in the spot markets. Figure 2 illustrates the methodical procedure of consecutive trading in the day-ahead and intraday markets with rolling price forecast horizons, where each horizontal bar displays the prices known in the respective optimization run of three hours. At 12 noon of day *d*, for instance, a market participant sees averaged continuous intraday prices of the following 12 quarter-hourly products. At the same time, less precise quarter-hourly prices of the continuous intraday are assumed for the interval from 3 pm to midnight. For day d + 1, day-ahead market prices are known and for d + 2 a forecast of the day-ahead market prices is presented. The participant's trading decision, which is the optimized marketing strategy, is based on this limited foresight.



**Figure 2.** Schematic representation of the optimization steps with limited foresight of market prices.

As described by the example, foresight of market prices varies for the individual markets. Since the auction on the day-ahead market takes place daily at 12 noon for the respective following day d + 1, precise prices forecasts for d + 1 are known shortly before 12 noon on day d. To prevent unrealistic trading behavior at the end of day d + 1, such as discharging all batteries to maximize revenues, estimated prices for day d + 2 are included in the forecast horizon, where prices are also presented at 12 noon of day d. The forecast period d + 2 can be one or more days representing a worse or better foresight and is evaluated in Appendix B. The length of the optimization time steps for day-ahead trading is 1h.

For the intraday auction, precise price forecasts of day d + 1 are known shortly before 3 pm of day d, since the auction takes place at 3 pm. The length of the optimization time steps is 0.25 h.

Following the intraday auction, continuous intraday trading starts at 4 pm with quarterhourly products, which defines the length of the optimization time steps. Here, a first forecast horizon of relatively precise prices is set to three hours covering the following 12 quarter-hourly products, where prices are based on trading transactions of these three hours. For the period following the three-hour time window, all continuous intraday transactions of this interval are used to calculate a second forecast price, thereby reflecting the uncertainty of market prices.

Hence, optimization runs before noon include market price information of the remaining day d and the following day d + 1. The optimization runs from 12 noon on the trading day also include day d + 2. The total revenue of the marketed EV battery corresponds to the

summed costs and revenues of all traded products (filled areas). The cross-hatched areas in Figure 2 are not regarded as revenues, since these are only price forecasts serving as reference points for the trading strategy.

If consecutive trading takes place in several markets, storage capacities already marketed must be taken into account in subsequent optimization runs and can be countertraded as described before. The storage level at the end of real continuous intraday trading (filled blue area) of each optimization run determines the actual charging and discharging behavior of the vehicle. This storage level is applied as the starting value for the subsequent optimization run.

#### 2.4. Input Data and Parameterization of Electric Vehicle (EV) Pool Scenarios

In the model, parameters related to the EV are the battery's storage capacity *C*, charging and discharging power  $P_{charge/discharge}$ , and different efficiency parameters. To investigate the range of revenue potentials in detail, three different sets of EV parameters are implemented: First, a currently common-sized EV is modeled (EV1), comparable to a 2018 BMW i3 [18] and a 2018 Renault Zoe [19], using realistic values regarding storage capacity, charging and discharging power and efficiencies. Second, a relatively large EV and a highly efficient charging station are defined representing a future EV (EV2). Third, a set of ambitious, yet plausible future values is selected to model maximum revenue potentials (EV3). These parameter sets were discussed and agreed upon within the research project BCM. Table 1 summarizes the chosen parameter values for the three sets of EV models.

Parameter	EV1	EV2	EV 3	
Storage capacity	С	38 kWh	100 kWh	100 kWh
Charging power	$P_{charge}$	11 kW	11 kW	22 kW
Discharging power	$P_{discharge}$	10 kW	11 kW	22 kW
Charging efficiency (AC-DC)	$\eta_{charge}$	92.5%	94.5%	95.0%
Discharging efficiency (DC-AC)	$\eta_{discharge}$	92.0%	94.5%	95.0%
Roundtrip efficiency (AC-AC)	$\eta_{roundtrip}$	85.1%	89.3%	90.3%

Table 1. List of relevant electric vehicle (EV) parameters and chosen set of parameter values.

All losses and efficiencies considered in the model are based on discussions and on the consultation with experts from the BCM project. Other studies assume roundtrip efficiencies that are similar to EV1 [13] or slightly lower [14]. Constant values are set for the efficiencies in order to allow for a linear optimization problem, which results in much faster optimization times and thus enables many more optimization runs, i.e., more results. In real operation, however, efficiencies follow a declining, non-linear course for decreasing charging power. Hence, resulting revenue potentials of the presented model overestimate real revenues of bidirectional charging.

The user parameters result from characteristics, requirements and behavior of the vehicle user. In contrast to a large-scale stationary storage system, the battery of an EV is not continuously connected to the grid. The availability of an EV battery for V2G use cases largely depends on the individual driving profile of the user, the location of an appropriate charging station and the probability that the user has connected the vehicle to this charging station.

As a detailed representation of the driving behavior of the user, vehicle-specific driving profiles describe the EV's whereabouts as well as its energy consumption while driving in a chronological sequence. Based on data regarding household and route information as well as individual user logbooks from the "Mobility in Germany 2017" study [20] and a methodology first developed in the MOS 2030 [21] project, annual driving profiles of various EVs are created that are available as supplementary material (see Section 6). Each profile meets the following standards:

- A change of location is always accompanied by a driving phase.
- During each driving phase, the EV has discrete consumption, which leads to a reduction of the storage level.
- The EV can be located and connected either at the place of residence, the place of work or the public space

The temporal resolution of the driving profiles is quarter-hourly intervals. For each profile, the energy consumption is calculated based on information regarding driving speed, outside temperature and vehicle type.

The basic data is additionally used to cluster these driving profiles into user groups to further analyze the influence of user behavior on revenue potentials resulting in a set of commuter groups which display typical commuter behavior, and a set of non-commuter groups with homogeneous behavior different to commuter behavior. The commuter set consists of 12 commuter groups. These are defined by the time of arrival of the vehicle at the place of work and the distance traveled from the place of residence to the place of work. The non-commuter set is made up of three user groups, which are determined by age and number of persons in a household. The number of created commuter and non-commuter profiles per group reflects the real distribution within the German vehicle fleet [22]. Since revenue potentials are strongly related to driving behavior, these two different pools of driving profiles are defined as input for the model:

- a commuter pool consisting of representatives of all 12 commuter groups;
- a non-commuter pool consisting of representatives of all 3 non-commuter groups.

Table 2 summarizes the characteristics of the two pools of driving profiles including the probability of the EVs' whereabouts, which is the averaged probability of the EV's location at any given point in time. The sum of probability of all three locations apart from the driving phase is 94.5% for the commuter pool and 96.8% for the non-commuter pool, which represents the theoretical availability for bidirectional charging management if an appropriate charging station is installed at their location. To analyze the influence of possible charging station locations on revenue potentials of the discussed V2G use cases, the charging point location parameter can be flexibly selected in the model for each individual EV, where the distinguished three locations can be individually defined as available for bidirectional charging or not available.

Pools of Driving Profiles	Probability of Whereabouts				Averaged	Averaged
	Place of Residence	Place of Work	Public Space	Driving Phase	Consumption (kWh/100km)	Driving Distance (km/a)
Commuter Pool	68.8%	22.1%	3.6%	5.5%	17.4	13,600
Non-commuter Pool	87.5%	1.4%	7.9%	3.2%	17.4	8300

#### Table 2.Characteristics of user pools.

The probability of each individual EV user to plug the vehicle into an available bidirectional charging station upon arrival determines the plug-in probability, where the expected value of a normally distributed probability is defined as a parameter. A higher plug-in probability results in a greater availability of the EV for V2G use cases. The parameter can be set flexibly to any value between 0% and 100%. As users will most likely be rewarded in some way for plugging in their EV, the plug-in probability is expected to be very high, up to a 100% certainty.

The parameter  $SoC_{min,safe}$  states the minimum storage level not to be undercut when the EV is connected to the electric grid, which guarantees a certain safety range in the event of an unscheduled departure. This parameter can be set flexibly to meet the requirements of users. The storage level that must be reached at the time of a scheduled departure,  $SoC_{min,dep}$ , should be adjustable by the user according to his/her preferences in a real implementation. In the model, the parameter can be set between 0% and 100%.

The charging and discharging behavior model is determined in particular by the time series of market prices. For the use cases of arbitrage trading, actual price time series of day-ahead and intraday markets from 2019 are used to represent price forecasts of a maximum of two and a half days [23,24]. For trading in the day-ahead and the intraday markets, corresponding auction prices of 2019 are used. Regarding the continuous intraday market, real prices are bilaterally determined for each transaction, where buy and sell orders are constantly matched. Thus, two representative forecast prices are determined. For the relatively precise forecast of the next three hours after modeling time t, ID3 is calculated, which is the volume-weighted quarter-hourly price of all transactions in the market for the last three hours, where market liquidity is sufficient to determine a representative market price. For the more uncertain time beyond three hours after modeling time, ID<sub>Avg</sub> is used, which is the volume-weighted quarter-hourly price of all transactions for this forecasted time horizon.

The regulatory framework for bidirectional charging applications is not yet fully defined to the point that simulated revenue potentials might determine what kind of regulatory incentive or obstacle enables or respectively prevents the considered V2G use cases. A market design with a reduction of different electricity price components such as grid fees and taxes would decrease the marginal costs of the EV accordingly and thus lead to an increased discharging behavior. To incorporate this highly important role of the market design in the model, various values are assigned to the additional charges on purchased energy parameter and resulting differences in revenues are assessed. The applied additional charges on purchased energy range from 0 €/MWh, which corresponds to a complete exemption from all additional electricity price components, to 234 €/MWh, which reflects the amount of all electricity price components for households in Germany in 2019, excluding electricity purchase prices [25].

## 3. Results

For the following investigations, user and EV parameters that are introduced in Section 2.4 are combined to form six EV pool scenarios (EV1, EV2, EV3 each for commuters and noncommuters). Displayed revenues of bidirectional or smart charging EVs always refer to the difference of these revenues to revenues of the unmanaged charging scenario. As long as there is no presentation explicitly showing single profile revenues, displayed revenues always refer to mean revenues of the considered EV pool scenario. User, modeling and regulatory parameters are set to the values shown in Table 3. The process of determining suitable parameters is further discussed in Section 3.3 as well as Appendixes B and C.

Parameter	Value Type		Further Discussion of Parameters' Influence on Revenue Potentials	
Minimum SoC at departure	70%	User	Section 3.3.1	
Minimum safety SoC	20% (EV1 and EV2) 30% (EV3)	User	Section 3.3.1	
Plug-in probability	100%	User	Section 3.3.1	
Charging point location	At place of residence	User	Section 3.3.1	
Additional charges of purchased energy	0 €/MWh	Regulatory	Section 3.3.2	
Forecast period	1 day	Model	Appendix B	
EV pool size	Commuter: 50 Non-commuter: 75	Model	Appendix C	

Table 3. Fixed user, modeling and regulatory parameters for the investigation of revenue potentials.

#### 3.1. Revenue Potentials for Vehicle-to-Grid (V2G) Use Cases

All resulting datasets in Section 3.1, individual revenue potentials depending on EV pool scenario, and driving profiles, have been made freely available (see Section 6). The following analysis shows average revenue potentials and thus represents an aggregated extract of the provided result data.

#### 3.1.1. Revenue Potential in the German Spot Market

The German spot market is divided into the day-ahead auction with hourly products, the intraday auction with quarter-hourly products, and the continuous intraday trading offering quarter-hourly and hourly products. Figure 3 shows the revenue potentials for the EV pool

scenarios considered in the different markets with a separate participation in the markets compared to consecutive marketing. In comparison, the revenue potential of smart charging is shown in red bars. For commuters, smart charging is more attractive than for non-commuters because of the higher annual driving and thus the higher need for charging. However, for both commuters and non-commuters arbitrage trading with bidirectional charging leads to much higher revenues compared to smart charging. Revenues of non-commuters are slightly higher than revenues of commuters due to the higher availability at the place of residence.

For trading in the day-ahead market, revenues range from almost 200  $\notin$ /EV/a (EV1) to 600  $\notin$ /EV/a (EV3). Quarter-hourly intraday trading leads to higher revenues due to the higher volatility in prices. Consecutive trading in the day-ahead market, the intraday auction and continuous intraday trading results in best-case revenues of 400  $\notin$ /EV/a (EV1) to 1300  $\notin$ /EV/a (EV3). Comparing the EV1 to EV2 pool, the 2.6 times higher capacity of EV2 implicates higher flexibility for charging and discharging times. Therefore, revenues of EV2 are 200  $\notin$ /EV/a higher than the revenues of EV1. EV3 almost doubles the revenues of EV2 due to doubled charging and discharging power. The increase of charging and discharging power is, thus, even more relevant for revenue potentials than the increase of battery capacity.



Figure 3.Revenues for bidirectionally chargeable and smart charging EVs participating in<br/>different spot markets in Germany.

#### 3.1.2. Revenue Potential in European Markets

Since the energy systems change to a volatile and renewable production in most European countries, flexibility will be needed to cover the demand at any particular time. Therefore, bidirectionally chargeable EVs are a possible flexibility option in all European countries. To quantify revenue potentials in European countries other than Germany, we use entso-e data (European Network of Transmission System Operators for Electricity) of electricity day-ahead prices for 2019 as an input for the developed optimization model [24].

Figure 4 shows the resulting revenues for bidirectionally chargeable EVs compared to unmanaged charging for 28 European countries for commuters and non-commuters. Revenues are the highest in Ireland, Romania, Bulgaria and Hungary. These countries had scarcity prices of more than 100 €/MWh during approximately 200 hours to go along with a high standard deviation of electricity prices in 2019, giving bidirectionally chargeable EVs an opportunity to use arbitrage trading more profitably. On the other hand, the revenues are the lowest for Norway and Sweden. High capacities of hydropower and some nuclear power plants in Sweden characterize the energy system in those countries [26] with almost constant marginal costs, resulting in barely volatile electricity prices.

In other countries, like Germany, Austria and France, bidirectionally chargeable EVs generate medium revenues in arbitrage trading. These energy systems are more heterogenic with some volatile renewable production as well as gas-fired, coal-fired or nuclear electricity production. Revenues in this group of countries are still varying. For example, nuclear power plants with almost constant marginal costs dominate electricity production in France [27] resulting in narrow price spreads. As another example, Austria's energy system shows high capacities of pump storage facilities [26] resulting in flattened electricity prices. These structural characteristics lead to slightly lower revenues for bidirectionally chargeable EVs for the year 2019. On the other hand, Germany has a heterogenic production portfolio of volatile wind and solar generation as well as conventional power plants with widely varying marginal costs. Price spreads and the resulting revenues for bidirectionally chargeable EVs are thus higher there.





The structure of the energy system is consequently crucial for revenue potentials for bidirectionally chargeable EVs. In European countries, structural characteristics of electricity production differ a lot. In regard to energy transition in Europe accompanied by a shift to

different volatile renewable production technologies, revenue potentials could vary even more in the future. For this reason, future revenue potentials in Germany are quantified and discussed in the following section.

#### 3.1.3. Revenue Potential for Future Day-Ahead Market Prices

In addition to an assessment of current revenue potentials, future revenue potentials are also important for an investment in a bidirectionally chargeable EV and corresponding EV supply equipment (EVSE). For an estimation of the changed revenues, price time series for future years from the DYNAMIS project are used [28] (see Section 6). The DYNAMIS project performs a dynamic and intersectoral evaluation of measures for the cost-efficient decarbonization of the energy system. The multi-energy system model ISAaR (Integrated simulation model for unit dispatch and expansion with regionalization) determines the design of the future energy system in a model-based way to be able to evaluate the measures [29].

Based on the multi-stage, exploratory assessment of measures and packages of measures, a climate protection scenario has been developed that aims to reduce greenhouse gas emissions by 95% by 2050. This scenario is characterized on the supply side by a cost-optimized provision of energy sources and on the application side takes into account the technology- and sector-specific boundary conditions and restrictions. The expansion of renewable energies is the most important measure. Green fuels (including all solid, liquid and gaseous fuels produced from biomass, renewable electricity or a combination of both) will increasingly be used from 2040 onwards in applications that can only be electrified at considerable expense. Domestic power-to-x technologies increase the available flexibility in the electricity system due to the good storage capacity of green fuels. Bidirectionally chargeable electric vehicles are not yet modeled in this scenario path and, therefore, their effects on the energy system could not be investigated there either.

One output of the model are hourly marginal costs representing day-ahead electricity prices for the years 2020 to 2050. The mean annual day-ahead price and its daily standard deviation are shown in Table 4. It can be seen that the level and in particular the standard deviation of the electricity price increases sharply. This is mainly due to the severely changed energy system that includes high capacities of renewable energies resulting in production surpluses. This results in increasing times with electricity prices of  $0 \notin MWh$ . On the other hand, rising fuel and carbon prices many times lead to very high electricity prices due to the unavoidable use of power plants with expensive marginal costs.

Table 4.Modeled mean day-ahead prices and standard deviation of day-ahead prices<br/>for the years 2020 to 2050 in Germany [28] compared to empirical prices in 2019<br/>[23].

Year	2020 (Modeled)	2030 (Modeled)	2040 (Modeled)	2050 (Modeled)	2019 (Real Prices)
Mean day-ahead price in €/MWh	46.3	61.2	63.8	80.4	37.7
Daily standard deviation of day-ahead price in €/MWh	5.0	8.7	15.8	25.9	9.0

The hourly price time series are transferred as input into the optimization model for bidirectionally chargeable EVs to estimate future revenue potentials in the day-ahead market. Figure 5 shows the revenues from 2020 to 2050 as a box plot. The black cross shows the mean revenues for the EVs considered. The top and bottom edges of the blue boxes indicate the 25th and 75th percentiles. The whiskers show the lowest and highest revenues, excluding outliers. Outliers that represent values that are 1.5 times bigger than the interquartile range are illustrated as red plus signs. Comparing the revenues of 2020 to those pointed out in Section 3.1.1 for 2019, mean revenues for the modeled prices are much lower than the mean revenues for the empirical data. Modeled electricity prices of energy system models often tend to be less volatile than real prices [30]. Consequently, lower price spreads lead to lower revenues.

Regarding modeled electricity prices for future years, mean revenues of bidirectionally chargeable EVs compared to unmanaged charging EVs increase by a factor of 5 to 6. This is mainly due to the future structural change of production units to renewable volatile production in combination with high carbon and fuel prices leading to many low electricity prices around  $0 \notin$ /MWh and many high electricity prices. Bidirectionally chargeable EVs can harvest the resulting high price spreads to generate revenues. Another interesting aspect is the range of revenues within a user group that increases considerably in future years, showing higher uncertainty of revenue potentials. For non-commuters in particular, there is a heterogenic distribution of revenues.

The results for future electricity prices show a much higher revenue potential than for current electricity prices. Regarding these results, one has to keep in mind that there are many uncertainties about the design of the future energy system and the resulting electricity prices. Furthermore, bidirectionally chargeable EVs will have a retroactive effect on electricity prices, reducing price spreads and revenue potentials. Nevertheless, the use case of arbitrage trading for bidirectional EVs will most certainly get more attractive in future years.





#### 3.2. Effect of V2G Use Cases on Full Cycles and Operating Hours

#### 3.2.1. Effect of Unrestricted Trading in the Electricity Markets

The V2G use case of arbitrage trading leads to higher usage of the battery of the EVs and relevant supplying equipment as well as information and communication technology. Relevant parameters that show the additional charge of EVs are full battery cycles and total operating hours. High yearly full cycles and operating hours in particular mean a faster ageing of the battery. For the revenue modeling in Section 3.1, we deliberately applied no restrictions on cycles or operation hours. In the BCM project, a separate model will be used for evaluation of the impact of V2G use cases on EV components.

Table 5 shows the impact of the EV operation in Section 3.1.1 on the EV parameter full cycles, revenues per full cycle and operating hours. For arbitrage trading, a strong increase in full cycles by 100–500 full cycles/a and in operating hours by 1500–5000 hours/a is determined. Revenues per full cycle are around 1 to 3 €/full cycle for arbitrage trading.

If one sets the highlighted parameter values in relation to currently warrantied lifetime values of lithium-ion batteries (e.g., 5000 to 6000 full cycles for residential storage systems [31,32] and a typical 10,000 operating hours in automotive applications [33]), it becomes clear that strong, relevant, additional loads of the battery arise for the use cases of arbitrage trading. These additional loads are significant, yet the use cases can still become economic
without overloading the battery. Since large battery systems (as in EV2 and EV3) do not have many full cycles from driving, an alternative usage of the battery is a logical addition.

Market	Affected		Commuters		Non-Commuters			
Modeling	EV Parameter	EV1	EV2	EV3	EV1	EV2	EV3	
Reference	Full cycles	60	25	25	35	15	15	
Unmanaged charge	Operating Hours	400	400	340	250	250	190	
	Full cycles	230	210	300	320	270	400	
Arbitrage: Day-ahead market	Revenues/ Full cycle	0.8	1.7	1.7	0.6	1.5	1.5	
	Operating Hours	1860	3900	2920	2710	5070	3930	
	Full cycles	490	270	490	640	340	630	
Arbitrage: Intraday auction	Revenues/ Full cycle	0.7	2.0	1.9	0.7	1.9	1.7	
udetion	Operating Hours	3760	4880	4660	4890	6180	5970	
Arbitrage:	Full cycles	450	250	470	590	320	600	
Continuous intraday	Revenues/ Full cycle	0.8	2.2	2.0	0.7	2.0	1.9	
trading	Operating Hours	3450	4670	4450	4490	5950	5680	
	Full cycles	440	240	450	570	300	570	
Arbitrage: Consecutive trading	Revenues/ Full cycle	1.0	2.6	2.5	0.9	2.4	2.3	
	Operating Hours	3280	4350	4110	4340	5590	5310	

Table 5.Impact of V2G use cases on EV's full cycles, revenues per full cycle and operating<br/>hours.

### 3.2.2. Effect of Restricted Trading in the Electricity Markets

Since modeled full cycles and operating hours in the previous section could critically decrease the lifetime of the EV's battery and power electronics, a minimum spread of electricity prices as a limit value for arbitrage trading could lower the EV's operation while still generating high profits. The minimum spread refers to the spread of sold to purchased energy. Consequently, the selling price for these simulations has to be lowered by the minimum, modeled price spread divided by the roundtrip efficiency.

Figure 6 illustrates the effect of a minimum price spread of 0 to 50 €/MWh on full cycles and operating hours compared to revenues of EV2. Revenues refer to the difference of revenues of bidirectionally chargeable EVs to the revenues of smart charging EVs to show only the added benefit by bidirectional charging. The decrease of full cycles and operating hours is displayed in percentage referring to the simulation with no minimum price spread. The maximum decrease of full cycles and operating hours arises when increasing the minimum price spread from 0 to 5 €/MWh, whereas revenues do not decrease significantly

for this change. For a minimum price spread of 10 €/MWh, additional full cycles for bidirectional charging of EV2 decrease by more than 50% to 95 full cycles per year for commuters and to 125 full cycles per year for non-commuters. For the same restriction, additional operating hours for bidirectional charging of EV2 also decrease by more than 50% to 1800 operating hours per year for commuters and to 2400 operating hours per year for non-commuters and to 2400 operating hours per year for non-commuters. For a minimum price spread of 10 €/MWh, revenues for both commuters and non-commuters decrease by only 20% to 240 €/EV/a and to 310 €/EV/a, respectively. By applying higher minimum price spreads, full cycles and operating hours further decrease and both the revenue per full cycle rate and revenue per operating hour rate increase. Full data for full cycles and operating hours of all EV scenarios is attached in Appendix D.

As a result, applying a minimum price spread is an effective method of limiting full cycles and operating hours while maintaining adequate profits. Even though revenues are generally decreased, this approach leads to an increase of the revenue per full cycle rate and revenue per operating hour rate and might thus represent a practicable approach for a future operation of bidirectionally chargeable EVs.



Figure 6. Effect of restricted minimum price spread on revenues, full cycles and operating hours of EV2 for commuters (a) and non-commuters (b).

# 3.3. Analysis of User Parameters and Regulatory Framework on Revenue Potentials of V2G Use Cases

The revenues of bidirectional charging are determined by a multitude of input parameters. We use German day-ahead prices in 2019 (Section 3.3.1 and Section 3.3.2) as well as German intraday auction prices in 2019 (Section 3.3.2) to show the influence of user and regulatory parameters on the revenue potentials of bidirectionally chargeable EVs.

### 3.3.1. Influence of User Parameters

### Minimum SoC at Departure

The user parameter  $SoC_{min,dep}$  describes the minimum battery storage level that has to be reached at a scheduled departure. Considering this restriction, a higher minimum SoC at departure leads to a reduction of flexibility regarding the bidirectional charging strategy due to the smaller range between  $SoC_{max}$  and  $SoC_{min,dep}$  in which the storage level can vary. This reduces the extent to which profitable price spreads and thus a revenue-maximizing discharge behavior can be used. The effects of different parameterization of  $SoC_{min,dep}$  on the revenue potential in the day-ahead market are shown for the six different EV pool scenarios in Figure 7a. Potential revenues with  $SoC_{min,dep}$  of 10% to 100% are compared to a reference of no minimum SoC at departure ( $SoC_{min,dep} = 0\%$ ).

All scenarios show an exponential decrease in revenues with an increasing  $SoC_{min,dep}$ . If flexibility is gradually increased starting from a  $SoC_{min,dep}$  of 100% up to a  $SoC_{min,dep}$  of 0%, the impact on revenues is highest at the first adjustment from 100% to 90%. This is because with this first increase in flexibility, the highest electricity prices present at the considered time can be used to discharge at great profit. As low prices are used for charging, a great specific profit can be made. As flexibility is further increased due to a lower minimum SoC on departure, lower spot prices are also increasingly used for discharging. The price spread between charging and discharging decreases, which means that specific profits are reduced and revenues change less. Consequently, if an EV user can define the parameter himself, he/she should choose the lowest possible minimum SoC on departure to maximize revenues. In particular, users should avoid selecting an unnecessarily high  $SoC_{min,dep}$ . For investigations in this paper, a realistic minimum SoC of 70% at departure is assumed after consultation with the project BCM.

## Minimum Safety SoC

The user parameter  $SoC_{min,safe}$  describes the minimum battery storage level that an EV always should have when connected in order to guarantee a drive to the hospital or other relevant short-distance routes at any time. If an EV arrives at a charging station with a lower SoC than  $SoC_{min,safe}$ , it will start charging immediately until it reaches the minimum parameterized SoC. A higher minimum safety SoC alike a higher minimum SoC at departure leads to a reduction of flexibility, since the useable capacity for marketing in the spot markets of  $SoC_{max} - SoC_{min,safe}$  decreases as the minimum safety SoC is increased. To quantify the impact of a minimum safety SoC, revenues with varied parameterization of  $SoC_{min,safe}$  are compared in Figure 7b showing the relative decrease of revenues compared to a reference with no safety SoC.

There is an exponential decrease of revenues for all EV pool scenarios depending on the minimum safety SoC. EV1 and EV3 have similar functions, while EV2 has a much smaller gradient for low safety SoCs and a steeper gradient for higher safety SoCs. This is due to the large battery capacity of EV2 and the fact, that the capacity of the charging/discharging power ratio (E/P) is much higher for EV2 at around 9, compared to the ratios of EV1 and EV3 at 3.5 and 4.5, respectively. A large battery capacity of the EV in combination with a

fixed SoC on departure makes it less likely that the vehicle comes back with a low SoC on arrival and thus it is less limited by a low safety SoC than an EV with a low capacity. A low E/P ratio, on the other hand, means that EV batteries can be charged and discharged quickly with a great cycle depth, which is limited even with a low safety SOC. EV2 rarely gets these low SoCs, so a low safety SoC has little effect.

Consequently, users with EVs that have a low E/P ratio should care more about a parameterization of a very low safety SoC than users that own an EV with a higher E/P ratio. In the investigations in this paper, a minimum safety SoC of 30% for EV1 representing the status quo for BMW electric vehicles and a lower safety SoC of 20% for EV2 and EV3 representing future electric vehicles are assumed.



**Figure 7.** (a) Influence of a minimum SoC at departure on revenue potentials of bidirectionally chargeable EVs using arbitrage trading; (b) influence of a minimum SoC at the place of residence on revenue potentials of bidirectionally chargeable EVs using arbitrage trading.

(b)

## Plug-in Probability

(a)

An EV user has the possibility to connect his vehicle to a charging station upon arrival at a location where a charging station is available. The probability that the user connects his vehicle to the charging station is called plug-in probability. There are several factors influencing plug-in probability. First, there is the incentive for a user to plug in his/her EV. Most importantly, the user is motivated to charge his EV for the next driving phase, where the desire could be to charge the EV directly or the possibility of smart or bidirectional charging. Since the incentives for EVs using bidirectional charging have hardly been investigated yet, there are no data on the plug-in probability of those EVs. Therefore, Figure 8a shows the influence of a changed plug-in probability on revenue potentials.

In all EV pool scenarios, there is a positive, approximately linear relationship between the plug-in probability and the revenues. If the EV is connected to a bidirectional charging

station more frequently, a discharge process that maximizes revenues can be carried out more often. The gradient of the curves differs in the various scenarios. The increase in revenues in this investigation varies between  $2 \in$  for EV1 and  $5 \in$  to  $6 \in$  for EV3 with a 1% increase in the plug-in probability. Hence, for bidirectionally chargeable EVs participating in the spot markets, a higher plug-in probability means equally higher revenues. For investigations in this paper, a plug-in probability of 100% is assumed under the assumption that using bidirectional charging for arbitrage trading is profitable and users are incentivized to plug-in their EVs.

### **Charging Point Location**

A bidirectional charging management system can only be operated if a bidirectional charging station is available at the EV's location. In accordance with the driving profiles from Section 2.4, possible locations for the EV are the place of residence, the place of work and the public space. An extensive expansion of bidirectional charging stations in public spaces is unlikely since the main reason for using public charging stations is the fast charging of the EV. Further analysis, therefore, concentrates on charging points at the place of residence and the place of work.

Figure 8b shows the revenue potentials of bidirectionally chargeable EVs depending on the charging point location. Comparing mean revenues at the place of residence and the place of work, there is a much higher incentive even for commuters to use bidirectional charging at the place of residence for the use case of arbitrage trading. This is due to the shorter period of time spent at the place of work while restrictions for minimum safety and departure SoC still have to be regarded. EVs can be discharged profitably less frequently than at the place of residence. In the non-commuter pool, 55% of the driving profiles are never located at a place of work, which is why an evaluation is not appropriate for these vehicle pools. If charging points are located both at the place of residence and at the place of work, revenues are slightly higher than if a charging point is only available at the place of residence. Compared to total revenues, the increase is quite low.

Consequently, the revenue potential of bidirectional charging at the place of work as opposed to the place of residence is low. Bidirectional charging should, therefore, be prioritized for EV users who are ready to install a bidirectional charging station at the place of residence. In the investigations within the framework of this paper, a charging station located at or near the place of residence is assumed.





(b)

Figure 8.(a) Influence of plug-in probability on revenue potentials of bidirectionally<br/>chargeable EVs using arbitrage trading; (b) influence of charging point location<br/>on revenue potentials of bidirectionally chargeable EVs using arbitrage trading.

### 3.3.2. Impact of Regulatory Framework on Revenue Potentials

Bidirectionally chargeable EVs are a new technology whose regulatory framework conditions have not yet been developed at the European Union (EU) level [34]. An essential question is whether bidirectionally chargeable EVs are classified as storage devices and, consequently, what additional charges they have to pay for charged electricity that is discharged later.

In Germany, the wholesale market price of electricity accounts for just around 15% (around 40 to 50 €/MWh) of the price of electricity for households. The other 85% (around 260 €/MWh) of the price is accounted for by additional charges such as the EEG surcharge (surcharge of electricity for remuneration of renewables) and grid fees as well as distribution [25]. Pumped storage facilities on the other hand are exempted from most of the EEG surcharge, grid fees and other levies, so that only additional charges of around 18 €/MWh have to be paid on electricity purchases [35], which can lead to a profitable arbitrage trading for storage facilities. As most of the exemptions refer to electricity purchases, storage losses are included [36].

For the use case of arbitrage trading, Figure 9 shows the influence of additional charges for purchased energy in the day-ahead or intraday market on revenue potentials of bidirectionally chargeable EVs for commuters and non-commuters. The solid lines represent revenues of bidirectionally chargeable EVs compared to unmanaged charging EVs and the dashed lines show revenues of smart charging EVs compared to the unmanaged charging ones. The diagram makes clear that possible revenues are strongly dependent on the regulatory framework. Starting from mean revenues with no additional charges at around 550 to 600 €/MWh for intraday auction trading, the revenues decrease by over 40% for additional charges of only 10 €/MWh. Non-commuters have an even

sharper decrease since the starting revenues are a bit higher and the minimum revenues representing smart charging are lower. Revenue potential of smart charging is higher because of the increased annual driving the associated increase in the annual charging demand. If a bidirectionally chargeable EV is regulatory-equal to a pumped storage facility with additional charges of 18  $\in$ /MWh on purchased electricity, mean revenue of non-commuters will decrease by over 60%, and commuters will have less than 50% revenue. For additional charges of more than 50  $\in$ /MWh, the added revenue of bidirectionally chargeable EVs compared to smart charging EVs is less than 20  $\in$ /EV/a.

Consequently, the future regulatory framework will decide if there is a chance for profitable arbitrage trading for bidirectional EVs. In investigations in this paper, additional costs are set to zero €/MWh to show the potential revenues for bidirectionally chargeable EVs.





## 4. Discussion

Bidirectionally chargeable EVs can use arbitrage trading to generate revenues for a potential improvement of their economic efficiency. Section 3.1 pointed out that revenues differ widely depending not only on the EV pool scenarios but also on the considered markets and the years under review. For an evaluation based on empirical prices of 2019 in Germany, revenues by bidirectional arbitrage trading range from 200 to 1300 €/EV/a

depending on EV pool and market participation. Profits on other European day-ahead markets compared to the German day-ahead market vary between +70% and –90% and are, therefore, highly dependent on the structure of the considered energy system. For future electricity prices, there is high revenue potential for bidirectionally chargeable EVs, which is pointed out by revenues for arbitrage trading in 2050 up to six times as high as the revenue potentials determined for 2020.

Most V2G related studies in literature refer to profits in reserve markets [37–39]. However, some studies deal with V2G profits of arbitrage trading. Peterson et al. point out revenues of 140 to 250 US\$/EV/a (120 to 210 €/EV/a) for 16 kWh EV batteries in local markets in three US cities [13]. Pelzer et al. determine 60 to 300 US\$/EV/a (50 to 250 €/EV/a) depending on spatial and temporal variation in US and Singapore markets under consideration of battery degradation costs [14]. These revenues have a similar level as our calculated EV1 revenues for day-ahead trading in Germany in 2019. We do not consider battery degradation costs but show the maximization of revenues by consecutive trading in day-ahead and intraday markets and the crucial influence of user parameters and regulatory framework on revenue potentials.

Concerning analyzed user parameters, our results confirm those of Szinai et al., who found that smart charging value at residential locations is much higher than at work or public locations [40]. In addition, we found this to also be true for bidirectionally chargeable EVs. Geske et al. indicate that the minimum range and range anxiety are the most important determinants for users participating in V2G use cases [41]. In this regard, we show the quantitative effect of the parameters 'minimum SOC at departure' and 'safety minimum SOC' on the revenue potentials of bidirectionally chargeable EVs, thereby addressing the minimum range and range anxiety. Increasing these parameters leads to an exponential decrease of revenues depending on the EV type. Hence, a tradeoff exists between the users' range anxiety and potential revenues.

With regard to the indicated high future revenues for arbitrage trading, one has to consider the retroactive effects that bidirectional EVs participating in the considered markets will have on market prices. As for arbitrage trading, where EVs charge when spot prices are low and discharge when spot prices are high, a flattening impact on spot prices is foreseeable. From a spot market perspective, offers of electric energy increase during times of high spot prices, and the demand for electricity increases during times of low prices, leading to higher prices when prices are low and lower prices when prices are high. These retroactive effects will lower revenue potentials if significant quantities of bidirectionally chargeable EVs participate in the markets. For quantitative evaluation of these retroactive effects, an energy system model is needed that models the supply and demand curves and thus can model price changes attributable to bidirectionally chargeable EVs. This additional research will be addressed by the BCM project in a following publication.

For an assessment of the impact of bidirectionally chargeable EVs in the markets, Table 6 shows the market volumes of considered and relevant markets and fitting EV quantities with 10 kW bidirectional charging station that would cover the market completely. Regarding the table and considering Germany's aim of 10 million EVs by 2030, the probable

retroactive effect in the markets can be derived. If a significant quantity of those EVs will have a bidirectional charging station and use V2G, there will be a high impact on prices in the intraday auction, and the quarter-hourly and the hourly continuous intraday trading market. A lower retroactive effect will occur for day-ahead prices in the German spot market.

Market	Average Market Volume in Germany 2019	EVs with 10 kW Charging Station to Completely Cover the Market		
Day-ahead market	26,000 MW (EPEX Spot) <sup>1</sup> 58,000 MW (German demand) <sup>2</sup>	2.6 mil (EPEX Spot) 5.8 mil (German demand)		
Quarter hourly intraday auction	800 MW <sup>1</sup>	80,000		
Hourly continuous intraday trading	4500 MW <sup>1</sup>	450,000		
Quarter hourly continuous intraday trading	800 MW <sup>1</sup>	80,000		

Table 6.Average market volumes and fitting EV quantities for complete covering of<br/>considered and relevant German markets.

<sup>1</sup> Average power trading based on data of EPEX Spot [23].

<sup>2</sup> Average demand calculated by net consumption in 2019 of 512 TWh [42] divided by 8760 hours.

Besides restrictions of market volumes, Section 3.2 points out the effect of V2G use cases on full cycles and operating hours of the EV's battery storage. For the use case of arbitrage trading, there is a high increase in battery usage resulting in an accelerated ageing of the battery. Although warranties for the full cycle lifetime of battery systems increase, the additional charge on the battery will reduce the use case's economic efficiency. However, in Section 3.2.2, we point out that both full cycles and operating hours can be decreased significantly, while revenues are still high by implementing a minimum price spread for arbitrage trading. In the BCM project, a detailed battery-ageing model is used for evaluating the impact of V2G on the battery and power electronics of the EV.

Another important restriction is the regulatory framework. Section 3.3.2 has shown the immense effect of additional charges on the profitability of the use case of arbitrage trading. It will be decisive if bidirectionally chargeable EVs are regulatory classified as a storage and if so, which additional charges will arise.

Regarding model limitations, constant values are set for the efficiencies of charging and discharging in order to achieve much faster optimization times. For arbitrage trading, charging and discharging processes usually use the highest possible power as price signals express either purchasing, selling or doing nothing. In a following publication in the BCM project, revenues of vehicle-to-home (V2H) use cases are compared by using constant and non-constant efficiencies, in which it is pointed out that modeling a non-constant efficiency for V2H use cases is necessary.

For the use case of arbitrage trading, we model revenues without perfect foresight for a rolling horizon of two to three days. The implemented market prices are real and not

forecasted prices, which are used for real trading. On the other hand, it can be assumed that price forecasts over- and underestimated actual market prices to the same extent and on an average alignment with real market prices resulting in a realistic modeling of revenues for bidirectionally chargeable EVs.

Finally, for the evaluation of the economy of V2G use cases, additional costs have to be considered. The main additional hardware costs result from a bidirectional charging station. Currently, the cost for the only bidirectional charging station soon available in the German market is around  $6000 \in [43]$ . Medium-term cost projections in the project BCM for a bidirectional charging station is around  $2000 \in$ . In regard to the current stage of the BCM project, cost projections for additional hardware and operating costs are not defined, and the economic efficiency of the use cases can, therefore, not yet be evaluated.

# 5. Conclusions

Based on the developed aggregated storage optimization model, revenues of bidirectionally chargeable EVs have been calculated for the V2G use case arbitrage trading. As a detailed description of optimization constraints and input data are provided, readers are able to reconstruct the indicated revenues of bidirectionally chargeable EVs. The major findings of this research are:

- We developed a rolling optimization model that regards real trading times of European spot markets and allows countertrading in consecutive traded markets while considering user behavior parameters leading to a realistic representation of revenue potentials of bidirectionally chargeable EVs using arbitrage trading.
- Revenues of bidirectionally chargeable EVs are dependent on user parameters. An increase of the safety minimum SoC at the place of residence or the minimum SoC at departure leads to an exponential decrease of revenues for bidirectionally chargeable EVs.
- For a participation of bidirectionally chargeable EVs in the German spot markets in 2019, potential revenues range from 200 to 1300 €/EV/a depending on the modeled EV pool scenario under the assumption of no additional charges for purchased electricity.
- Revenues of currently available EV models participating in the day-ahead market are comparable to findings of other literature, while our research shows a significant increase in revenues for consecutive trading in all spot markets.
- The regulatory framework concerning additional charges of purchased energy is the most decisive parameter for the potential revenues of bidirectionally chargeable EVs.
- Considering additional charges amounting for example to the payments of a pumped storage facility for bidirectionally chargeable EVs results in a decrease of revenues by 50% to 60%. Thus, if V2G arbitrage trading is supposed to give flexibility

to the future energy system, the market regulator will have to exempt bidirectionally chargeable EVs from the major part of additional charges.

- Unrestricted arbitrage trading of bidirectionally chargeable EVs results in a sharp increase of full cycles and operating hours by 200 to 600 full cycles/a, respectively, by 2000 to 6000 h/a resulting in much faster battery degradation. Restricted arbitrage trading with a minimum price spread can lower this additional load for EV and EVSE. For a minimum price spread of 10 €/MWh, operating hours and full cycles decrease by 50% while revenues only decrease by 20%.
- Revenues of bidirectionally chargeable EVs differ widely depending on the electricity production structure of the energy system. European day-ahead market revenues for EV2 in 2019 range from 50 €/EV/a in Norway to 700 €/EV/a in Ireland. Modeled potential future revenues are 2 times higher in 2030 and 5 to 6 times higher in 2050 than modeled revenues in 2020.

In general, potentially high revenue opportunities are identified for bidirectionally chargeable EVs in the electricity markets. Thus, participating in V2G use cases could promote electric mobility and, thereby, provide the flexibility needed for the energy system of the future. For future profitable usage of V2G use cases, the design of the regulatory framework and battery lifetime are decisive. For further investigations, especially retroactive effects of bidirectionally chargeable EVs on market prices and resulting decrease of revenue opportunities is of interest.

## 6. Data Availability

The driving profile data are available in supplementary material 'input and results data\driving profiles'. The modelled future electricity prices are available in 'https://openenergy-

platform.org/dataedit/view/scenario/ffe\_dynamis\_emission\_factors\_marginal\_cost´.

In Section 3.1 which analyzed revenue potentials of individual EVs for German spot market prices, European day-ahead market prices and future day-ahead market prices are available in supplementary material 'input and results data\revenues\_Section\_3\_1'.

Supplementary Materials: The following are available online at http://www.mdpi.com/1996-1073/13/21/5812/s1,

Driving Profiles data: 50 commuter and 75 non-commuter driving profiles including quarter-hourly resolved time series for location and consumption of individual EVs: http://opendata.ffe.de/dynamisemission-factors, Revenues Section 3.1: Individual user revenues depending on driving profiles as additional data for shown aggregated revenues in Section 3.1.

**Author Contributions:** Conceptualization, T.K.; Data curation, T.K.; Formal analysis, T.K. and P.D.; Investigation, T.K.; Methodology, T.K. and P.D.; Software, T.K.; Validation, T.K.; Visualization, T.K.; Writing—original draft, T.K. and P.D.; Writing—review & editing, T.K., P.D. and S.v.R. All authors have read and agreed to the published version of the manuscript.

**Funding:** The described work is conducted within the project BCM by Forschungsgesellschaft für Energiewirtschaft mbH and funded by Bundesminsterium für Wirtschaft und Energie (BMWi) under the funding code 01MV18004C.

**Conflicts of Interest:** The authors declare no conflict of interest. The funders had no role in the design of the study; in the collection, analyses, or interpretation of data; in the writing of the manuscript; or in the decision to publish the results.

## Appendix A

Table AT.	Та	ble	A1.	
-----------	----	-----	-----	--

List of time-dependent variables of the storage equation and their respective limit values.

Time-dependent V	Minimum Value	Maximum Value	
State of charge	SoC	$SoC_{min}$	SoC <sub>max</sub>
Charging power	$P_{charge}$	$P_{charge,min}$	$P_{discharge,max}$
Discharging power	$P_{discharge}$	$P_{discharge,min}$	$P_{discharge,max}$
Discharging boolean	$b_{charge}$	0	1
Charging boolean	$b_{discharge}$	0	1
Counter purchase power	$P_{counter-purchase}$	0	$P_{schedule,sale}$
Counter sale power	$P_{counter-sale}$	0	$P_{schedule,purchase}$
Counter purchase boolean	$b_{counter-purchase}$	0	1
Counter sale boolean	$b_{counter-sale}$	0	1
Supplementary power	$P_{supplement}$	0	$\infty$
Fast charging power	$P_{fastcharge}$	0	$\infty$

## Appendix B

## Influence of Forecast Period

For modeling realistic revenues of bidirectionally chargeable EVs, a rolling, limited time horizon is implemented. The selected limited time horizon is important for two reasons. Firstly, market prices are perfectly forecasted, which leads to a perfect EV charging strategy for the considered time horizon. As an example, if there are very low prices 10 days ahead of the starting point, the EV may shift its charging in the future. In reality, the low prices may depend on renewable energies that cannot be forecast 10 days in advance accurately. Secondly, EV driving behavior is perfectly forecast. In reality, regular driving (e.g., commuting to work) can be predicted most of the time, whereas more spontaneous driving (e.g., for free time activities) is much more uncertain. In this regard, it is hard to declare a realistic horizon for forecasting in the model.

The upper diagram in Figure A1 illustrates the effect of an adapted forecasting period on revenue potentials of bidirectionally chargeable EVs with no additional charges on purchased energy. Starting from a forecast horizon of ten days going down to four days, there is no change in revenues. A shorter forecast period results in slightly decreased revenues, but the impact of the forecast period on revenues is very low. This is mainly due to daily characteristics of the electricity price. Price spreads are used to charge with low prices and discharge with high prices with just a slight dependence on future departures or future electricity prices. Depending on the regulatory framework, additional charges can arise for purchased energy. The bottom diagram shows the effect of a varied forecast horizon on the revenues with additional charges of 100 €/MWh on purchased energy. There is huge decrease of revenues even resulting in negative revenues for short forecast periods, since EVs often do not know a future departure and consequently discharge although the future charging price is much higher. Only for forecast periods of 7 days and longer are revenues relatively stable compared to a horizon of 10 days.

The forecast horizon is mainly applied to prevent unrealistic future trading after the optimized period that is remunerated. For simulations with no additional charges, the rolling optimization is still necessary as it allows consideration of real sequential market trading. Consequently, for investigations in this paper a short forecast period of one day is defined. For an evaluation of the regulatory framework in Section 3.3.2, a longer forecast period of seven days is applied to prevent unrealistic discharging although charging prices are higher.



**Figure A1.** Influence of forecasting horizon on revenue potentials of bidirectionally chargeable EVs for the use cases of arbitrage trading with no additional charges (top) and additional charges of 100 €/MWh (bottom) on purchased energy.

## Appendix C

### Determination of a Realistic Pool Size

The modeling of a realistic vehicle pool is characterized by the EV pool size. The revenue potential of bidirectional charging of a single vehicle depends strongly on the individual driving profile of the user. In order to show a meaningful average revenue potential for different EV pool groups, a relevant number of vehicles must be determined for the model calculations. As computational time is linearly dependent on the number of profiles considered, a trade-off has to be faced. The aim of the investigations is to identify a pool size at which the addition of further vehicles has a low influence on revenues compared to the higher computational time. For all evaluated EV pool scenarios, the revenue potential on the German day-ahead market in 2019 was investigated for different pool sizes. For 5 to 150 profiles, a random drawing of 50,000 profile groups leads to a statistically significant analysis. Figure A2 shows the maximum revenue deviation for different numbers of profiles is 150 in order to limit computational time. If all 150 profiles are drawn, there will be no revenue deviation to the best case of 150 profiles.





Maximum revenue deviation for a varying number of profiles compared to the best case of 150 profiles for a drawing of 50,000 EV profile groups for commuters (**a**) and non-commuters (**b**).

The more vehicles that are modeled, the less is the maximum revenue deviation. A relevant pool size for the EV pool scenarios can be determined by the revenue difference falling below a defined threshold value. A maximum deviation of less than 5% per vehicle per year is assumed to be sufficiently accurate while limiting computational time. This results in a relevant pool size of 50 vehicles for a commuter EV pool. A representative non-commuter EV pool needs a number of 75 vehicles. Revenues of non-commuters are more heterogenic

than revenues of commuters because commuters have a more regular driving profile during weekdays.

# Appendix D

Table A2.

Revenues, full cycles and operating hours for all EV scenarios (difference of bidirectional charging to smart charging) with restricted minimum price spread.

EV1—Commuter								
Minimum Price spread in €/MWh	Revenues in €/EV/a	Full Cycles per Year	Operating Hours per Year	Average Price Spread in €/MWh	Revenue/ Full Cycle in €/Full Cycle	Revenue/ Operating Hour in €/Operating Hour		
0	125.1	231.0	1898	14.2	0.54	0.07		
5	117.7	166.7	1363	18.6	0.71	0.09		
10	97.8	102.9	841	25.0	0.95	0.12		
15	75.5	60.4	494	32.9	1.25	0.15		
20	59.7	38.7	319	40.6	1.54	0.19		
25	46.2	25.6	210	47.4	1.80	0.22		
30	37.1	18.6	153	52.6	2.00	0.24		
35	27.7	13.1	109	55.7	2.12	0.25		
40	17.2	8.0	67	56.8	2.16	0.26		
45	6.6	2.8	24	62.9	2.39	0.27		
50	2.7	0.9	7	77.2	2.93	0.35		

#### EV1—Non-Commuter

Minimum Price Spread in €/MWh	Revenues in €/EV/a	Full Cycles per Year	Operating Hours per Year	Average Price Spread in €/MWh	Revenue/ Full Cycle in €/Full Cycle	Revenue/ Operating Hour in €/Operating Hour
0	173.3	324.2	2737	14.1	0.53	0.06
5	162.5	226.7	1911	18.9	0.72	0.08
10	135.6	139.5	1173	25.6	0.97	0.12
15	107.1	85.0	712	33.2	1.26	0.15
20	86.1	56.1	472	40.4	1.54	0.18
25	68.7	38.8	327	46.6	1.77	0.21
30	56.5	29.0	245	51.3	1.95	0.23
35	44.2	22.1	189	52.7	2.00	0.23
40	33.7	16.5	140	53.9	2.05	0.24
45	19.9	10.1	87	51.9	1.97	0.23
50	10.8	5.5	48	51.4	1.95	0.23

Evz—Commuter						
Minimum Price Spread in €/MWh	Revenues in €/EV/a	Full Cycles per Year	Operating Hours per Year	Average Price Spread in €/MWh	Revenue/ Full Cycle in €/Full Cycle	Revenue/ Operating Hour in €/Operating Hour
0	296.1	211.4	3963	14.0	1.40	0.07
5	278.2	150.5	2819	18.5	1.85	0.10
10	236.1	96.8	1815	24.4	2.44	0.13
15	187.3	59.9	1123	31.3	3.13	0.17
20	152.5	41.3	766	37.0	3.70	0.20
25	125.0	30.3	561	41.3	4.13	0.22
30	102.1	23.5	434	43.4	4.34	0.24
35	77.2	17.5	322	44.0	4.40	0.24
40	50.3	11.9	216	42.3	4.23	0.23
45	22.9	5.2	93	44.3	4.43	0.25
50	8.6	1.6	30	52.9	5.29	0.29

## EV2—Commuter

#### EV2—Non-Commuter

Minimum Price Spread in €/MWh	Revenues in €/EV/a	Full Cycles per Year	Operating Hours per Year	Average Price Spread in €/MWh	Revenue/ Full Cycle in €/Full Cycle	Revenue/ Operating Hour in €/Operating Hour
0	383.4	270.7	5113	14.2	1.42	0.07
5	361.0	192.5	3645	18.8	1.88	0.10
10	307.0	125.4	2383	24.5	2.45	0.13
15	245.0	78.7	1498	31.1	3.11	0.16
20	200.7	55.2	1045	36.4	3.64	0.19
25	167.5	41.9	795	40.0	4.00	0.21
30	142.8	34.5	654	41.5	4.15	0.22
35	117.1	28.5	541	41.0	4.10	0.22
40	88.6	22.8	429	38.9	3.89	0.21
45	58.5	15.9	299	36.8	3.68	0.20
50	34.7	9.3	174	37.3	3.73	0.20

Minimum Price Spread in €/MWh	Revenues in €/EV/a	Full Cycles per Year	Operating Hours per Year	Average Price Spread in €/MWh	Revenue/ Full Cycle in €/Full Cycle	Revenue/ Operating Hour in €/Operating Hour
0	451.2	301.4	2994	15.0	1.50	0.15
5	430.6	227.5	2243	18.9	1.89	0.19
10	369.2	151.5	1487	24.4	2.44	0.25
15	290.4	92.4	898	31.4	3.14	0.32
20	231.8	60.6	582	38.3	3.83	0.40
25	187.6	42.6	402	44.0	4.40	0.47
30	152.0	31.9	297	47.7	4.77	0.51
35	123.6	24.5	227	50.4	5.04	0.54
40	85.1	17.1	156	49.8	4.98	0.55
45	41.7	8.5	75	49.2	4.92	0.55
50	16.7	2.8	25	59.1	5.91	0.66

#### EV3—Commuter

#### EV3—Non-Commuter

Minimum Price Spread in €/MWh	Revenues in €/EV/a	Full Cycles per Year	Operating Hours per Year	Average Price Spread in €/MWh	Revenue/ Full Cycle in €/Full Cycle	Revenue/ Operating Hour in €/Operating Hour
0	574.6	397.5	3979	14.5	1.45	0.14
5	545.8	291.1	2913	18.8	1.88	0.19
10	466.9	191.4	1916	24.4	2.44	0.24
15	370.3	118.3	1188	31.3	3.13	0.31
20	298.4	79.2	795	37.7	3.77	0.38
25	243.9	56.8	568	42.9	4.29	0.43
30	202.6	44.2	444	45.9	4.59	0.46
35	171.8	36.2	365	47.5	4.75	0.47
40	132.8	28.4	287	46.7	4.67	0.46
45	92.3	20.7	211	44.6	4.46	0.44
50	56.0	12.7	128	44.1	4.41	0.44

# References

1. BMU. *Klimaschutz in Zahlen—Fakten, Trends und Impulse deutscher Klimapolitik;* BMU: Berlin, Germany, 2018.

- 2. Fattler, S.; Regett, A. Environmental Impact of Electric Vehicles: Influence of Intelligent Charging Strategies. In *Grid Integration of Electric Mobility 2019;* Forschungsstelle für Energiewirtschafte e.V: München, Germany, 2019.
- 3. European Comission. A Clean Planet for All –A European Strategic Long-Term Vision for A Prosperous, Modern, Competitive and Climate Neutral Economy; European Commission: Brussels, Belgium, 2018.
- 4. Hinterstocker, M.; Ostermann, A.; Muller, M.; Dossow, P.; von Roon, S.; Kern, T.; Pellinger, C. *Bidirectional Charging Management—Field Trial and Measurement Concept for Assessment of Novel Charging Strategies*; Forschungsstelle für Energiewirtschaft e.V.: München, Germany, 2019.
- 5. Pham, T. Do German renewable energy resources affect prices and mitigate market power in the French electricity market ? *Appl. Econ.* **2019**, *51*, 1–14, doi:10.1080/00036846.2019.1624919.
- 6. EPEX SPOT SE. Trading on EPEX SPOT; EPEX SPOT SE: Paris, France, 2019.
- Illing, B.; Warweg, O. Analysis of international approaches to integrate electric vehicles into energy market. In Proceedings of the 12th International Conference on the European Energy Market (EEM), Lisbon, Portugal, 19–22 May 2015.
- Shafiullah, M.; Al-Awami, A.T. Maximizing the profit of a load aggregator by optimal scheduling of day ahead load with EVs. In Proceedings of the 2015 IEEE International Conference on Industrial Technology (ICIT), Seville, Spain, 17–19 March 2015.
- 9. Illing, B., Warweg, O. Achievable revenues for electric vehicles according to current and future energy market conditions. In Proceedings of the 13th International Conference on the European Energy Market (EEM), Porto, Portugal, 6–9 June 2016.
- 10. Bessa, R.J.; Matos, M.A.; Soares, F.J.; Lopes, J.A.P. Optimized Bidding of a EV Aggregation Agent in the Electricity Market. *IEEE Trans. Smart Grid* **2011**, *3*, 443–452.
- 11. Rominger, J.; Losch, M.; Steuer, S.; Koper, K.; Schmeck, H. *Analysis of the German Continuous Intraday Market and the Revenue Potential for Flexibility Options*; IEEE: Piscataway, NJ, USA, 2019.
- 12. Schmidt, R.; Schnittmann, E.; Meese, J.; Dahlmann, B.; Zdrallek, M.; Armoneit, T. *Revenue-Optimized Marketing of Electric Vehicles' Flexibility Options*; IEEE: Piscataway, NJ, USA, 2019.
- 13. Peterson, S.B.; Whitacre, J.; Apt, J. *The Economics of Using Plug-In Hybrid Electric Vehicle Battery Packs for Grid Storage*; RELX Group: London, UK, 2009.
- 14. Pelzer, D.; Ciechanowicz, D.; Knoll, A. Energy arbitrage through smart scheduling of battery energy storage considering battery degradation and electricity price forecasts. In Proceedings of the IEEE Innovative Smart Grid Technologies Asia (ISGT-Asia), Melbroune, Australia, 28 November–1 December 2016.
- 15. Hanemann, P.; Bruckner, T. Effects of electric vehicles on the spot market price. *Energy* **2018**, *162*, 255–266.
- 16. Rodriguez, C.D.B.; Segura, J.A.J.; Ruiz, M.C.C.; Estevez, G.A.J.; Araya, P.A.M. *Evaluating the Impact of a V2G Scheme on the Demand Curve*; IEEE: Piscataway, NJ, USA, 2019.
- 17. Kern, T.; Hintersrocker, M.; von Roon, S. Rückwirkungen von Batterie-Vermarktungsoptionen auf den Strommarkt. In Proceedings of the 11 Internationale Energiewirtschaftstagung an der TU Wien (IEWT 2019), Vienna, Austria, 13–15 February 2019.
- Technical Specifications BMW i3 (120 Ah). Available online: https://www.press.bmwgroup.com/global/article/attachment/T0284828EN/415571 (accessed on 7 October 2020).
- 19.
   New
   Renault
   Zoe
   2017.
   Available
   online:

   https://www.renault.com.au/sites/default/files/pdf/brochure/ZOE-Brochure-Desktop\_0.pdf
   (accessed on 7 October 2020).
- 20. infas Institut für angewandte Sozialwissenschaft GmbH. *Mobilität in Deutschland 2017–Datensatz*; infas Institut für angewandte Sozialwissenschaft GmbH: Bonn, Germany, 2019.
- 21. Pellinger, C.; Schmid, T. *Merit Order der Energiespeicherung im Jahr 2030—Hauptbericht*; Forschungsstelle für Energiewirtschaft e.V. (FfE): München, Germany, 2016.
- 22. Fattler, S.; Böing, F.; Pellinger, C. Ladesteuerung von Elektrofahrzeugen und deren Einfluss auf betriebsbedingte Emissionen. In Proceedings of the IEWT 2017, 10 Internationale Energiewirtschaftstagung Wien, Vienna, Austria, 15–17 February 2017.
- 23. Power Market Data. Available online: https://www.epexspot.com/en/market-data/ (accessed on 26 August 2020).
- 24. Day-Ahead Prices. Available online: https://transparency.entsoe.eu/transmissiondomain/r2/dayAheadPrices/show (accessed on 31 August 2020).

- 25. BDEW. *BDEW-Strompreisbestandteile Januar 2020—Haushalte und Industrie*; BDEW Bundesverband der Energie-und Wasserwirtschaft e.V.: Berlin, Germany, 2020.
- 26. Platts. WEPP Database (Europe); Platts: Washington, DC, USA, 2018.
- 27. International Energy Agency. *Energy Policies of IEA Countries France 2016 Review*; International Energy Agency: Paris, France, 2017.
- 28. FfE. Hourly CO<sub>2</sub> Emission Factors and Marginal Costs of Energy Carriers in Future Multi-Energy Systems (Germany); Forschungsstelle für Energiewirtschaft e. V. (FfE): München, Germany, 2019. Available online: http://opendata.ffe.de/dataset/dynamis-emission-factors/ (accessed on 28 August 2020).
- 29. Böing, F.; Regett, A. Hourly CO<sub>2</sub> Emission Factors and Marginal Costs of Energy Carriers in Future Multi-Energy Systems. *Energies* **2019**, *12*, 2260.
- 30. Beran, P.; Pape, C.; Weber, C. *Modelling German Electricity Wholesale Spot Prices with a Parsimonious Fundamental Model—Validation and Application*; House of Energy Markets and Finance: Duisburg-Essen, Germany, 2018.
- 31. BYD Company Limited. Battery Box 2.5. Available online: https://www.climaverd.com/sites/default/files/2019-06/ft\_sb\_byd\_b-box\_2.5-10.0\_b-plus\_2.5\_en.pdf (accessed on 27 August 2020).
- 32. BMZ Energy Storage Systems. Maximum Performance for Your Independence. Available online: https://bmz-group.com/images/PDF-Downloads/Broschuere-ESS\_EN\_with\_ESSX.pdf (accessed on 27 August 2020).
- 33. Ma, K.; Yang, Y.; Wang, H.; Blaabjerg, F. Design for Reliability of Power Electronics in Renewable Energy Systems, In *Use, Operation and Maintenance of Renewable Energy Systems*; Sanz-Bobi, M.A., Ed.; Springer: Basel, Switerzland, 2014.
- 34. EASE. *Energy Storage: A Key Enabler for the Decarbonisation of the Transport Sector*; EASE—European Association for Storage of Energy: Brussels, Belgium, 2019.
- 35. Conrad, J.; Pellinger, C.; Hinterstocker, M. *Gutachten zur Rentabilität von Pumpspeicherkraftwerken*; Forschungsstelle für Energiewirtschaft e.V. (FfE): München, Germany, 2014
- 36. Bundesnetzagentur für Elektrizität, Gas, Telekommunikation, Post und Eisenbahnen. *Regelungen zu Stromspeichern im deutschen Strommarkt*; Bundesnetzagentur für Elektrizität, Gas, Telekommunikation, Post und Eisenbahnen: Bonn, Germany, 2020.
- 37. Sandelic, M.; Stroe, D.-I.; Iov, F. Battery Storage-Based Frequency Containment Reserves in Large Wind Penetrated Scenarios: A Practical Approach to Sizing. *Energies* **2018**, *11*, 3065.
- 38. Ciechanowicz, D.; Knoll, A.; Osswald, P.; Pelzer, D. *Towards a Business Case for Vehicle-to-Grid—Maximizing Profits in Ancillary Service Markets*; Springer: Singapore, 2015.
- 39. Clairand, J.M. Participation of electric vehicle aggregators in ancillary services considering users' preferences. *Sustainability* **2020**, *12*, 8.
- 40. Szinai, J.K.; Sheppard, C.J.; Abhyankar, N.; Gopal, A.R. Reduced grid operating costs and renewable energy curtailment with electric vehicle charge management. *Energy Policy* **2020**, *136*, 111051, doi:10.1016/j.enpol.2019.111051.
- 41. Geske, J.; Schumann, D. Willing to participate in vehicle-to-grid (V2G)? Why not! *Energy Policy* **2018**, *120*, 392–401, doi:10.1016/j.enpol.2018.05.004.
- 42. BDEW. *Energy Market Germany 2020*; Bundesverband der Energie- und Wasserwirtschaft (BDEW): Berlin, Germany, 2020.
- 43. Besserladen. Wallbox Quasar—Bidirektionale Ladestation. Available online: https://besserladen.de/produkt/wallbox-quasar-bidirektionale-ladestation/ (accessed on 28 August 2020).

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



 $\bigcirc$  2020 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).

Kern, T., Dossow, P., & Morlock, E. (2022). Revenue opportunities by integrating combined vehicle-to-home and vehicle-to-grid applications in smart homes. Applied Energy, 307. https://doi.org/10.1016/j.apenergy.2021.118187

©2021 by the authors. Published by Elsevier Ltd.

## Publication 2 (Pub2): Revenue opportunities by integrating combined vehicle-to-home and vehicle-togrid applications in smart homes

### Timo Kern<sup>1,2,3</sup>, Patrick Dossow<sup>1</sup> and Elena Morlock<sup>1</sup>

- <sup>1</sup> Forschungsgesellschaft für Energiewirtschaft mbH (FfE), 80995 Munich, Germany
- Department of Electrical and Computer Engineering, Technical University of Munich (TUM), Arcisstraße
   21, 80333 München, Germany
- <sup>3</sup> Corresponding author, tkern@ffe.de

Abstract: A smart integration of electric vehicles (EVs) in the future energy system will be crucial in decarbonizing the energy sector. Bidirectional EVs can provide flexibility for the system and generate revenues for the user through multiple use cases. We model both exclusive photovoltaic (PV) self-consumption optimization and the combined usage of PV self-consumption optimization and arbitrage trading for a household with an unmanaged, smart, and bidirectional charging EV in a linear (LP) and mixed-integer linear programming (MILP). Since power flows in a typical household are low, varying non-linear charging and discharging efficiencies of the bidirectional EV in the MILP result in more realistic revenues that are 30% lower than in the LP with fixed efficiencies. For a typical German household using a bidirectional EV for optimizing PV self-consumption, these revenues are about 310 €/a, mostly generated during the summer. Arbitrage trading well complements this vehicle-to-home use case in the winter months, resulting in revenues up to 530 €/a. These significant revenue potentials can lead to more profitable and interactive EVs incentivizing users to change from internal combustion vehicles to electric mobility.

Keywords:	PV	self-consumption	optimization;	arbitrage	trading;	bidirectional	electric
vehicles; V2	2G; \	/2H					

Nomenclature			
Abbreviations		$p_{el,sell,v2g}$	electricity selling prices for V2G
BCM	bidirectional charge management	$P_{EV,l,const,c/d}$	constant charging/discharging losses
BCM	Bidirectional Charging Management	$P_{EV,l,const,st}$	constant standby losses of EV and
	(project)		EVSE
COM	commuter	<b>P</b> <sub>HH,el</sub>	electrical household demand
EFC	equivalent full cycle	$P_{HH,th}$	thermal household demand
EV	electric vehicle	$P_{PV}$	PV generation
EVSE	electric vehicle supply equipment	SoC <sub>dep</sub>	minimum SOC at departure
GCP	grid connection point	SoC <sub>safe</sub>	minimum SOC when connected
HP	heat pump	t	timestep
LP	linear programming	Т	total timesteps
MILP	mixed-integer linear programming	х	start timestep of V2G interval
OH	operating hour	у	end timestep of V2G interval

SBS	stationary battery storage			
SoC	state of charge	Variables		
V2B	vehicle-to-business	$\boldsymbol{b}_{EV,c}$	boolean variable if EV is charging	
PV	photovoltaic			
V2G	vehicle-to-grid	$\boldsymbol{b}_{EV,d}$	boolean variable if EV is discharging	
V2H	vehicle-to-home	$E_{EV}$	charge level of EV battery	
		$E_{EV,pub,c}$	public charging energy	
Parameters		P <sub>EV,c</sub>	charging power of EV	
$b_{EV,dep}$	timeseries if EV is departing	$P_{EV,d}$	discharging power of EV	
C <sub>connected</sub>	timeseries if EV is connected to the	$P_{EV,l,c}$	charging losses of EV	
	EVSE			
$E_{EV,drive}$	EV consumption while driving	$P_{EV,l,d}$	discharging losses of EV	
$m_c$	gradient of charging losses	$P_{EV,l,s}$	standby losses of EV and EVSE	
$m_d$	gradient of discharging losses	$P_{GCP,in}$	power from grid	
n <sub>c</sub>	minimum charging losses	$P_{GCP,in,v2g}$	power from grid as V2G process	
$n_d$	minimum discharging losses	P <sub>GCP,out</sub>	power to grid	
$\eta_{EV,roundtrip,max}$	maximum roundtrip efficiency of EV	$P_{GCP,out,v2g}$	power to grid as V2G process	
$\eta_{SBS,roundtrip}$	roundtrip efficiency of SBS	P <sub>HP,el</sub>	power consumption of heat pump	
$p_{el,buy}$	electricity purchase prices for	P <sub>PV,curt</sub>	curtailment of PV generation	
	household			
$p_{el,buy,v2g}$	electricity purchase prices for V2G	$P_{SBS,c}$	charging power of SBS	
$p_{el,sell}$	feed-in tariff	$P_{SBS,d}$	discharging power of SBS	

## 1. Introduction

The electrification of mobility is often considered an essential component in combatting climate change. While CO<sub>2</sub> emissions in the German energy sector have decreased sharply, because of the strong expansion of renewable energies, emissions in the transport sector have remained roughly the same as they were in 1990 [1]. Coupling of the energy and mobility sectors is seen as a major opportunity for reducing emissions in the transport sector, with electric mobility playing a key role [2]. The German government has introduced some subsidies, e.g. for private individuals purchasing electric cars, to increase the share of electric vehicles (EVs) on German roads [3]. Nevertheless, the target of one million registered electric vehicles in Germany by 2020 was missed. A survey by the German Association of Energy and Water Industries found that the high investment costs for an EV are the main argument against switching to electromobility [4]. If the economic viability of EVs could be increased, it would provide an additional incentive for citizens to purchase an electric vehicle.

Here, one possibility is use of bidirectional charging technology. In contrast to unidirectional charging systems, bidirectional charging systems not only allow energy to be drawn from the grid or a generation plant to charge the electric vehicle, but they also allow the energy from the vehicle to be fed back in a smart form. Therefore, during periods of inactivity, the vehicles can also be used in a manner analogous to a stationary battery storage (SBS) [5]. In vehicle-to-grid (V2G) applications, bidirectionally chargeable EVs can contribute to grid stability in a system-serving manner [6] while offering economic benefits for EV owners [7].

Similarly, vehicle-to-home (V2H) use cases offer the benefit of optimized use of locally generated renewable energy while also providing revenue opportunities; here some studies have already addressed the profitability of bidirectional charging. Salpakari et al. show that smart and bidirectional charging can save 8-33% of annual electricity costs compared to an unmanaged charging strategy in a household fitted with a photovoltaic (PV) system in

Sweden. The additional cost savings from vehicle-to-microgrid are small if battery degradation costs are taken in consideration [8]. Chen et al. study energy use optimization strategies without and with V2H for a household with a PV system in Shanghai, taking into consideration time-varying electricity tariffs with high and low tariff time windows. They show that economic benefits can be achieved for the household in all use cases. PV tariff, weather, and EV driving behavior are the key influencing factors here [9]. Erdinc et al. compare bidirectional charging and unmanaged charging for both V2H and V2G use cases for a household with a small PV system in Portugal, again using time-varying tariffs. In their calculations, they obtain a cost reduction potential of up to 48% by limiting power at peak times and up to 63% for time-variable tariffs by bidirectional charging compared to unmanaged charging [10]. Kataoka et al. evaluate the effect of V2H applications on the economic and environmental performance of a typical household in Japan. They find that V2H can be economically and environmentally more beneficial than SBSs, but results differ between commuters and non-commuters. For future work, they suggest further sensitivity analysis, e.g. regarding the size of EV battery or EV charge and discharge power [11]. The cost-effectiveness of a V2H system in Germany is investigated by Cacilo et al.. Yet, no economic evaluation of the vehicle's bidirectional capability, only smart charging, is performed [12]. Keiner et al. analyze smart homes for an average German single-family household including heat pumps (HPs), thermal energy storage systems, SBSs and EVs and find that V2H can assume the role of an SBS [13]. However, they largely focus on different thermal energy storages scenarios and not different EV charging strategies. The abovementioned studies model fixed charging and discharging efficiencies and thus neglect energy losses resulting from low-powered, ineffective charging and discharging processes in V2H use cases [14]. Further, there is a lack of in-depth research in the literature investigating the influence of different household component set-ups on cost reduction potentials through the use of smart or bidirectional charging EVs.

Regarding V2G use cases, there are numerous papers that deal with the revenue potentials of EVs participating in the spot market [15]. Smart charging optimized by electricity prices can reduce charging costs [16], [17] and bidirectional charging can further reduce these costs or even generate revenues [18]. None of the studies mentioned deal with V2H use cases complementing arbitrage trading in spot markets. Since arbitrage trading often results in lots of operating hours (OHs) and equivalent full cycles (EFCs) of the battery [19], the temporary replacement by V2H use cases could reduce the battery ageing effect.

In this paper, we address the aforementioned research gaps by modeling V2H use cases exclusively with and without varying charging and discharging efficiencies. This novel evaluation of more realistic modeling of V2H as mixed-integer linear programming (MILP) with varying efficiencies versus modeling of V2H as linear programming (LP) with fixed efficiencies is necessary to assess whether MILP with varying efficiencies is beneficial or even necessary for V2H analyses. For this purpose, an optimization model is developed that optimizes the electric power flows of a household with the objective of minimizing electricity costs while taking technical restrictions into account. Three different charging strategies are compared regarding household energy flows: an unmanaged charging strategy, a smart charging strategy and a bidirectional charging strategy. The V2H revenue

potentials are shown based on the influencing factors of PV size and PV feed-in tariff, household size, household components, EV size, maximum charging power and maximum operating hours. Finally, we model a combined application of V2H and V2G arbitrage trading to show differences and benefits on the electricity costs of a household. This combined, novel modeling of V2H and V2G can realistically combine the seasonally different revenue opportunities of the use cases and shows that a separate modeling of a V2H or V2G use case can underestimate the revenue opportunities. The investigations are embedded in the project 'Bidirectional charging management' (BCM) that analyzes the technical, economic, and regulatory issues of bidirectional charging [5].

## 2. Methodology

The methodological approach for investigating V2H use cases of bidirectional EVs is divided into two parts: first, we present the optimization model for considering V2H cases exclusively. Then, the combined modeling of V2H and V2G is outlined.

We developed an optimization model in order to assess revenue potential efficiently and accurately in the area of V2H applications. The model determines the best possible charging strategy for one or multiple EVs connected via a grid connection point (GCP) with multiple optional other components. This approach allows a wide range of different analyses, including V2H, V2G and V2B use cases. Figure 1 shows a schematic of the model structure. As displayed, the model structure *eFlame* (electric Flexibility assessment modeling environment) itself consists of two sub-models. Firstly, the household profile generator, where the electrical and thermal consumption of a specific household are modeled based on parameterization of the household and user behavior, modeling as well appropriate EV driving profiles [20]. Secondly, the optimization model *ResOpt* (Residential Optimizer) formulates the objective function and constraints of variable defined components.



Figure 1: Schematic representation of the developed optimization model for use cases of bidirectional charging

The model ResOpt comprises several decision variables, the values of which are set for each time step in the course of the optimization: SBS charging  $P_{SBS,c}$  and discharging  $P_{SBS,d}$ , PV curtailment  $P_{PV,curt}$ , HP demand  $P_{HP,el}$ , EV charging  $P_{EV,c}$ , discharging  $P_{EV,d}$  and standby losses  $P_{EV,l,s}$  as well as power from grid  $P_{GCP,in}$  and power to grid  $P_{GCP,out}$ . For the combined modeling of V2H and V2G use cases (Section 2.2), we integrate the decision variables  $P_{GCP,in,v2g}$  and  $P_{GCP,out,v2g}$  that represent power from and to the grid that is additionally constrained. A fixed thermal  $P_{HH,th}$  and electrical  $P_{HH,el}$  household demand is provided as well as the fixed PV generation  $P_{PV}$ .

### 2.1. Modelling of V2H applications with fixed and varying efficiencies

The objective of V2H optimization is to minimize the household's electricity costs, which is expressed in the objective function in Equation 1. The German household's electricity costs to be minimized consist of the purchase costs  $p_{el,buy}$  multiplied by the purchased power  $P_{GCP,in}$  minus the feed-in tariff  $p_{el,sell}$  multiplied by the power  $P_{GCP,out}$ , which is fed into the grid if an electricity generator is present, over all time steps t.

$$min\left(\sum_{t=1}^{T} \left[p_{el,buy}(t) \cdot P_{GCP,in}(t) - p_{el,sell}(t) \cdot P_{GCP,out}(t)\right]\right)$$
(1)

Relevant constrains are implemented for the household's grid connection point, the EV and other optional components. We refrain from a detailed description of the constraints for optional components such as SBS or HP at this point. Most importantly, the power fed into the grid or supplied from the grid must equal the sum of power generated and consumed within the boundaries of the system at any time:

$$P_{GCP,in}(t) - P_{GCP,out}(t) = P_{HH,el}(t) + P_{HP,el}(t) - P_{PV}(t) + P_{PV,curt}(t) + P_{SBS,c}(t) - P_{SBS,d}(t) + P_{EV,c}(t) - P_{EV,d}(t) + P_{EV,l,st}(t)$$
(2)

Standby losses  $P_{EV,l,st}(t)$  represent constant losses of the inverter and EV  $P_{EV,l,const,st}$ , which occur when the EV is connected  $C_{connected}(t)$  at the household's charging point (electric vehicle supply equipment, EVSE) and is neither charging  $b_{EV,c}(t)$  nor discharging  $b_{EV,d}(t)$ .

$$P_{EV,l,st}(t) = P_{EV,l,const,st} \cdot \left[ \mathcal{C}_{connected}(t) - b_{EV,c}(t) - b_{EV,d}(t) \right]$$
(3)

To account for such losses, the connection status  $C_{connected}$  (t) (a time series which is 1 if the EV is connected and 0 if it is not connected) and the discrete boolean variables  $b_{EV,c}$  (1 if charging, otherwise 0) and  $b_{EV,d}$  (equals 1 if discharging, else 0) are used. Thus, standby losses can only be included for MILP.

Another relevant constraint concerns the electric energy stored in an EV at time t, which is determined by Equation 4.

$$E_{EV}(t) = E_{EV}(t-1) + [P_{EV,c}(t) - P_{EV,l,c}(t)] \cdot \Delta t + E_{EV,pub,c}(t) -E_{EV,drive}(t) - [P_{EV,d}(t) + P_{EV,l,d}(t)] \cdot \Delta t - P_{EV,l,const,c/d} \cdot [b_{EV,c}(t) + b_{EV,d}(t)] \cdot \Delta t$$
(4)

Here, energy losses affecting the EV's state of charge (SoC) occur during charging and discharging. Again, to maintain linear programming (LP) optimization, constant losses  $P_{EV,l,const,c/d}$  would have to be neglected and charging  $P_{EV,l,c}$  and discharging  $P_{EV,l,d}$  losses modeled in proportion to charging and discharging power. As the EVSE in the case of bidirectional charging contains an additional inverter converting alternating current (AC) into direct current (DC) if charging and DC to AC if discharging, EVSE losses for bidirectional charging are modeled as variable over time in our work.

For this purpose, we draw on mathematical descriptions of the inverter efficiency deduced in [21], where inverter power losses are expressed as a quadratic function of the corresponding output power. By rearranging the respective equations as described in Appendix A, the inverter losses can be stated as functions of the AC-side power for both charging and discharging processes. As implementation of these non-linear loss functions directly resulted in unacceptably long computation times, we adopted a linear approximation approach. We linearized the power losses equations for charging and discharging by linear regression (method of least squares) as suggested by [22] in a similar context, which results in a linear system of equations:

$$P_{EV,l,c}(t) = m_c \cdot P_{EV,c}(t) + n_c \cdot b_{EV,c}(t)$$
<sup>(5)</sup>

$$P_{EV,l,d}(t) = m_d \cdot P_{EV,d}(t) + n_d \cdot b_{EV,d}(t)$$
(6)

Here,  $m_{c/d}$  is the gradient of the function and  $n_{c/d}$  represents the minimum losses at zero power.

By dividing the possible range of charging and discharging power into a number of equally large intervals and applying the method of least squares to each interval individually, the residual sum of squares can be reduced to improve accuracy. Hence, we conducted preliminary simulations to determine a suitable number of intervals for sufficiently high accuracy with acceptable computation time. The resulting deviations obtained when linearizing power losses for zero, one, or two intervals are presented in Table 1 (sum of constant EV losses and varying inverter losses), where zero losses correspond to constant losses. Since respective variations of efficiency are small in any case apart from the simulations with zero intervals, we implemented linearized functions of inverter power losses for both charging and discharging for a single interval  $[0; P_{EV,c/d,max}]$  based on real inverter data to limit the complexity of the optimization problem.

 Table 1:
 Deviation between real and linearized losses and efficiencies for a 11 kW EVSE

	Charging process			Discharging process		
Number of intervals	0 (LP)	1	2	0 (LP)	1	2
Maximum deviation of power losses	250 W	42 W	10 W	250 W	55 W	14 W

Additional constraints are related to the EV's SOC. If an EV is connected at the household's EVSE, the minimum amount of energy, represented by  $SoC_{safe}$ , must be reached at all times, which is ensured by Equation 7. Here,  $E_{EV,max}$  is the maximum amount of energy to be stored in the EV (i.e. the battery's capacity). Equation 8 guarantees that at the time of departure, which must be set when arriving at the household, a fixed minimum amount of energy, represented by  $SoC_{dep}$ , is stored in the EV's battery. To do so,  $b_{EV,dep}$  is introduced, which is 1 if t is the time of departure. Otherwise,  $b_{EV,dep}$  equals 0.

$$E_{EV}(t) \ge SoC_{safe} \cdot E_{EV,max} \cdot C_{connected}(t)$$
(7)

$$E_{EV}(t) \ge SoC_{dep} \cdot E_{EV,max} \cdot b_{EV,dep}(t)$$
(8)

Optional features of the model are the limitation of the EV battery's number of EFCs or the EV's OHs per day, which can be used to reduce battery aging or respectively the wear of the EV's power electronics. Equation 9 limits the maximum EFCs ( $EFC_{max}$ ) and Equation 10 the average maximum OHs per day ( $OH_{max,day}$ ). Since OHs are related to the boolean variables  $b_{EV,c}$  and  $b_{EV,d}$ , Equation 10 can only be used for MILP.

$$EFC_{max} \ge \frac{\sum_{t=1}^{T} ([P_{EV,c}(t) - P_{EV,l,c}(t)] \cdot \Delta t + E_{EV,pub,c}(t))}{E_{EV,max}}$$
(9)

$$OH_{max,day} \ge \frac{\sum_{t=1}^{T} [b_{EV,c}(t) + b_{EV,d}(t)]}{T} \cdot 24$$
(10)

### 2.2. Modelling of combined V2H and V2G applications

Adding V2G arbitrage trading to the model leads to several adaptions in the objective functions and constraints of the optimization problem. The regulatory framework for arbitrage trading is not yet defined for bidirectionally chargeable EVs at the European Union level [23]. Since power purchased and sold through arbitrage trading by SBS is exempted from multiple duties and taxes [24], we assumed this exemption for V2G such that modeled V2G prices differ from the normal household prices for purchased and feed-in energy. Therefore, the objective function is expanded in Equation 10 by the costs and revenues of

the V2G component, considering V2G prices  $p_{el,buy,V2G}(t)$  and  $p_{el,sell,V2G}(t)$ , and V2G power  $P_{GCP,in,v2g}(t)$  and  $P_{GCP,out,v2g}(t)$ .

$$min\left(\sum_{t=1}^{T} \left[ p_{el,buy}(t) \cdot P_{GCP,in}(t) - p_{el,sell}(t) \cdot P_{GCP,out}(t) + p_{el,buy,v2g}(t) \cdot P_{GCP,in,v2g}(t) - p_{el,sell,v2g}(t) \cdot P_{GCP,out,v2g}(t) \right] \right)$$
(10)

The energy purchased  $P_{V2G,in}(t)$  and feed-in  $P_{V2G,out}(t)$  is added to the power balance of the household grid connection point in Equation 11.

$$P_{GCP,in}(t) - P_{GCP,out}(t) + P_{GCP,in,v2g}(t) - P_{GCP,out,v2g}(t) = P_{HH,el}(t) + P_{HP,el}(t) - P_{PV}(t) + P_{PV,curt}(t) + P_{SVS,c}(t) - P_{SBS,d}(t) + P_{EV,c}(t) - P_{EV,d}(t) + P_{EV,l,st}(t)$$
(11)

In contrast to a SBS, bidirectionally chargeable EVs consume electricity by driving and thus charging and discharging energy do not balance. Since V2G electricity prices will only be exempted from multiple duties and taxes if the same amount of purchased energy is fed back into the grid at a different time, we add constraints to the V2G powers in Equation 12. The purchased energy  $sum\left(\sum_{t=1}^{T} \left[P_{GCP,in,v2g}(t)\right]\right)$ equals the fed-in energy  $sum\left(\sum_{t=1}^{T} \left[ P_{GCP,out,v2g}(t) \right] \right)$ divided by the maximum V2G roundtrip efficiency  $\eta_{EV,roundtrip,max}$ , because losses of a V2G roundtrip are included in the exemption from additional electricity charges [24]. The roundtrip efficiency refers to the charging and discharging efficiency of the bidirectionally chargeable electric vehicle (from AC to AC). The timeframe of the power equation t=x to y can be set variable depending on the regulatory framework of the considered household.

$$\left(\sum_{t=x}^{y} \left[ P_{GCP,in,v2g}(t) \right] \right) = \left(\sum_{t=x}^{y} \left[ P_{GCP,out,v2g}(t) \right] \right) / \eta_{EV,roundtrip,max}$$
(12)

To ensure that the purchased and fed-in energy is associated with the EV and not a different component of the household, we add Equations 13 and 14. Purchased energy  $P_{GCP,in,v2g}(t)$  has to be lower than charged electricity  $P_{EV,c}(t)$ . In addition, fed-in energy  $P_{GCP,out,v2g}(t)$  must be lower than discharging of the EV  $P_{EV,d}(t)$ .

$$P_{GCP,in,v2g}(t) \le P_{EV,c}(t) \tag{13}$$

$$P_{GCP,out,v2g}(t) \le P_{EV,d}(t) \tag{14}$$

Finally, considering a household with PV generation leads to possibilities of misuse when V2G energy is exempted from duties and taxes. In such a case, PV energy could first be charged into the EV and later be discharged into the grid as a V2G process (duties and taxes exemptions). As the EV is allowed to purchase the corresponding amount of energy described in Equation 12 as a V2G process, such an EV could purchase energy at a different time which would replace purchases of energy at household electricity prices. In this way, as much PV energy as desired can be fed through the EV into the grid in order to purchase the corresponding amount of energy at much lower prices at a different time. To prevent this effect, only V2H or V2G may be performed within a set period. Equations 15 and 16 constrain the optimization problem for a V2G time frame. If boolean variable  $b_{V2G}$  is set to 0 for a time interval t = x to y, V2G is not allowed. If  $b_{V2G}$  is set to 1, discharged energy of the EV must be fed to the grid and is not allowed to be fed to the household. Thus, V2H self-consumption optimization is not allowed for the respective time interval.

$$P_{GCP,out,v2g}(t) \le b_{EV,v2g} \cdot P_{EV,d,max}$$
(15)

$$P_{EV,d}(t) \le P_{GCP,out,v2g}(t) + (1 - b_{EV,v2g}) \cdot P_{EV,d,max}$$
(16)  
with  $b_{EV,v2g} = [0,1]$  for time interval  $t = x$  to  $y$ 

## 3. Results

Every scenario study includes three EV operation strategies: unmanaged charging, smart charging, and bidirectional charging. Revenues of a bidirectional or smart charging strategy are always compared to the reference of unmanaged charging.

### 3.1. Input data and set-up of average household scenario

To evaluate the economic viability of the use case 'self-consumption optimization' by bidirectional electric vehicles, a medium-sized single-family house (150 m<sup>2</sup> living space) is defined, which is subsequently referred to as the base scenario. Table 1 shows the main characteristics of the household and potential additional components. These optional components include a PV system, an EV, a SBS and a HP. Table 2 shows the parameterization of these components, additional input for their modeling, and the associated references. Prices and profiles are based on data from the year 2018.

Concerning the household profiles, Appendix B shows that 20 discrete profiles are sufficient to represent average revenues of a household class. Therefore, 20 profiles are used below, having on average the characteristics shown in Table 2. We focus on EVs that are used by non-commuters, since approximately 75% of EVs in Germany are non-commuting EVs [25], but show the revenues for a commuting EV as a sensitivity in chapter 3.2.

Element	Parameter	Value	Additional Input	
Household	$\sum P_{HH,el} \sum P_{HH,th}$	3800 kWh (yearly)* 9000 kWh (yearly)*	Load profile* Load profile*	
Grid	p <sub>el,buy</sub>	29.9 ct/kWh [26]		
PV system	$\begin{array}{c} P_{PV,max} \\ \sum P_{PV} \\ \mathcal{P}_{el,sell} \end{array}$	5.5 kWp*** 6200 kWh (yearly) 11.6 ct/kWh [27]	Generation profile**** based on [28]	
EV	$\begin{array}{l} P_{EV,c,max} \\ P_{EV,d,max} \\ E_{EV,max} \\ \eta_{EV,roundtrip,max} \\ \text{Annual mileage} \\ \text{EV consumption} \\ (\text{including charge losses}) \end{array}$	11 kW** 11 kW** 60 kWh***** 85%** 10,000 km* 22 kWh/100 km**	Driving profile*	
EV user	User type SoC <sub>safe</sub> SoC <sub>dep</sub>	Non-Commuter 20%** 70%**		
SBS	P <sub>SBS,c,max</sub> P <sub>SBS,d,max</sub> E <sub>SBS,max</sub> η <sub>SBS,roundtrip</sub>	2.8 kW 2.8 kW 5.5 kWh***** 88%		

Table 2.	elevant narameterization of elements connected to the household
	וטעצפווטע.

\*Average load value, annual driving, load profile and driving profile are output from household profile generator [20] and provided in supplementary material

\*\*Based on specifications of BCM project [5]

\*\*\*Calculated by usable roof space multiplied by specific energy 0.15 kWp/m<sup>2</sup> [31]

\*\*\*\*Provided in supplementary material

\*\*\*\*\*E.g. Volkswagen ID.3 Pro [32] / Opel Ampera-e [33]

\*\*\*\*\*\* Capacity and power of SBS based on PV power [34]

### 3.2. Revenues for V2H operation

Various sensitivity calculations are evaluated to assess the benefits of bidirectionally chargeable EVs for household owners. Figure 2 compares the revenues and PV self-consumption of smart and bidirectional charging strategies for the base scenario to the linear base scenario (LP optimization with constant efficiencies), the commuter EV scenario 'COM' and scenarios with the additional components HP and SBS.

Additional revenues for the base scenario are around 210  $\notin$ /a for a smart charging EV and around 310  $\notin$ /a for a bidirectionally chargeable EV. These revenues represent cost reductions of 25% to 36% compared to a household with an unmanaged charging EV. Self-consumption of electricity from the PV system increases on average from 23% for unmanaged charging to 45% for smart charging and 65% for a bidirectionally chargeable

EV. This improved use of cheaper PV energy rather than energy from the grid leads to higher revenues and a higher degree of autarky.

The 'Base linear' scenario shows a similar PV self-consumption increase, but significantly higher revenues for both smart and bidirectional charging compared to the base scenario. Since the household power demand and the PV power are both below 1 kW in 90% respectively 73% of all timesteps, where a varying inverter efficiency would drop significantly, the fixed efficiency of the EV's charging and discharging leads to a higher utilization with much lower losses. A more detailed description of the resulting differences in charging and discharging behaviour between fixed and varying efficiency is provided in Appendix C. PV self-consumption in the linear base scenario is slightly lower than in the non-linear base scenario due to lower charging and discharging losses. The diverging revenues of around 100  $\epsilon/a$  for a bidirectionally chargeable EV compared to the base scenario (divergence of 32%) show that for our considered use case, it is not valid to model a LP with fixed efficiencies. Therefore, all of the following results are based on a MILP optimization with a varying efficiency for charging and discharging of the EV.

In the commuter EV scenario 'COM', we model a commuter EV instead of a non-commuter EV in the household. The average annual mileage increases from 10,000 km/a to 20,000 km/a and the average availability of the EVs at the charging point decreases from 91% to 67%. For smart charging EVs, the average commuter EV generates only slightly lower revenues compared to the revenues of the smart charging EV in the base scenario. The lower availability of the commuter EV is compensated by the higher EV consumption that leads to more possibilities of smart charging. In contrast, the bidirectionally chargeable, commuting EV generates revenues that are about 100  $\notin$ /a lower than the non-commuting EV's revenues. The lower availability of the commuting EV during the day leads to fewer opportunities of charging PV electricity. Therefore, the PV electricity is used for consumption of the EV, but not for discharging to the household.

Adding additional smart components to the household leads to lower revenues for managed EV operation strategies, since these components also use the cheaper PV energy to optimize the household's electricity costs. In these scenarios, the reference case of unmanaged EV charging is already much more efficient than in the base scenario leading to a lower revenue potential of bidirectionally chargeable EVs. In scenario HP, the heat pump can be operated flexibly such that part of the low-cost PV electricity is used for heat generation. In particular, the battery storage in the SBS scenario acts similarly to the bidirectional electric vehicle, such that the revenue potential there is even more limited. The combination of these additional smart components in the household then provides the lowest revenue potentials. The impact on revenue potential is in turn linked to the level of self-consumption of the solar energy for the different EV operating strategies. With additional smart components in the household, a higher self-consumption is already apparent with unmanaged charging of the EV, such that the possible increase is limited.



Figure 2: Revenues and PV self-consumption for base MILP scenario in comparison to base linear scenario, commuter scenario COM and added components scenarios HP, SBS, and HP SBS

Other relevant factors potentially influencing revenue potential include:

- EV battery capacity
- Maximum power of EVSE
- PV peak power
- PV feed-in tariff
- Household size
- Maximum OHs of EV and EVSE

Figure 3 shows the effect of parameter variation on the revenue potential for smartly and bidirectionally chargeable EVs. The use case of self-consumption optimization is highly sensitive. In particular, the design of the PV system has a strong impact on the revenues. A large PV system with a low feed-in tariff generates significantly higher revenues through both smart and bidirectional charging of an electric vehicle. The parameterization of the EV and the EVSE has much less effect. In this case, larger designs cause no or only small increases in revenue. As for the EV, the selected capacity in the base scenario is sufficient to exploit the major part of revenue potentials for the fixed household and PV configuration. The small effects of the EVSE are related to the configuration of the household in the base configuration (maximum PV feed-in 4.5 kW, maximum demand household 7.7 kW). An EVSE of 22 kW instead of 11 kW has no effect on revenues. In contrast, the household size has a large impact on the revenue potential for bidirectionally chargeable EVs, since a larger household with higher demand power enables more efficient discharging of the EV.

Additionally, we evaluate a limitation of the resulting OHs of the bidirectionally chargeable EV reducing the impact on battery ageing and the additional load on power electronics. OHs of the bidirectionally chargeable EV in the base scenario without limitation are

around 6.9 h/d compared to 0.7 h/d for the unmanaged charging EV. EFCs are less affected and increase only by 18 EFC/a from 34 to 52 EFC/a for the bidirectionally chargeable EV. Since OHs are much more affected than EFC, we present constrained OHs per day to show the effect of limited usage on revenues. A reduction of OHs by 64% to 2.5 h/d results in a moderate revenue reduction of 19%. EFCs of the battery then reduce from 52 EFC/a to 41 EFC/a.





Lastly, for a maximum revenue estimate, we examine a household that has the best suited configuration of each of the analyzed parameters. An average annual household demand of 5900 kWh is combined with a 9.5 kWp PV system, which receives a feed-in tariff of 3.6 ct/kWh, and an EV with a battery capacity of 100 kWh. This household set-up receives revenues of 835  $\leq$ /a for a bidirectionally chargeable EV and revenues of 390  $\leq$ /a for a smart charging EV, showing the maximum potential revenues for the use case self-consumption optimization.

## 3.3. Revenues for combined V2H and V2G operation

Since self-consumption optimization achieves more profits when more PV energy is generated, V2H is more useful in summer times than in winter times. To benefit from the bidirectionally chargeable EV in the best possible way, we model a combined use of V2H and V2G arbitrage trading. For explained regulatory reasons, the EV is not allowed to optimize self-consumption and arbitrage trading at the same time. Instead, the optimizer can switch the use case of the bidirectionally chargeable EV daily.

For arbitrage trading, we use German day-ahead market prices from 2018 [35] matching the parameterization of the other components. Duties and taxes for purchased energy are not added to the electricity prices, because the regulatory framework for bidirectionally chargeable EVs is not yet fully defined. The revenues shown are thus an upper estimation.

Table 3 compares potential revenue of a combined optimization of V2H and V2G with and without restricted OHs to the revenues of the base V2H scenario. Revenue potentials of unrestricted, combined V2H and V2G increase by 220 €/a (plus 71%) compared to the base scenario, while EFCs increase by 270 EFCs/a and OHs by 5.5 OHs/d, meaning a significant additional load for the EV's battery. Limiting the maximum OHs to 10 h/d barely reduces revenues. In this case, EFCs are reduced by 70 EFCs/a relative to the unconstrained scenario, which is still an increase of 200 EFCs/a compared to the base V2H scenario. Limiting the maximum OHs to 5 h/d leads to revenues, which are 100 €/a higher than for the base scenario, yet 120 € lower than for the unrestricted combined scenario. Here, the EV usage is reduced significantly (only 70 EFCs/a more than the base scenario).

	Scenario		V2H base	V2H base + V2G	V2H base + V2G max. 10 OHs/day	V2H base + V2G max. 5 OHs/day
	Average	Total	310	530	510	410
	revenues	V2H	310	300	300	270
_	in €/a	V2G	-	230	210	140
	EFCs/a		50	320	250	130
_	OHs/d		6.9	12.5	10	5

Table 3:Revenues, operation hours and EFCs for bidirectional operating strategy of the<br/>EV

The generally increased revenues show that V2G arbitrage trading represents a very good complement to V2H use cases. To further analyze the combination of these use cases, Figure 4 shows the share of daily usage of V2H and V2G for the evaluated 20 households in the restricted optimization by 5 OHs/day as well as the weekly standard deviation of electricity prices and the weekly PV generation. The usage of V2H correlates to the PV generation that is the highest from April to September. In contrast, the daily standard deviation of day-ahead prices does not vary on a seasonal basis, meaning that daily arbitrage trading is on average equally profitable in summer and winter times. Therefore, V2G complements V2H in winter times. Since Figure 4 shows the OHs restricted optimization, there are days on which no V2G and V2H is used, which are days of low revenue potential.





Daily share of households using V2H and V2G correlated to weekly standard deviation of electricity prices and weekly PV generation

## 4. Discussion on results and limitations

Our results show that smartly and bidirectionally chargeable EVs can reduce a household's electricity costs significantly by optimizing the self-consumption of PV energy. Profits of the V2H use case are highly sensitive to components of the household and their parameter variations. While the design of the EV and EVSE has a small impact on the revenues of the use case, the higher the maximum PV power and the lower the PV feed-in tariff, the more profitable V2H is. Our simulations show revenues of about 300  $\in$ /a for a typical German household with varying efficiencies (MILP). Potential cost reductions for households with bidirectionally chargeable EVs in a scenario best fitted for maximum cost reduction of 390  $\in$ /a. Furthermore, we show that commuting EVs are not well suited for bidirectional self-consumption optimization but should rather only use smart charging. The average revenues indicated in this paper are slightly higher than revenues in the literature relating to V2H [8], [11], [13]. However, we show that revenues related to smart and bidirectional charging EVs are strongly sensitive to parameterization of household components.

By comparing results modeled as MILP with varying efficiencies to modeling with fixed efficiencies (LP), we found that it is highly important to model V2H use cases for bidirectional charging as MILP with varying charging and discharging efficiencies. Since for V2H applications only low charging and discharging powers are needed due to low household power demand and low PV system generation power, in reality low charging and discharging efficiencies occur at many times. Thus, modeling with a fixed efficiency (LP) led to revenues that are over 30% higher than the more realistic results of MILP, which is why we recommend to model V2H use cases exclusively with varying efficiencies.

All revenues presented are based on the German regulatory framework. However, through the sensitivity analyses, detailed conclusions can be drawn about V2H revenue potentials in other countries. The sensitivity analyses show that the difference between household electricity price and feed-in tariff is the most important influencing factor. According to Figure 3, a higher feed-in tariff of almost 20 ct/kWh results in 60% lower revenues, i.e. approx. 120 €/a for an average household. In a country with a price spread between household electricity price and feed-in tariff of 10 ct/kWh, these strongly reduced revenues can be expected. Similar conclusions can be drawn with regard to the country-specific revenues for V2G applications, which we discuss in detail in [19]. In this way, the detailed findings of the sensitivity analyses can be transferred very well to the conditions in other countries.

With the novel combined modeling of the use cases self-consumption optimization (V2H) and arbitrage trading (V2G), we show that the two use cases are highly complementary in terms of potential revenues. Since the V2H use case is strongly dependent on the PV generation, cost reduction options are high during the summer months. V2G arbitrage trading, where revenues do not alter significantly over the course of a year, can thus make greater use of the bidirectionally chargeable EV in the wintertime, resulting in an increased utilization of the vehicle. Dependent on the permitted usage of the EV, revenues of the base scenario increase from a maximum of  $220 \notin/a$  to  $530 \notin/a$ . As this increase shows, our more complex yet more realistic modeling highlights the great economic potential of combined V2H and V2G. As implementing multiple use cases at a time is likely to become common for EV users in the near future, our results show the future perspective of such multi-use-implementation for greater flexibility and higher revenues.

OHs and EFCs of the EV's battery increase for V2H use cases, but far less than for arbitrage trading [19]. We show that limited use of the EV of 2.5 h/d still generates high V2H revenues, which are only 19% below the unconstrained revenues. Comparing these limited OHs of around 900 h/a and EFCs of 41 EFC/a to currently warrantied lifetime values for battery and power electronics in automotive applications, which are around 10,000 OHs [36] and up to 5,000 EFCs [37], [38], additional OHs are more critical than additional EFCs, but V2H is still suitable as a use case for EVs. In this context, we want to emphasize the trade-off that although V2G arbitrage trading can generate significant additional revenues, it also leads to significant additional OHs and EFCs.

For the arbitrage trading, modeled electricity prices for selling and purchasing energy are equal. Depending on the regulatory framework there might be some additional duties and taxes for purchased energy, making the use case less profitable. Therefore, the presented revenues for arbitrage trading in the day-ahead market are to be interpreted as an upper bound of revenues. For a more detailed revenue estimation of arbitrage trading in European electricity markets, we refer to Kern et al. [19].

Finally, an economic evaluation of V2H and V2G use cases must include the additional investment costs of a bidirectional EVSE. To bring the presented revenue potentials into perspective for both present and future circumstances, we roughly estimate the economic viability of V2H by including additional annual costs via the annuity method [39]. Currently,

there are only few offers for bidirectional EVSEs suitable for on low-volume production, resulting in high investment costs of around €6,000 [40]. The medium-term cost projections of experts in the BCM project for such an EVSE are around €2,000 [5]. Assuming an EVSE lifetime of 15 years [41], an interest rate of 3.5% [42] and unmanaged charging EVSE costs of €599 [43] leads to additional annual costs of the bidirectional EVSE of currently 469 €/a and medium term 122 €/a. A comparison with the V2H revenues of the typical German household of 310 €/a shows that V2H will most probably not be economical for this household at current costs but is likely to become profitable in the medium term. However, this is only a rough estimate and other additional costs of the bidirectional EVSE and potential additional costs for the bidirectional EVSE and potential additional costs for the bidirectional evsE and potential additional costs for the bidirectional evsE.

# 5. Conclusions

Vehicle-to-home (V2H) use cases and a combination of V2H and vehicle-to-grid (V2G) use cases can be highly beneficial for electric vehicle (EV) users. We provide a detailed description of the modeling and input data, that allows readers to reconstruct the revenues for these use cases. The major findings of our study are:

- Indicated revenues for an average German household are around 310 €/a for bidirectional charging and 210 €/a for smart charging compared to an unmanaged charging EV, increased to 830 €/a and 390 €/a respectively for a maximum revenue estimation.
- Revenues of V2H use cases should consider varying charging and discharging efficiencies in a mixed-integer linear programming, since modeled fixed efficiencies in a linear programming led to 30 % higher revenues and an unrealistic charging/discharging behavior.
- Revenues of bidirectionally chargeable EVs are highly case-sensitive depending on the composition of a household. The dimensioning of the photovoltaic system and the household size are more decisive than the size of the EV.
- Smart additional components in a household, such as heat pumps and stationary battery storages, significantly limit the revenues of a smartly chargeable or bidirectionally chargeable EV.
- V2G arbitrage trading works well with V2H self-consumption optimization because arbitrage trading revenues do not depend on the time of the year, while V2H primarily generates revenues during the summer.
- Combined use of V2H and V2G leads to maximum additional revenues of 220 €/a, but increases operating hours of the EV by more than 5 h/d and equivalent full cycles (EFCs) by as much as 270 EFCs/a.
**Funding:** The described work is conducted within the project BCM by Forschungsgesellschaft für Energiewirtschaft mbH and funded by Bundesminsterium für Wirtschaft und Energie (BMWi) under the funding code 01MV18004C.

**CRediT authorship contribution statement:** Timo Kern: Conceptualization, Methodology, Investigation, Formal analysis, Software, Validation, Data curation, Writing – original draft, Writing – review & editing, Supervision. Patrick Dossow: Conceptualization, Writing – review & editing, Investigation. Elena Morlock: Methodology, Visualization.

**Declaration of Competing Interest:** The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

**Acknowledgement:** The authors would like to thank Sabine Englberger for implementing the initial structure of residential optimization model ResOpt in her master thesis as well as Alexander Djamali, Steffen Fattler and Mathias Müller for their valuable input in further model developments.

### Appendix A

A key element of the presented model development is the varying efficiency of the charging and discharging process depending on the respective power. In the case of bidirectional charging, losses are neither constant nor directly proportional to the charging or discharging power due to the additional inverter at the charging point. Thus, we describe the derivation of the relevant equations for the MILP in detail. According to [21], power losses of an inverter  $P_l$  consist of a constant self-consumption  $v_{const}$ , voltage losses at diodes and transistors  $v_l$  that are proportional to the output power, and quadratic powerdependent losses caused by ohmic loss resistances  $v_q$ . As the efficiency of the inverter  $\eta$  is the ratio of output power  $P_{out}$  to input power  $P_{in}$ , we formulate the following equation:

$$\eta = \frac{P_{out}}{P_{in}} = \frac{P_{out}}{P_{out} + P_l} = \frac{P_{out}}{P_{out} + v_{const} + v_l \cdot P_{out} + v_q \cdot P_{out}^2}$$
(A.1)

We express the efficiency for charging  $\eta_c$  and discharging  $\eta_d$  as a function of the AC-side power. For charging, AC-power before the inverter is converted into DC-power. Thus,  $P_{in}$ equals  $P_{AC,c}$  and  $P_{out}$  equals  $P_{DC,c}$ . We express  $P_{DC,c}$  as a function of  $P_{AC,c}$  (Equation A.2) to derive Equation A.4. For discharging, directions are reversed, such that  $P_{in}$  equals  $P_{DC,d}$ and  $P_{out}$  equals  $P_{AC,d}$  resulting in Equation A.3 and A.5.

$$P_{DC,c} = P_{AC,c} - P_{l,c} = P_{AC,c} - (v_{const,c} + v_{l,c} \cdot P_{DC,c} + v_{q,c} \cdot P_{DC,c}^{2})$$

$$= \frac{-(v_{l,c} + 1) + \sqrt{(v_{l,c} + 1)^{2} - 4 \cdot v_{q,c} \cdot (v_{const,c} - P_{AC,c})}}{2 \cdot v_{q,c}}$$
(A.2)

$$P_{DC,d} = P_{AC,d} + P_{l,d} = P_{AC,d} + (v_{const,d} + v_{l,d} \cdot P_{DC,d} + v_{q,d} \cdot P_{DC,d}^{2})$$

$$= \frac{-(v_{l,d} - 1) + \sqrt{(v_{l,d} - 1)^{2} - 4 \cdot v_{q,d} \cdot (v_{const,d} + P_{AC,d})}}{2 \cdot v_{q,d}}$$
(A.3)

$$\eta_{c} = \frac{P_{DC,c}}{P_{AC,c}} = \frac{-(v_{l,c}+1) + \sqrt{(v_{l,c}+1)^{2} - 4 \cdot v_{q,c} \cdot (v_{const,c} - P_{AC,c})}}{2 \cdot v_{q,c} \cdot P_{AC,c}}$$
(A.4)

$$\eta_{d} = \frac{P_{AC,d}}{P_{DC,d}} = \frac{2 \cdot v_{q,d} \cdot P_{AC,d}}{-(v_{l,d} - 1) + \sqrt{(v_{l,d} - 1)^{2} - 4 \cdot v_{q,d} \cdot (v_{const,d} + P_{AC,d})}}$$
(A.5)

Based on these equations, we express the inverter power losses of charging and discharging as a function of the respective AC-power (Equation A.6 and A.7). These equations constitute the basis for the linearization described for implementation in this work.

$$P_{l,c} = P_{AC,c} - P_{DC,c} = P_{AC,c} - \frac{-(v_{l,c} + 1) + \sqrt{(v_{l,c} + 1)^2 - 4 \cdot v_{q,c} \cdot (v_{const,c} - P_{AC,c})}}{2 \cdot v_{q,c}}$$
(A.6)

$$P_{l,d} = P_{DC,d} - P_{AC,d} = \frac{-(v_{l,d} - 1) + \sqrt{(v_{l,d} - 1)^2 - 4 \cdot v_{q,d} \cdot (v_{const,d} + P_{AC,d})}}{2 \cdot v_{q,d}} - P_{AC,d}$$
(A.7)

### Appendix B

To determine the number of profiles for a representative mapping of a household group, a number of profiles is randomly drawn from a maximum of 150 profiles. This is done 10,000 times per number of profiles. The calculated mean values of these 10,000 draws are compared to the mean value of the 150 profiles and the maximum deviation of these mean values is displayed in Figure B1. The deviations of the mean revenues are a maximum of 5% for 20 profiles, which is tolerated as a maximum deviation. Therefore, in the following, all studies are based on 20 randomly drawn, discrete household profiles.



Figure B1: Relative maximum deviation of mean revenues in dependance on number of modeled profiles

# Appendix C

Figure C1 shows the power flows at the GCP for a typical summer day. Power in the direction of the GCP is shown as positive. This includes EV discharging, PV generation and grid supply. Power flowing away from the GCP is shown as negative. This includes EV charging, household demand and grid feed-in (electricity supplied to the grid). For LP with fixed efficiency, household demand is balanced by EV discharging at night. For MILP with varying efficiencies, household demand is mainly balanced by grid supply, because the discharging efficiency of the EV is too low for these small power demands.



Figure C1: Power flows at the GCP for a typical summer day

# References

- 1. Klimaschutz in Zahlen: Sektorenziele 2030. Berlin: Bundesministerium für Umwelt, Naturschutz, Bau und Reaktorsicherheit (BMUB), 2018.
- Robinius, Martin; Otto, Alexander; Syranidis, Konstantinos; Heuser, Philipp; Welder, Lara; Ryberg, David et al.: Linking the Power and Transport Sectors—Part 1: The Principle of Sector Coupling. In: Energies Juli/2017. Jülich, Germany: Institute of Electrochemical Process Engineering (IEK-3), Forschungszentrum Jülich GmbH, 2017.
- 3. Bundesregierung: Klimaschutz Verkehr. In: https://www.bundesregierung.de/bregde/themen/klimaschutz/verkehr-1672896. (accessed on 2020-04-05); Berlin: Bundesregierung, 2020.
- 4. Fakten und Argumente: Meinungsbild E-Mobilität. Berlin: BDEW Bundesverband der Energieund Wasserwirtschaft e.V., 2019.
- Hinterstocker, Michael; Müller, Mathias; Kern, Timo; Ostermann, Adrian; Dossow, Patrick; Pellinger, Christoph et al.: Bidirectional Charging Management – Field Trial and Measurement Concept for Assessment of Novel Charging Strategies. München: Forschungsstelle für Energiewirtschaft e.V. (FfE), 2019.

- Weiß, Andreas; Müller, Mathias; Franz, Simon: Spitzenlastkappung durch uni- und bidirektionales Laden von Elektrofahrzeugen und Analyse der resultierenden Netzbelastung in Verteilnetzen. In: Forschung im Ingenieurwesen, Volume 85(2). 469-476. Berlin: Springer Nature, 2021. https://doi.org/10.1007/s10010-020-00424-z.
- 7. Lindegaard, Søren Bernt; Juhl, Lasse Thorbøll; Høj, Jens Christian Morell Lodberg: V2G—An Economic Gamechanger in E-Mobility? In: World Electric Vehicle Journal August/2018. Horsens, Denmark: Insero Energy Innovation, 2018.
- Salpakari, Jyri; Rasku, Topi; Lindgren, Juuso; Lund, Peter: Flexibility of electric vehicles and space heating in net zero energy houses: an optimal control model with thermal dynamics and battery degradation. In: Applied Energy 190/2017. Espoo, Finland: New Energy Technologies Group, Department of Applied Physics, School of Science, Aalto University, 2017.
- 9. Chen, Jianhong; Zhang, Youlang; Li, Xinzhou; Sun, Bo; Liao, Qiangqiang; Tao, Yibin et al.: Strategic integration of vehicle-to-home system with home distributed photovoltaic power generation in Shanghai. In: Applied Energy 263/2020. Shanghai: Nanjing Branch of China Electric Power Research Institute, Shanghai Key Laboratory of Materials Protection and Advanced Materials in Electric Power, Shanghai Engineering Research Center of Electric Energy Conversion, Shanghai University of Electric Power, School of Electric Power, South China University of Technology, 2020.
- 10. Erdinc, Ozan; Paterakis, Nikolaos; Mendes, Tiago; Bakirtzis, Anastasios; Catalão, João: Smart Household Operation Considering Bi-Directional EV and ESS Utilization by Real-Time Pricing-Based DR. In: IEEE TRANSACTIONS ON SMART GRID Vol. 6, No. 2, Mai/2015. ohne Ort: IEEE, 2015.
- Kataoka, Ryosuke; Shichi, Akira; Yamada, Hiroyuki; Iwafune, Yumiko; Ogimoto, Kazuhiko: Comparison of the Economic and Environmental Performance of V2H and Residential Stationary Battery: Development of a Multi-Objective Optimization Method for Homes of EV Owners. In: World Electric Vehicle Journal 10/2019. Tokyo: Toyota Central R&D Labs., The University of Tokyo, 2019.
- 12. Wickert, Manuel; Gerhard, Norman; Trost, Tobias; Prior, Johannes; Cacilo, Andrej; Hartwig, Matthias et al.: Wissenschaftliche Unterstützung bei der Erstellung von fahrzeugbezogenen Analysen zur Netzintegration von Elektrofahrzeugen unter Nutzung erneuerbarer Energien -Endbericht zum Vorhaben FKZ UM 11 96 107. Kassel und Bremerhaven: Fraunhofer-Institut für Windenergie und Energiesystemtechnik, IWES, 2013.
- 13. International Journal of Sustainable Energy Planning and Management Cost and selfconsumption optimised residential PV prosumer systems in Germany covering residential electricity, heat and mobility demand. Vol. 21 2019 35–58. Regensburg, Lappeenranta: OTH Regensburg, LUT University Lappeenranta, 2019.
- 14. Driesse, Anton; Jain, Praveen; Harrison, Steve: Beyond the curves: Modeling the electrical efficiency of photovoltaic inverters. San Diego, CA, USA: Dept. of Mechanical Engineering, Queen's University, Canada, 2009.
- 15. Illing, Bjoern; Warweg, Oliver: Analysis of international approaches to integrate electric vehicles into energy market. Lisbon: 2015 12th International Conference on the European Energy Market (EEM), 2015.
- 16. Shafiullah, Md; Al-Awami, Ali T.: Maximizing the profit of a load aggregator by optimal scheduling of day ahead load with EVs. Seville: 2015 IEEE International Conference on Industrial Technology (ICIT), 2015.
- 17. Illing, Bjoern; Warweg, Oliver: Achievable revenues for electric vehicles according to current and future energy market conditions. Porto: 2016 13th International Conference on the European Energy Market (EEM), 2016.
- Bessa, Ricardo; Matos, Manuel; Soares, Filipe; Peças Lopes, João A: Optimized Bidding of a EV Aggregation Agent in the Electricity Market. New Jersey: IEEE Transactions on Smart Grid, VOL. 3, NO. 1, MARCH 2012, 2012.
- 19. Kern, Timo; Dossow, Patrick; von Roon, Serafin: Integrating Bidirectionally Chargeable Electric Vehicles into the Electricity Markets. Basel, Switzerland: Energies 2020, 13(21), 5812, 2020.
- 20. Müller, M.; Biedenbach, F.; Reinhard, J: Development of an Integrated Simulation Model for Load and Mobility Profiles of Private Households. Energies 2020, 13, 3843.
- 21. Sauer, D.U.; Schmidt, H.: Wechselrichter-Wirkungsgrade: Praxisgerechte Modellierung und Abschätzung. Freiburg: Fraunhofer-Institut für Solar Energiesysteme (ISE), 1994.
- 22. Roth, Hans: Angewandte Simulation und Optimierung in der Energiewirtschaft Operations Research - Detailliertere Modellierung des Kraftwerkbetriebs: Umsetzung nichtlinearer und ganzzahliger Probleme in der linearen Programmierung. 2019.
- 23. Energy Storage: A Key Enabler for the Decarbonisation of the Transport Sector. Brussels: EASE European Association for Storage of Energy, 2019.
- 24. Regelungen zu Stromspeichern im deutschen Strommarkt. Bonn: Bundesnetzagentur für Elektrizität, Gas, Telekommunikation, Post und Eisenbahnen, 2020.

- 25. BMVI: Mobilität in Deutschland MiD Ergebnisbericht. In: https://www.bmvi.de/SharedDocs/DE/Artikel/G/mobilitaet-in-deutschland.html. (accessed on 2020-04-01); Bonn: Bundesministerium für Verkehr und digitale Infrastruktur, 2017.
- 26. Monitoringbericht 2018. Bonn: Bundesnetzagentur für Elektrizität, Gas, Telekommunikation, Post und Eisenbahnen, 2019.
- 27. EEG-Vergütungsübersicht für Inbetriebnahmejahr 2018. Munich: Verband der Bayerischen Energie- und Wasserwirtschaft e.V., 2018.
- 28. Schroedter-Homscheidt, Marion et al.: User's Guide to the CAMS Radiation Service Status December 2016. Shinfield Park: ECMWF, 2016.
- 29. aroTHERM air source heat pump. In: https://www.vaillant.co.uk/specifiers/products/arothermair-source-heat-pump-53824.html. (accessed on 2021-07-08); Derbyshire: Vaillant Group, 2021.
- 30. Bründlinger, Thomas; König, Julian Elizalde; Frank, Oliver; Gründig, Dietmar; Jugel, Christoph; Kraft, Patrizia et al.: dena-Leitstudie Integrierte Energiewende - Impulse für die Gestaltung des Energiesystems bis 2050 - Impulse für die Gestaltung des Energiesystems bis 2050. Berlin: Deutsche Energie-Agentur, 2018.
- 31. Wirth, Harry: Aktuelle Fakten zur Photovoltaik in Deutschland. Freiburg: Fraunhofer ISE, 2020.
- 32. Long range and rapid charging: the battery system is at the heart of the Volkswagen ID.3, ID.4 and ID.4 GTX. In: https://www.volkswagenag.com/en/news/2021/05/long-range-and-rapid-charging.html#. (accessed on 2021-07-08); Wolfsburg: Volkswagen, 2021.
- 33. AMPERA-E. In: https://www.opel.be/content/opel/worldwide/master/en/index/cars/amperae/model-overview.html. (accessed on 2021-07-08); Rüsselsheim am Main: Opel Automobile GmbH, 2021.
- Köppl, Simon; Samweber, Florian; Bruckmeier, Andreas; Böing, Felix; Hinterstocker, Michael; Kleinertz, Britta et al.: Projekt MONA 2030: Grundlage für die Bewertung von Netzoptimierenden Maßnahmen - Teilbericht Basisdaten. München: Forschungsstelle für Energiewirtschaft e.V. (FfE), 2017
- 35. Day-ahead Prices. In: https://transparency.entsoe.eu/transmissiondomain/r2/dayAheadPrices/show. (accessed on 2020-08-31); Brussels: ENTSO-E, 2020.
- 36. Ma, Ke; Yang, Yongheng; Wang, Huai; Blaabjerg, Frede: Design for Reliability of Power Electronics in Renewable Energy Systems. Basel: Springer International Publishing AG, 2014.
- 37. Maximum performance for your independence. In: https://bmz-group.com/images/PDF-Downloads/Broschuere-ESS\_EN\_with\_ESSX.pdf. (accessed on 2020-08-27); Karlstein am Main: BMZ Energy Storage Systems, 2020.
- Battery Box 2.5. In: https://www.climaverd.com/sites/default/files/2019-06/ft\_sb\_byd\_b-box\_2.5-10.0\_b-plus\_2.5\_en.pdf. (accessed on 2020-08-27); Shenzhen, China: BYD Company Limited, 2020.
- 39. Wilde, Joerg: Generalization of the Annuity Factor. In: Accounting and Finance Research Vol. 7, No. 2/2018. Essen, Germany: Pension Fund of Water Associations (Ruhrverband), 2018.
- 40. Wallbox Quasar Bidirektionale Ladestation. In: https://besserladen.de/produkt/wallbox-quasarbidirektionale-ladestation/. (accessed on 2020-08-28); Dittelbrunn: Besserladen, 2020.
- 41. Borlaug, B.; Salisbury, S.; Gerdes, M.; Muratori, M.: Levelized Cost of Charging Electric Vehicles in the United States, Joule (2020), https://doi.org/10.1016/j.joule.2020.05.013.
- 42. Ekins, Paul; Kesicki, Fabian; Smith, Andrew: Marginal Abatement Cost Curves A call for caution. München: University College London Energy Institute, 2011
- 43. Charging Station Webasto Pure 11kW. In: https://webasto-charging.com/en\_de/webasto-purewallbox-11kw.html. (accessed on 2021-09-30); Gilching: Webasto Thermo & Comfort SE, 2021.

Kern, T., Kigle, S. (2022). Modeling and Evaluating Bidirectionally Chargeable Electric Vehicles in the Future European Energy System. Energy Reports. (Accepted, not yet published)

©2022 by the authors. Published by Elsevier Ltd.

### Publication 3 (Pub3): Modeling and Evaluating Bidirectionally Chargeable Electric Vehicles in the Future European Energy System

#### Timo Kern<sup>a,b</sup>, Stephan Kigle<sup>c</sup>

- <sup>a</sup> Forschungsgesellschaft für Energiewirtschaft mbH (FfE), 80995 Munich, Germany
- <sup>b</sup> School of Engineering and Design, Technical University of Munich (TUM), Boltzmannstraße 15, 85748 Garching, Germany
- <sup>c</sup> Forschungsstelle für Energiewirtschaft e.V. (FfE), 80995 Munich, Germany

Abstract: In addition to a massive expansion of renewable energies, a successful change towards a decarbonized energy system requires the flexibilization of consumers and the integration of storage and sector coupling technologies. Bidirectionally chargeable electric vehicles (EVs) represent such a consumer flexibility. They are able to charge when there is an electricity generation surplus and to discharge when there is a shortage in electricity generation. Therefore, they can act as a storage from the perspective of the energy system. This paper analyzes different modeling approaches of bidirectionally chargeable EVs in large-scale energy systems and evaluates the impact of bidirectionally chargeable EVs on the future European energy system design. We compare the modeling of discrete EV profiles, clustered EV profiles as well as an aggregated EV profile with simplified constraints. Aggregation of EV profiles per country leads to significantly lower computation times, while still achieving results close to the reference case. The number of bidirectionally chargeable EVs in a cost optimal future European energy system increases from 6 million EVs in 2025 to over 60 million EVs in 2050. We show that bidirectionally chargeable EVs lead to a better integration of PV generation, to lower installed capacities of gas- and hydrogen-fired power plants as well as stationary battery storages. They also lead to decreasing electricity prices and total European energy system costs.

Peer-review under responsibility of the scientific committee of the 7th International Conference on Advances on Clean Energy Research.

**Keywords:** bidirectionally chargeable EVs; vehicle-to-grid, V2G; energy system modeling; EV charging strategies; European energy system; flexibility options; demand-side flexibility

# 1. Introduction

In order to achieve the European climate protection targets, the future European energy system will be strongly characterized by volatile renewable energies [1]. Numerous studies analyze a future decarbonized energy system at the national level [2]. Due to the increase in renewable energies, additional flexibility is necessary to balance electricity generation and consumption in the energy system. Electric vehicles (EVs) represent a possible large demand-side flexibility. Smart charging allows EVs to charge when there is a surplus of electricity generation. Bidirectional charging allows to charge smartly, and even feed electricity back into the system when there is a shortage of electricity generation [3]. There are multiple use cases for bidirectionally chargeable EVs [4]. In vehicle-to-home (V2H) applications, energy from the EV is fed back into the household (on the energy user's side of the meter). Vehicle-to-grid (V2G) applications feed-back energy into the grid. The added value of bidirectionally chargeable EVs for the future energy system has not yet been sufficiently investigated. Therefore, in this paper we investigate modeling approaches of bidirectionally chargeable EVs in energy system models and resulting system-optimal penetration rates of smart and bidirectionally chargeable EVs in the future European energy system.

Numerous papers discuss the modeling of bidirectionally chargeable EVs from the user's perspective [5]. In V2H applications they must be modeled with varying charging and discharging efficiencies, since low power flows in a household lead to low efficiencies that cannot be modeled by one fixed efficiency [6]. Kern et al. show that revenue potentials vary strongly depending on user behavior, electricity prices and feed-in tariffs as well as other household components and usually range from 200 to 500 €/EV/a. Salpakari et al. model V2H applications in a household with a photovoltaic (PV) system in Sweden, quantify the annual electricity cost savings to 8% to 33% and show the impact on battery degradation [7]. Charging and discharging powers in V2G applications, like arbitrage trading, are usually at maximum charging and discharging power. Therefore, modeling a fixed charging and discharging efficiency is sufficient here [8]. Bidirectionally chargeable EVs can significantly reduce charging costs and even generate revenues to the owner of the EV [9]. Revenue potentials vary strongly depending on the EV and electric vehicle supply equipment (EVSE) characteristics, the regulatory framework, user behavior, and electricity market prices and usually range from 100 to 1,000 €/EV/a [8]. Since all aforementioned studies focus on the modeling of individual bidirectionally chargeable EVs, the impacts of bidirectionally chargeable EVs on the energy system are not evaluated. For example, neglected repercussions of EVs participating in the spot markets on electricity market prices potentially lead to overestimated revenues. Therefore, we model smart and bidirectionally chargeable EVs with a cost-optimal penetration rate in the energy system and estimate their revenues in future electricity markets.

There are some studies that discuss the impact of bidirectionally chargeable EVs on the electricity markets. Hanemann et al. discuss the flattening effect of EVs on electricity demand and electricity prices depending on the smart or bidirectional charging strategy of

the EVs [10]. However, they model bidirectionally chargeable EVs highly simplified regarding the EV profiles and EV constraints and limit the observation area to Germany. Rodríguez et al. also show the smoothing effect of bidirectionally chargeable EVs on the electricity demand of Bogotá D.C. in Colombia, South America, resulting in a higher load factor [11]. However, once again the observation area is small and the modeling of the EVs is highly simplified. Huang et al. model the large-scale penetration of electric vehicles by a simplified charging behavior that is only according to statistic customers' travel needs [12]. They show that bidirectionally chargeable EVs can ensure the safe operation of distribution systems, but do not discuss repercussions on the energy system. Wei et al. optimize energy system planning with V2G applications and show the positive economic and environmental effect of V2G on the energy system [13]. However, the modeled EV fleet is simplified by stochastic features and the observation area is limited to cities, which leads to different requirements for the modeling approach compared to large energy systems. Child et al. show the decreasing impact of bidirectionally chargeable EVs on the need for storage and generation capacity for the Åland Islands near Finland [14]. In this paper, we also evaluate the impact of bidirectionally chargeable EVs on energy system components, such as electric storages and generation capacities. However, we will focus on the modeling approaches of bidirectionally chargeable EVs in more detail. Additionally, we evaluate their impact on total energy system costs and electricity prices in the European energy system.

None of the aforementioned studies model cost-optimal penetration rates of bidirectionally chargeable EVs in a highly coupled, large-scale multi-energy carrier system, since the optimization problem quickly becomes very complex. In this paper, we address this research gap. We formulate and analyze different modeling approaches of bidirectionally chargeable EVs and publish resulting EV profiles. In our results section, we use the best fitting modeling approach to evaluate cost-optimal penetration rates of bidirectionally and smart chargeable EVs in the future European energy system from 2025 to 2050. Therefore, this paper provides policy recommendations for action, i.e., where and when to promote the new technology of bidirectionally chargeable EVs in Europe. Furthermore, stakeholders in the field of energy can align business models with the cost-optimal expansion of this new technology. Ultimately however, scientists can also use the modeling approaches and the EV profiles for their own energy system analyses.

# 2. Methodology

### 2.1 Modeling approaches of V2G applications in energy system models

Bidirectionally chargeable EVs are modeled like stationary storage systems with additional constraints on availability and driving consumption. Therefore, the absolute battery state of charge (SoC) SoC(t) in a timestep t is determined by the absolute battery SoC of the previous timestep SoC(t-1) added to the charged energy  $P_c(t) \cdot \eta_c \cdot \Delta t$  and subtracted by the discharged energy  $\frac{P_d(t)}{\eta_d} \cdot \Delta t$  and the driving consumption  $P_{con}(t) \cdot \Delta t$ , where  $\Delta t$  is the

time step length.  $P_c$  and  $P_d$  refer to charging and discharging powers, while  $\eta_c$  and  $\eta_d$  describe charging and discharging efficiencies.

$$SoC(t) = SoC(t-1) + P_c(t) \cdot \eta_c \cdot \Delta t - \frac{P_d(t)}{\eta_d} \cdot \Delta t - P_{con}(t) \cdot \Delta t$$
(1)

To model realistic user behavior, we introduce Equations 2 and 3 that further constrain the SoC. The minimum percentage safety SoC  $min_{safe}$  in Equation 2 represents the requirement to perform a safety-related drive, like a drive to the hospital, at any time when the EV is connected. The minimum percentage SoC at departure  $min_{dep}$  respects the user's desire of a minimum SoC of the EV battery when departing. The relative parameters  $min_{safe}$  and  $min_{dep}$  are multiplied by the maximum absolute SoC  $SoC_{max}$  to compare it to the absolute state of charge SoC(t). If the EV is not connected to an EVSE,  $min_{safe}$  and  $min_{dep}$  are set to zero, since there is no minimum SoC desired by the user.

$$SoC(t) \ge SoC_{max} \cdot min_{safe}$$
 for all timesteps t with connected EV (2)

$$SoC(t) \ge SoC_{max} \cdot min_{dep}$$
 for all timesteps t with departing EV (3)

Furthermore, usual constraints restrict the charging and discharging power by maximum values  $P_{c,max}$  and  $P_{d,max}$  as well as the battery SoC by a parameterized maximum  $SoC_{max}$  in Equations (4-6) for all considered timesteps t. If the EV is not connected,  $P_{c,max}$  and  $P_{d,max}$  are set to zero, since charging and discharging is not possible.

$$P_c(t) \le P_{c,max}$$
 for all timesteps t with connected EV (4)

$$P_d(t) \le P_{d,max}$$
 for all timesteps t with connected EV (5)

$$SoC(t) \le SoC_{max}$$
 for all timesteps t (6)

Since the modeling of the European energy system is a highly complex optimization problem, the integration of every single EV as a discrete element in the energy system model would lead to unacceptable computing times. For this reason, we further discuss three modeling approaches of bidirectionally chargeable EVs in energy system models. The described modeling of bidirectionally chargeable EVs can be equally applied for smart chargeable EVs with the restriction that the discharging power of the EVs is set to zero.

#### Modeling of bidirectionally chargeable EVs with discrete EV profiles

In the first approach, we model EVs by discrete driving and location profiles. The EV consumption profile sets the variable  $P_{con}(t) \cdot \Delta t$  in Equation 1. The EV location profile is transferred to an EV connection profile. Depending on the modeled location of the EV supply equipment (at home, at work or in a public space), the EV status is connected or not connected. The connection status of the EV sets the variables in Equations 2-5.

Since the number of EVs in the future European energy system will increase to multiple millions, there is still a need to lower the number of discrete profiles. In an evaluation of synthetic EV profiles based on a mobility study in Germany in 2017 [15], Fattler shows that 10,000 discrete EV profiles are statistically viable to represent the average German mobility behavior [16]. Randomly drawing 1,000 EV profiles from this pool of 10,000 EV profiles leads to a deviation of indicators, such as equivalent full cycles or daily charging

hours of below 3%. Therefore, in this approach, we integrate the bidirectionally chargeable EVs in the energy system model by 1,000 synthetic EV consumption and location profiles.

Since 1,000 modeled EV elements differ from the number of bidirectionally chargeable EVs integrated into the energy system in reality, a scaling factor is introduced to set the constraining variables in Equations 1-6. In Equation 7, we calculate the scaling factor  $f_{num,EVs}$  by the number of bidirectionally chargeable EVs  $n_{EVs}$  divided by 1,000, which represents the 1,000 equally weighted EV elements.

$$f_{num,EVs} = \frac{n_{EVs}}{1,000} \tag{7}$$

The scaling factor is used to set the constraining variables according to Equations 8-11. The parameters of a single EV  $E_{con,1EV}(t)$ ,  $P_{c,max,1EV}$ ,  $P_{d,max,1EV}$  and  $SoC_{max,1EV}$  are multiplied by the scaling factor. As a result, one EV element represents  $f_{num,EVs}$  EVs with the same features. One million integrated bidirectionally chargeable EVs, for example, would consequently result in one discrete EV element representing 1,000 EVs.

$$E_{con}(t) = E_{con,1EV}(t) \cdot f_{num,EVs}$$
(8)

$$P_{c,max} = P_{c,max,1EV} \cdot f_{num,EVs} \tag{9}$$

$$P_{d,max} = P_{d,max,1EV} \cdot f_{num,EVs} \tag{10}$$

$$SoC_{max} = SoC_{max,1EV} \cdot f_{num,EVs}$$
(11)

#### Modeling of bidirectionally chargeable EVs with clustered, discrete EV profiles

Since 1,000 discrete bidirectionally chargeable EV elements integrated in an energy system model might still lead to high computing times, we further reduce the number of EV elements by using clustering algorithms. We use a k-means algorithm for the clustering of EV profiles based on an approach shown by Schmidt-Achert et al. [17] to best represent the mobility behavior of EVs.

We define representative driving and location profiles for each cluster by a three-step approach. First, features for the clustering algorithm are defined. Since the location, and thus the availability, of the EVs is most important for the value of smart or bidirectional charging [8], the availability at home and at work for 6-hour time frames per weekday results in 56 features for the clustering, e.g., Monday from 0am to 6am is one feature. Second, a k-means clustering algorithm is applied to find a predefined number of clusters to represent the population best. Third, the best representative of a cluster is chosen. Like Schmidt-Achert et al., we also define representatives by minimizing the sum of maximum and mean error compared to the reference, instead of using the traditional selection of the closest representative to the centroid. This approach leads to an economically viable choice of representatives, e.g., avoiding unrealistic low availabilities.



Fig. 1.Location of EVs in each feature for reference of 1,000 EV profiles compared to 5<br/>discrete clustered EV profiles.

In this paper, we develop clustered EV profiles by using the described approach for 5, 10, 20 and 50 clusters. Figure 1 shows a comparison of defined features for the reference case of 1000 EV profiles (a) and a representation of EV profiles by 5 clusters (b). The fundamental characteristic of the reference can be represented well by the 5-cluster approach. The lowest availability of EVs at home is just below 60%, the availability of EVs at work from Monday to Friday from 6am to 6pm is around 20% and the availability of EVs elsewhere (meaning at public spaces or driving) mostly varies from 10% to 20%. However, one can recognize concise differences. The availability of EVs at home during the weekend is much higher for the clustered profiles, leading to lower availabilities at work and elsewhere in this time frame. The same EV profile differences can be recognized during the weeknights.

In Table 1, we further analyze the different statistical behavior of clustered EV profiles (*CluX* for X discrete EV profiles) in comparison to the reference case of 1,000 EV profiles (*Ref*). The overall mean availability of EVs at home, at work and elsewhere is represented quite well by the clustered profiles. However, one can observe that the deviation of all characteristics from the reference increases with a decreasing number of clusters. The availability of EVs at home is increasingly overestimated with a decreasing number of clusters and the availability of EVs at work and elsewhere is increasingly underestimated with a decreasing number of clusters. Taking a closer look at the maximum and mean absolute deviation of the availability of EVs in each defined feature, the deviation of the characteristics from the reference increases for all clustered EV profiles. The extent of the influence of these deviating characteristics on other elements in the energy system is analyzed in Chapter 3.1 to further evaluate the clustered EV profiles.

	Mean availability					Max deviation from Ref				Mean deviation from Ref			
	Ref	Clu50	Clu20	Clu10	Clu5	Clu50	Clu20	Clu10	Clu5	Clu50	Clu20	Clu10	Clu5
Home	77.6%	79.5%	80.9%	81.0%	81.4%	6.5%	9.6%	10.7%	12.6%	2.1%	3.4%	3.6%	4.4%
Work	8.6%	8.5%	7.6%	8.1%	8.3%	3.4%	7.0%	7.0%	7.5%	0.9%	1.3%	1.8%	2.2%
Elsewhere	13.8%	12.0%	11.5%	10.9%	10.2%	6.3%	6.1%	7.7%	14.6%	1.9%	2.4%	3.3%	4.4%

Table 1.Availability at home, at work or elsewhere for clusters Clu50, Clu20, Clu10 and<br/>Clu5 in comparison to reference scenario Ref.

The clustered EV profiles include discrete EV location and consumption profiles and a share for each EV representative of the total EV number. Consequently, the number of EVs for each discrete EV profile is calculated by multiplying the total number of EVs  $n_{EVs}$  with the share of each EV profile:

$$f_{num,EVs} = n_{EVs} \cdot share_{EV} \tag{12}$$

Taking into account the scaling factor  $f_{num,EVs}$ , the constraining variables and parameters are set by Equations 8-11. The clustered, discrete EV profiles are published in [18].

#### Modeling of bidirectionally chargeable EVs with aggregated EV profile

To further reduce the complexity of the optimization problem, we aggregate all initial 1,000 EV profiles into one EV location and consumption profile that represents the total EV fleet. The aggregated profile contains shares of EV locations and average consumptions of EVs. This leads to a simplified modeling of bidirectional chargeable EVs, since discrete correlations cannot be mapped.

Figure 2 illustrates the different modeling of aggregated and discrete EV profiles by showing the minimum and maximum availability of a single EV at home compared to the aggregated EV profiles for an exemplary week. The maximum availability of the discrete EV profile is either one or zero, while the maximum availability of the aggregated EV profile varies between 50% and 95%. The minimum availability is determined by the minimum state of charge at departure (example set to 70%) and the minimum safety state of charge (example set to 30%). For the discrete EV profile, it is either 0%, 30% or 70%, while the aggregated minimum availability varies between 15% and 35%. Modeling discrete EV profiles leads to time-dependent correlations that are neglected by the modeling of aggregated EV profiles. A discrete EV that departs with an SoC over 70% comes back with an SoC depending on the SoC at departure. Using the aggregated EV profiles neglects this correlation, but still considers the overall SoC restrictions for the EV pool by the aggregated minimum availability.

Since the aggregated profile is normed to one EV, the scaling factor  $f_{num,EVs}$  is equal to the number of EVs  $n_{EVs}$  as shown in Equation 13. Again, the constraining variables and parameters are set by Equations 8-11. The aggregated EV profiles are published in [18].



$$f_{num,EVS} = n_{EVS} \tag{13}$$

Fig. 2. Different availability of aggregated and discrete EV profiles for an exemplary week.

### 2.2 Integration of V2G applications in the energy system model ISAaR

We use the linear optimization model ISAaR (Integrated simulation model for unit dispatch and expansion with regionalization) that provides a cost-optimal expansion and dispatch of units to model the future European energy system. Figure 3 illustrates the elements and energy carriers that are modeled in ISAaR. There are multiple elements that couple different energy carriers. Examples are gas-fired power plants or power-to-x technologies that produce other energy carriers from electricity. In addition, storage elements for electricity, heat, hydrogen, and methane provide additional flexibility to the energy system. Energy carriers from outside the geographical scope of ISAaR can be integrated from the model via import. Final energy consumption models provide the energy demand as input data for the optimization. A detailed description of the final energy consumption models Smind EU for the industry sector, TerM EU for the tertiary sector, TRAM EU for the transport sector and PriHM EU for the building sector can be found in [19].

ISAaR minimized the total energy system costs, while balancing consumption and generation per energy carrier for every timestep and every region. Equation 14 shows these system constraints. The demand of the final energy consumption sectors for every energy carrier c, every timestep t and every region  $r P_{demand}$  is equal to the generation  $P_{gen}$  of all elements added to the imports of the energy carrier  $P_{import}$  subtracted by the consumption of all elements  $P_{cons}$  and the exports of the energy carrier  $P_{export}$ . A storage element, such as a bidirectionally chargeable EV, that discharges results in a generation  $P_{gen}$  from the energy system's perspective. A storage element that charges represents a consumption from the energy system's perspective. A more detailed description of the ISAaR energy system model can be found in [20].

$$P_{demand}(t,r,c) = \sum_{elements} P_{gen}(t,r,c) - \sum_{elements} P_{cons}(t,r,c) + P_{import}(t,r,c) - P_{export}(t,r,c)$$
for every timestep t, region r and energy carrier c
$$(14)$$

The final energy demand of the transport sector modeled in TRAM includes the electricity demand of passenger cars for every European country. Since the demand is transmitted to ISAaR statically, TRAM models all passenger cars as unmanaged charging electric vehicles. The integration of bidirectionally chargeable EVs ('BCM-EVs') leads to a decrease in unmanaged charging EVs. Since TRAM provides a static input of electricity demand, the decreasing number of unmanaged charging EVs is modeled by a negative demand `-Unman EVs` reflecting the electrical load of the integrated bidirectionally chargeable EVs. This approach allows a dynamic, model endogenous increase of bidirectionally chargeable EVs in ISAaR. If ISAaR decides to integrate a bidirectionally chargeable EV, this will result in a decrease of the static demand of the transport sector representing a removal of an unmanaged charging EV.

There are two options to integrate bidirectionally chargeable EVs into the energy system model. First, the number of bidirectionally chargeable EVs  $n_{EVs}$  can be fixed for future years, meaning a stock element from the model's perspective. Second, bidirectionally chargeable EVs can be modeled as an expansion element with expansion costs. In this case, their

number  $n_{EVs}$  is endogenously determined by ISAaR. The expansion costs reflect the differential costs of a bidirectionally chargeable EV including EVSE versus an unmanaged charging EV including EVSE.

The described approach for the integration of bidirectionally chargeable EVs in the energy system model can be applied equally for smart chargeable EVs.



**Fig. 3.** Illustration of technologies in the multi-energy system model ISAaR and added elements for modeling bidirectionally chargeable EVs.

## 3. Results

In the results section we discuss the advantages and disadvantages of the presented modeling approaches of bidirectionally chargeable EVs for an exemplary integration of the EVs in Germany. This is followed by a scenario evaluation of the European energy system with smart and bidirectionally chargeable EVs for the timeframe 2025 to 2050.

### 3.1 Evaluation of modeling approaches of bidirectionally chargeable EVs

For the evaluation of the different presented modeling approaches, we model the European energy system for the year 2030 with the ISAaR energy system model and integrate bidirectionally chargeable EVs into the German energy system. The parameterization of all modeled elements in the energy system, e.g., generation, consumption, or storage units, is based on the solidEU scenario of the eXtremOS research project. A detailed scenario description of solidEU is published in a summary report [1]. Table 2 shows the parametrization of the bidirectionally chargeable EVs. The EVs are parameterized by a medium passenger car battery capacity of 50 kWh [21].

	Parametrization of bidirectionally chargeable Evs.									
Battery capacity SoC <sub>max,1EV</sub>	Charging/ discharging power P <sub>c,max,1EV</sub> and P <sub>d,max,1EV</sub>	Charging/ discharging efficiency $\eta_c$ and $\eta_d$	Minimum safety SoC <i>min<sub>safe</sub></i>	Minimum SoC at departure <i>min</i> <sub>dep</sub>	Location of bidirectional EVSE	Number of bidirectionally chargeable EVs n <sub>EVs</sub>				
50 kWh	11 kW	94%	30%	70%	At home	13 million EVs				

Table 2.	Parametrization	of bidirectionally	chargeable FVs
	i urunictrizution	or branectionally	chargeable LV3.

The maximum charging and discharging power are set to 11 kW, based on the technical design of the EVSE in the BCM project [4]. The charging and discharging efficiencies of 94% (losses in EV and EVSE included) are the future efficiencies expected by experts from the BCM project [4]. The safety minimum SoC and the minimum SoC at departure are set to 30% and 70%, respectively, and the location of the bidirectional EVSE is at home. The fixed EV number of 13 million EVs is oriented to the underlying solidEU scenario [1].

We evaluate the three different modeling approaches presented in Chapter 2 in regard to differences in simulation results and computing time. The first presented modeling approach of a discrete EV modeling is the most detailed and thus serves as a reference for the other modeling approaches. Table 3 summarizes the impact of the different EV modeling approaches with respect to computation time, EV behavior and repercussions on the energy system.

The discrete EV modeling leads to a computation time of 410 million ticks for the one-year energy system optimization, which corresponds to a computation time of 3 days on the computing servers<sup>1</sup> used. The discrete clustered EV modeling results in much lower computation times of 155 million ticks for 5 EV profiles to 175 million ticks for 50 EV profiles. The aggregated EV profile modeling also has a computation time in this range with 172 million ticks. Although in this approach the EVs are modeled with only one profile, the computation time is slightly higher than for the approaches with 5, 10 and 20 discrete clustered EV profiles. This may be due to the more complex aggregated EV profile, but also to the feedback effects of the EV profile on the energy system.

As a second evaluation of the EV modeling approaches, Table 3 shows the EV behavior of the 13 million EVs modeled in the German energy system. The minimum, maximum and mean availability refers to the availability of the EVs at home over all simulated 8760 hours of the year. For the EV fleet in Germany, represented by the 1,000 discrete EV profiles, the minimum availability is 48%, the maximum availability is 95% and the mean availability is 78%. These characteristics are matched perfectly by the aggregated EV profile. The discrete clustered EV profiles lead to a large error for the minimum availability, especially for 5 to 20 EV profiles. This can lead to unrealistic scarcity situations, where hardly any EVs can interact with the energy system. The charged and discharged energy varies between the modeling approaches, but only by a maximum of 5%. The aggregated EV profile matches the reference very well. The revenues per EV are calculated by the EV energy sold minus the EV energy purchased, times the electricity prices and also vary by a maximum

<sup>&</sup>lt;sup>1</sup>Hardware: 2xAMD EPYC 7F52 – 16 Core, 1008 GB RAM Software: Matlab [22], Gurobi Optimization [23]

of 5%. Again, the aggregated EV modeling approach matches the reference and thus reflects the EV behavior very well.

Table 3.	Impact of different EV modeling approaches on computation time, EV behavior
	and repercussions in the energy system.

Modeling approach	Specifi- cation	Compu- tation		EV behavior					Repercussions on the energy system				European modeling
		time (ticks)	min avail- ability	max avail- ability	mean avail- ability	charge in TWh	dis- charge in TWh	revenues per EV	Overall costs in bn€/a	Wind onshore in TWh in DE	Wind offshore in TWh in DE	PV in TWh in DE	expansion element
Discrete EV profiles	1000 EV profiles	410 million	0.49	0.95	0.78	124.4	91.9	158.25	410.40	261.6	185.8	219.3	no
	50 EV profiles	175 million	0.38	1.00	0.80	124.2	95.4	165.51	410.25	261.9	188.5	219.5	no
Discrete	20 EV profiles	162 million	0.19	1.00	0.81	124.4	96.2	161.42	410.39	261.8	180.2	227.6	no
clustered EV profiles	10 EV profiles	157 million	0.11	1.00	0.81	123.1	96.6	163.53	410.37	262.0	178.9	229.0	no
	5 EV profiles	155 million	0	1.00	0.82	120.2	96.0	164.08	410.42	261.8	195.4	227.2	no
Aggregated EV profile	1 EV profile	172 million	0.49	0.95	0.78	125.9	91.7	158.33	410.38	262.2	191.3	231.8	yes

Finally, Table 3 shows the repercussions of the modeling approaches on the energy system. The overall costs represent all costs in the European energy system, with a large part of the costs being in the electricity sector. The overall costs in the different modeling approaches vary only slightly, with the aggregated EV profile matching the reference case very well. Furthermore, we evaluate the expansion of volatile renewable energies in Germany (DE). The electricity generated by wind turbines or PV plants varies by maximum 6%. In general, the expansion and generation of wind onshore, wind offshore and PV is well represented by the discrete clustered and aggregated modeling approach.

For a final evaluation of the modeling approaches, one must consider that the modeling will be applied to all European countries in the ISAaR energy system model. Furthermore, we will also evaluate an optimized expansion of bidirectionally chargeable EVs, which brings even more complexity. The modeling approach with 1,000 discrete EV profiles already has strong computational time disadvantages when modeling German bidirectionally chargeable EVs and is therefore excluded. The aggregated profile shows slight performance advantages compared to the discrete clustered EV profiles, especially for the EV behavior. A European modeling includes 30 European countries and results in 30 times more EV profiles than in our test simulations. In an optimized integration of bidirectionally chargeable EVs for all modeled European countries, the modeling of an aggregated EV profile resulted in a computation time similar to the test scenario, while the modeling of 5 discrete clustered EV profiles per country already led to 4 times higher computation times. For these reasons, we recommend modeling bidirectionally chargeable EVs with an

aggregated EV profile per country to significantly reduce computation time, while still providing a good realistic representation of bidirectionally chargeable EVs.

### 3.2 Future European Energy System with and without V2G applications

To evaluate the integration of smart and bidirectionally chargeable EVs in Europe, we set up a European reference scenario for the years 2025 to 2050 based on the solidEU scenario of the research project eXtremOS. A detailed scenario description of solidEU is published in a summary report [1]. This reference scenario '*Ref*' is characterized by a harmonized decarbonization of the European energy system until 2050. It does not integrate smart and bidirectionally chargeable EVs. For evaluating the value of smart and bidirectionally chargeable EVs, we set up a second scenario '*BCM*' that enables the integration of these EVs. The number of smart and bidirectionally chargeable EVs can be expanded endogenously in every European country using the energy system model. Since there are no European country-specific data for EV profiles publicly available, we use the aggregated German EV profile for every European country. This approach neglects country-specific driving behavior, but considers the weekly and seasonal driving characteristics, which in other European countries do not differ fundamentally from the German characteristics, since working days and working hours are similar in all European countries.

The smart and bidirectionally chargeable EVs are parameterized by the characteristics of a mid-size passenger car described in Table 2. The number of the EVs is endogenously optimized and not fixed by parameters. For this optimized integration, Table 4 shows the additional investment costs of smart and bidirectionally chargeable EVs compared to unmanaged charging EVs. These additional investment costs include additional investments and installation costs for the EVSE and for additional required measuring equipment. The costs do not include additional investment costs for the EV itself. The assumptions for the additional investment costs were made within the BCM research project in consultation with various experts [24]. Currently, only a few offers for bidirectional EVs and EVSEs are available that are consequently high-priced. The only available offer for a bidirectional EVSE in Germany is around 6,000 € [25] compared to around 500 € for an unmanaged EVSE [26] resulting in difference costs of 5,500 € only for the purchase of the EVSE. Table 4 shows that experts in the BCM research project expect a high cost degression for bidirectional EVSEs in the timeframe up to 2040, when unit numbers in production go up. The lifetime of the smart and bidirectionally chargeable EVs, including their EVSE, is set to 15 years [27] and the interest rate from the energy system's perspective is 3.5% [28]. Both values are needed for the calculation of annual investment costs in the energy system model.

	Year	2025	2030	2035	2040	2045	2050
Additional investment	Smart chargeable EVs	960	760	760	760	760	760
costs for EV in €/EV	Bidirectionally chargeable EVs	2840	2190	1890	1590	1590	1590

Table 4.Additional investment costs of smart and bidirectionally chargeable EVs<br/>compared to unmanaged charging EVs.

Figure 4 analyzes the EV numbers from 2025 to 2050 in a cost-optimized future European energy system in the *BCM* scenario. Supplementary Appendix A presents the exact numbers of EVs per charging strategy per country from 2025 to 2050. In Figure 4 (a), one can see the number of smart and bidirectionally chargeable EVs compared to unmanaged charging EVs. The first thing that stands out is that almost no smart chargeable EVs are added endogenously by the energy system model. This means that the added value of bidirectionally chargeable EVs to the energy system significantly exceeds the additional costs of these EVs, when costs are considered as shown in Table 4. The number of integrated bidirectionally chargeable EVs goes up from 7 million EVs in 2025 to 62 million EVs in 2050 and has the highest share in 2045 of 37%. This significant increase indicates that the future European energy system becomes more efficient with the provided flexibility through bidirectionally chargeable EVs.

Figure 4 (b) shows the integration of bidirectionally chargeable EVs in the six countries Germany (DE), Spain (ES), France (FR), Italy (IT), Poland (PL) and United Kingdom (UK) representing the European countries with the highest number of integrated EVs. The increase in bidirectionally chargeable EVs is not occurring equally in different countries, but rather has characteristic differences. In Spain and Italy, the energy system model already integrates around 2 million bidirectionally chargeable EVs in 2025, significantly more than in the other countries displayed. As of the year 2035, the countries Germany, France and Italy have the most bidirectionally chargeable EVs. In 2050 in Germany there is a sharp decrease in bidirectionally chargeable EVs, meaning that some of the previous bidirectionally chargeable EVs that reached their lifetime are replaced by unmanaged charging EVs. In 2050 in Germany, expanded hydrogen-fired power plants provide flexibility leading to a lower flexibility needed on the demand-side.

If unmanaged charging EVs are replaced by bidirectionally chargeable EVs, the EV loads due to additional equivalent full cycles (EFCs) of the battery will be increased. The EFCs of bidirectionally chargeable EVs are highest in 2025 at 280 EFCs/a, going down to 200 EFCs/a in 2035 and 185 EFCs/a in 2050 compared to approximately 60 EFCs/a for an unmanaged charging EV with 50 kWh battery capacity.





Analysis of future EV numbers from 2025 to 2050 in a cost-optimized European energy system: (a) total EV number per charging strategy; (b) integrated bidirectionally chargeable EVs per country.

For a more detailed European evaluation of the integration of bidirectionally chargeable EVs, Figure 5 analyzes the shares and numbers of EVs per charging strategy per country in 2030 and 2050. As additional information, the coloring of the countries is related to the full load hours of PV generation. In 2050, there are both higher numbers of EVs and higher shares of bidirectionally chargeable EVs in most countries compared to 2030. The integrated shares of bidirectionally chargeable EVs are much higher in most southern European countries compared to the northern European countries. It is noticeable that the share of bidirectionally chargeable EVs is higher in the countries that have higher full load hours of PV generation. This dependance can be explained by the fact that bidirectionally chargeable EVs often act as daytime storages. The parameterized EVs have an energy/power ratio of 50kWh/11kW. This means they can charge or discharge for a maximum of four and a half consecutive hours. The PV generation represents a regular daily generation profile with only seasonal differences. Therefore, bidirectionally chargeable EVs can provide the necessary flexibility to more evenly feed-in the PV generation into the energy system. Higher full load hours of PV generation indicate a more cost-effective expansion of PV generation and thus indicate a more attractive location for bidirectionally chargeable EVs.



**Fig. 5.** Dependance of shares and numbers of EVs per charging strategy per country in 2030 and 2050 on the full load hours of PV generation.

As a detailed scenario comparison, Figure 6 shows the capacities of conventional power plants, volatile renewable energies, and storage technologies for the scenarios *Ref* and *BCM* for the years 2030, 2040 and 2050 in Europe. The analysis points out structural differences, but also similarities, in the future European energy system. There is a high increase in all volatile renewable technologies in future years due to the ongoing decarbonization of the energy system. However, there are some differences between the installed capacities of renewable energies in the scenarios *Ref* and *BCM*. In general, the *BCM* scenario prefers an expansion of PV generation compared to an expansion of wind onshore and wind offshore generation. Bidirectionally chargeable EVs act as a daytime storage increasing the consumption at the time of peak PV generation and thus raise the market value of PV generation that is the volatile renewable energy source with the lowest levelized cost of electricity in ISAaR.

The installed capacities of conventional power plants increase from 2030 to 2050, although the full load hours and thus the electricity output decreases by over 50%. PV plants and wind turbines are characterized by a fluctuating electricity generation that leads to a higher demand of conventional power plants as back-up capacities. The gas-fired power plants in 2050 generate electricity with synthetic methane produced by renewable energies. When comparing the two scenarios *Ref* and *BCM*, it is noticeable that less gas-fired and hydrogen-fired power plant capacity is expanded in the BCM scenario. Bidirectionally chargeable EVs discharge and thus feed-in electricity into the energy system when there is a high need for electricity generation due to a high consumption or a low generation or both. Consequently, gas- or hydrogen-fired power plants can lower their electricity generation in these situations. Therefore, from the energy system's perspective, bidirectionally chargeable EVs lead to a decreased need for flexible gas- and hydrogen-fired power plants.

The installed capacity of storage technologies increases significantly from 2030 to 2050 in both considered scenarios. In the reference scenario *Ref*, the only storage expansion option for the energy system model is the increase of stationary battery storages. The BCM scenario has the expansion options of stationary battery storages and bidirectionally chargeable EVs. Pumped storage hydropower is set fixed for future years and cannot be endogenously expanded. The BCM scenario clearly favors the integration of bidirectionally chargeable EVs over the expansion of stationary battery storages. The lower investment costs of bidirectionally chargeable EVs compared to stationary battery storages outweighs the disadvantage of limited availability of EVs. In terms of one MWh of storage capacity, bidirectionally chargeable EVs have investment costs of 31,800 € in 2050 compared to stationary battery storages that cost around 123,000 € in 2050 based on eXtremOS project [29]. These cost advantages outweigh the limited availability of the EV's storage capacity due to the connection status of the EV to the EVSE and minimum SoC constraints because of the user behavior. These results show that bidirectionally chargeable EVs significantly lower the need for other storage technologies in a cost-optimal future European energy system.



# Fig. 6.Installed capacities of conventional power plants, volatile renewable energies,<br/>and storage technologies for the scenarios *Ref* and *BCM* for the years 2030, 2040<br/>and 2050 in Europe.

For the analysis of electricity prices, Figure 7 shows the annual hourly duration curve of electricity prices in Italy (IT) and the United Kingdom (UK) as exemplary countries with a high and a low share of bidirectionally chargeable EVs for the years 2030 and 2050 for both scenarios *Ref* and *BCM*. In general, the *Ref* scenario results in a slightly higher annual electricity price duration curve. Bidirectionally chargeable EVs in the *BCM* scenario charge when electricity prices are low, resulting in a partly electricity price increase of the low prices. On the other hand, they discharge when electricity prices are high, leading to a partly decrease of the highest electricity prices. These effects result in an overall smoothing of the electricity prices. Italy has a much higher share and number of bidirectionally chargeable EVs than the United Kingdom. Consequently, the impact of the bidirectionally chargeable EVs on the electricity price is also higher in Italy compared to the UK. But while the share of



bidirectionally chargeable EVs in the UK, at 6% in 2030 and 12% in 2050, is comparably low, the impact on electricity prices is nevertheless strongly recognizable.



The difference in the design of the energy system between the scenarios *Ref* and *BCM* leads to a difference in overall energy system costs and mean electricity prices. Table 5 summarizes these key indicators for both scenarios for the years 2025 to 2050. The mean European electricity price is calculated by the demand-weighted electricity price over all modeled European countries and timesteps. Bidirectionally chargeable EVs cause a decrease in electricity prices by up to 12% in 2040. Renewable energies are better and more cost-effectively integrated into the energy system costs are also reduced. While total energy system costs are only slightly lower in *BCM* than in *Ref* in 2025, costs decrease by over 9 billion  $\epsilon/a$  in 2045. Reduced costs on the power supply side clearly outweigh the additional investment costs of bidirectionally chargeable EVs.

	Year	Overall energy billio	system costs in n €/a	Mean European electricity price €/MWh			
_		Ref	ВСМ	Ref	ВСМ		
	2025	432.4	431.7	44.6	42.6		
	2030	414.6	412.4	42.5	39.4		
	2035	354.9	350.1	43.1	39.1		
	2040	332.2	325.7	44.6	39.4		
	2045	323.5	314.4	44.4	41.1		
	2050	353.1	345.0	41.6	38.8		

Table 5.	Overall energy system costs and mean European electricity price for the years
	2025 to 2050 for scenarios <i>Ref</i> and <i>BCM</i> .

### 4. Discussion

Bidirectionally chargeable EVs represent an essential component of the future energy system. In this study, we analyze different modeling approaches of bidirectionally chargeable EVs in a complex multi-energy system model and evaluate their integration as well as their impact on the European energy system in future years. For complex energy system models, we recommend the modeling of one aggregated EV profile to limit computation time, while modeling a realistic representation of the EV composition of a country. We show numerous positive effects of bidirectionally chargeable EVs:

- Bidirectionally chargeable EVs support the integration of PV generation. This is of great importance for the future energy system, since wind generation, especially onshore wind generation, deals with various acceptance issues across Europe [30].
- Bidirectionally chargeable EVs lower the required installed capacities of conventional power plants, such as gas- or hydrogen-fired power plants. Replacing the need for power plant capacities that operate with very few full load hours through bidirectionally chargeable EVs means contributing to the security of supply of the energy system.
- Bidirectionally chargeable EVs lower the required installed capacities of other storage technologies, such as stationary battery storages, and thus cause a lower demand of battery capacities.
- Overall energy system costs and electricity prices decrease with an integration of bidirectionally chargeable EVs, which makes the European energy system more cost-efficient and results in lower costs for the end user of electricity.

Therefore, our results support studies from Hanemann et al. [10], Rodríguez et al. [11], Wie et al. [13], Child et al. [14] that already showed positive impacts of smart or bidirectionally chargeable EVs on local energy systems. We further show interrelated effects in the coupled European energy system, discuss different detailed modeling approaches of

bidirectionally chargeable EVs in a large-scale energy system model and finally output costoptimal penetration rates of bidirectionally chargeable EVs in the future European energy system.

However, there are some limitations of the modeling and the results that show the need for further research. The aggregated EV profile is based on German mobility data of passenger cars. More detailed mobility data of other European countries could have an influence on repercussions in the energy system. However, the basic findings should not be affected. The effect of the parametrization of the EVs, including investment costs for smart and bidirectionally chargeable EVs, should be investigated further. Possible larger EV battery capacities in the future could even increase the added values of bidirectionally chargeable EVs.

The additional load on EVs through bidirectional charging is high. 200 EFCs per year in 2035 with a lifetime of 15 years result in a total of 3,000 EFCs for the EV battery capacity. However, comparing these EV loads to current EFC lifetime values for station battery storage systems that are around 5,000 EFCs ([31], [32]) suggests that these EV loads will be sustainable in the future.

The ISAaR energy system model optimizes the European energy system in an hourly annual simulation. Each European country represents one node. The European transmission grid is only modeled by cross border transfer capacities between countries, while the distribution grid is not modeled. Therefore, the impact of bidirectionally chargeable EVs on grid utilization is not addressed by this study and should be further investigated. Due to the high simultaneity of EV charging or discharging operations with market-driven use of EVs, there is a risk of grid overloading. Müller et al. [33] already deal with the impact of smart and bidirectionally chargeable EVs on the distribution grid utilization and show decreasing and increasing grid overloads dependent on the use case and the charging strategy.

Finally, the results of the future cost-optimized energy system shown in Chapter 3.2 are from the perspective of the energy system and not from the perspective of the users. Consequently, there are no fees and taxes on electricity prices modeled that could change the dispatch of storage units, like bidirectionally chargeable EVs.

# 5. Conclusion

To evaluate the impact of electric vehicles (EVs) on the future European energy system, we analyze modeling approaches of smart and bidirectionally chargeable EVs in a large-scale multi-energy system model and model the future European energy system with and without the option of integrating these EVs. We show that aggregated EV profiles per country are sufficient to realistically model the EV behavior and the repercussions on the energy system. We find future cost-optimized penetration rates of bidirectionally chargeable EVs per country from the energy system's perspective, and thus support stakeholders in the industry in planning their future business models. Furthermore, we encourage policy makers to enable bidirectional charging of EVs through regulatory measures in order to realize the

presented positive impacts on the energy system. Finally, the paper is intended to support scientists by providing detailed descriptions of modeling approaches of bidirectionally chargeable EVs and data of EV profiles.

**Funding:** The described work is conducted within the project BCM by Forschungsgesellschaft für Energiewirtschaft mbH and funded by Bundesminsterium für Wirtschaft und Energie (BMWi) under the funding code 01MV18004C.

Acknowledgements: The authors would like to thank Alexander Murmann for his constant support regarding all issues related to the energy system model ISAaR.

### References

- [1] Guminski, Andrej et al.: eXtremOS Summary Report Modeling Kit and Scenarios for Pathways Towards a Climate Neutral Europe. Munich: FfE, 2021.
- [2] Borasio, Marcello et al.: Deep decarbonisation of regional energy systems: A novel modelling approach and its application to the Italian energy transition. In: Renewable and Sustainable Energy Reviews Volume 153, January 2022, 111730. Torino, Italy: Politecnico di Torino, 2022.
- [3] Savio, Dominic et al.: Development of multiple plug-in electric vehicle mobile charging station using bidirectional converter. In: International Journal of Power Electronics and Drive System Vol. 11, No. 2, June 2020, pp. 785~791. India: SRM Institute of science and technology, 2020.
- [4] Hinterstocker, Michael et al.: Bidirectional Charging Management Field Trial and Measurement Concept for Assessment of Novel Charging Strategies. München: Forschungsstelle für Energiewirtschaft e.V. (FfE), 2019.
- [5] Illing, Bjoern et al.: Analysis of international approaches to integrate electric vehicles into energy market. Lisbon: 2015 12th International Conference on the European Energy Market (EEM), 2015.
- [6] Kern, Timo et al.: Revenue opportunities by integrating combined vehicle-to-home and vehicle-togrid applications in smart homes. In: Applied Energy online. München: Forschungsstelle für Energiewirtschaft, 2021.
- [7] Salpakari, Jyri et al.: Flexibility of electric vehicles and space heating in net zero energy houses: an optimal control model with thermal dynamics and battery degradation. In: Applied Energy 190/2017. Espoo, Finland: New Energy Technologies Group, Department of Applied Physics, School of Science, Aalto University, 2017.
- [8] Kern, Timo et al.: Integrating Bidirectionally Chargeable Electric Vehicles into the Electricity Markets. In: Energies 2020, 13(21), 5812; Basel: Forschungsstelle für Energiewirtschaft, 2020.
- [9] Bessa, Ricardo et al.: Optimized Bidding of a EV Aggregation Agent in the Electricity Market. New Jersey: IEEE Transactions on Smart Grid, VOL. 3, NO. 1, MARCH 2012, 2012.
- [10] Hanemann, Philipp et al.: Effects of electric vehicles on the spot market price. Amsterdam: Energy Volume 162, 1 November 2018, Pages 255-266, 2018.
- [11] Rodríguez, Cristian David Bedoya et al.: Evaluating the Impact of a V2G Scheme on the Demand Curve. New Jersey, US: IEEE, 2019.
- **[12]** Huang, Zhao et al.: Multi-objective optimization strategy for distribution network considering V2Genabled electric vehicles in building integrated energy system. In: Protection and Control of Modern Power Systems volume 5, Article number: 7. Changsha, China: Hunan college of information, 2020.
- [13] Wei, Hongqian et al.: Planning integrated energy systems coupling V2G as a flexible storage. In: Energy Volume 239, Part B. Beijing, 100081, China: School of Mechanical Engineering, Beijing Institute of Technology, 2022.
- [14] Child, Michael et al.: The Impacts of High V2G Participation in a 100% Renewable Åland Energy System. In: Energies 11/2018. Lappeenranta, Turku: Lappeenranta University of Technology, Åbo Akademi, 2018.

- [15] BMVI: Mobilität in Deutschland MiD Ergebnisbericht. In: https://www.bmvi.de/SharedDocs/DE/Artikel/G/mobilitaet-in-deutschland.html. (Abruf am 2020-04-01); Bonn: Bundesministerium für Verkehr und digitale Infrastruktur, 2017.
- [16] Fattler, Steffen: Economic and Environmental Assessment of Electric Vehicle Charging Strategies (eingereicht). Dissertation. Herausgegeben durch die TU München, geprüft von Wagner, Ulrich und Wietschel, Martin: München, 2021.
- **[17]** Schmidt-Achert, Tapio et al.: Using clustering algorithms to identify representative EV mobility profiles for complex energy system models. Munich: FfE München, 2021.
- [18] Clustered and aggregated EV profiles (Germany): http://opendata.ffe.de/dataset/clustered-and-aggregated-ev-profiles-germany; München: FfE München, 2022.
- [19] Fiedler, Claudia et al.: Modelling transformation pathways for EU27+3 final energy demand using temporally and spatially resolved sector models. In: Conference Proceedings Current and Future Challenges to Energy Security; 5th AIEE Energy Symposium, virtual conference, December 2020. München: Forschungsstelle für Energiewirtschaft e.V., 2020.
- [20] Böing, F., & Regett, A. (2019). Hourly CO2 Emission Factors and Marginal Costs of Energy Carriers in Future Multi-Energy Systems. Energies, 12(12), 2260. Basel, Switzerland: MDPI AG, 2019.
- [21] Ebner, Michael et al.: Kurzstudie Elektromobilität Modellierung für die Szenarienentwicklung des Netzentwicklungsplans. München: Forschungsstelle für Energiewirtschaft, 2019.
- [22] MATLAB® Release Notes. Massachusetts, USA: MathWorks, Inc, 2021.
- [23] Gurobi Optimization Product Brochure. Oregon, USA: Gurobi, 2022.
- [24] Dossow, Patrick et al.: BDL Mehrkosten Steckbrief. München: FfE München, 2022.
- [25] Wallbox Quasar Bidirektionale Ladestation. In: https://besserladen.de/produkt/wallbox-quasarbidirektionale-ladestation/. (Abruf am 2020-08-28); Dittelbrunn: Besserladen, 2020.
- [26] Webasto Pure Wallbox Black Edition 11kW with 4.5m Cable. In: https://ev-chargershop.com/shop/webasto/webasto-pure-wallbox-black-11kw-with-4-5m-cable-from-the-ev-chargershop/. (Abruf am 2022-01-08); Hamburg: The Electric Vehicle Innovation Group, 2022.
- [27] Borlaug, Brennan et al.: Levelized cost of charging electric vehicles in the United States. In: Joule 4 (7) (2020), pp. 1470-1485. Golden, CO, USA: National Renewable Energy Laboratory, 2020.
- [28] Ekins, Paul; Kesicki, Fabian; Smith, Andrew: Marginal Abatement Cost Curves A call for caution. München: University College London Energy Institute, 2011
- [29] Schmidt-Achert, Tapio et al.: Future stationary battery cost Assumptions for eXtremOS. München: FfE, 2021.
- [30] Suškevičs, Monika et al.: Regional variation in public acceptance of wind energy development in Europe: What are the roles of planning procedures and participation?. In: Land Use Policy Volume 81, February 2019, Pages 311-323. Tartu 51006, Estonia: Institute of Agricultural and Environmental Sciences, Estonian University of Life Science, 2019.
- [31] Maximum performance for your independence. In: https://bmz-group.com/images/PDF-Downloads/Broschuere-ESS\_EN\_with\_ESSX.pdf. (Abruf am 2020-08-27); Karlstein am Main: BMZ Energy Storage Systems, 2020.
- [32] Battery Box 2.5. In: https://www.climaverd.com/sites/default/files/2019-06/ft\_sb\_byd\_b-box\_2.5-10.0\_b-plus\_2.5\_en.pdf. (Abruf am 2020-08-27); Shenzhen, China: BYD Company Limited, 2020.
- [33] Müller, Mathias et al.: Future grid load with bidirectional electric vehicles at home. Berlin: International ETG Congress 2021. VDE, 2021

# Appendix A.

					,		-					
Country	2025		2030		2035		2040		2045		2050	
	Unma	Bidi	Unma	Bidi	Unma	Bidi	Unma	Bidi	Unma	Bidi	Unma	Bidi
AT	0.74	0	1.5	0	2.08	0.12	1.63	0.96	2.13	1.12	2.97	1.12
BE	0.46	0.12	0.71	0.63	1.47	1.34	1.71	1.55	2.35	1.9	3.9	1.39
BG	0.31	0	0.52	0	0.99	0.02	1.04	0.06	0.72	0.65	0.98	0.89
СН	0.95	0.02	1.42	0.3	1.44	1.31	1.77	1.6	2.3	2.06	3.57	1.78
CZ	1.38	0	1.57	0.11	1.2	1.09	1.8	1.09	1.86	1.69	2.45	2
DE	7.59	0.2	8.45	4.64	15.15	4.89	19.46	5.88	18.16	14.02	30.14	9.57
DK	0.34	0.01	0.47	0.18	0.96	0.3	1.13	0.3	1.06	0.75	1.76	0.58
ES	3.91	1.52	3.67	3.33	5.35	4.86	6.45	5.86	8.89	5.18	11.04	6.56
FI	0.7	0	0.92	0	1.4	0	1.68	0	1.99	0	2.53	0
FR	4.58	0.69	6.72	3.27	7.93	7.21	9.13	8.29	13.44	10.89	17.94	11.37
GR	0.91	0.19	1.29	0.19	1.2	0.75	1.19	0.94	1.41	0.92	1.51	1.37
HU	0.52	0.22	0.71	0.22	0.72	0.65	0.84	0.77	1.19	0.66	1.28	1.16
IE	0.35	0.02	0.63	0.02	1.17	0.03	1.18	0.3	1.1	0.71	1.43	0.71
IT	3.52	2.41	6.39	2.41	7.29	6.62	9.91	8.4	13.17	7.36	15.38	10.96
LT	0.37	0	0.35	0.01	0.51	0.03	0.63	0.03	0.62	0.04	0.82	0.03
NL	0.97	0.32	1.22	1.08	2.36	1.08	2.96	1.08	4	1.12	5.78	0.71
NO	0.66	0	1.01	0	1.4	0	1.67	0	2	0	2.49	0
PL	5.1	0.5	5.09	0.75	4.96	2.39	4.45	3.56	5.08	3.65	6.39	4.07
PT	0.93	0.09	0.6	0.55	0.94	0.85	1.16	1.06	1.51	1.2	2.18	1.11
RO	1.13	0	1.18	0	1.69	0.03	2.03	0.12	1.27	1.16	1.65	1.5
SE	0.89	0	1.56	0	2.5	0	2.98	0	3.8	0	4.73	0
SI	0.11	0.1	0.21	0.1	0.24	0.22	0.28	0.25	0.48	0.25	0.66	0.25
SK	0.3	0	0.38	0.08	0.42	0.38	0.51	0.46	0.64	0.58	0.87	0.65
UK	4.75	0.24	8.14	0.56	13.14	1.88	15.88	2.14	20.45	3.85	26.46	3.64

Table A1.Number of EVs per country in million EVs (Unma = Unmanaged charging EVs;<br/>Bidi = Bidirectionally chargeable EVs).

Kern, T., Wendlinger, C. (2022). Added Value of Providing Transmission Grid Congestion Management via Bidirectionally Chargeable Electric Vehicles. 2022 18th International Conference on the European Energy Market (EEM), 2022. (Accepted, not yet published) ©2022 IEEE

In reference to IEEE copyrighted material which is used with permission in this thesis, the IEEE does not endorse any of Technical University of Munich's products or services. Internal or personal use of this material is permitted. If interested in reprinting/republishing IEEE copyrighted material for advertising or promotional purposes or for creating new collective works for resale or redistribution, please go to http://www.ieee.org/publications\_standards/publications/rights/rights\_link.html to learn how to obtain a License from RightsLink.

\_\_\_\_\_

### Publication 4 (Pub4): Added Value of Providing Transmission Grid Congestion Management via Bidirectionally Chargeable Electric Vehicles

#### Timo Kern<sup>a</sup>, Christian Wendlinger<sup>b</sup>

- <sup>a</sup> FfE München, Technical University of Munich (TUM), Munich, Germany, tkern@ffe.de
- <sup>b</sup> FfE München, Munich, Germany, cwendlinger@ffe.de

**Abstract:** Congestion management in the European transmission grid is currently mostly provided by conventional power plants. As electricity generation in the future will be increasingly characterized by volatile renewable energies, new technologies need to be integrated into congestion management. This paper develops a methodology for modeling storages in congestion management and investigates the potential for bidirectional electric vehicles (EVs) in 2030. A direct current load flow via Power Transfer Distribution Factors (PTDF) is used to model the load flow in the transmission grid. Since the modeling involves great complexity due to time-coupling constraints of storages, different modeling approaches are investigated. The results show that bidirectional EVs can take over a significant part of the congestion management due to their decentralized distribution. A total of 26 TWh of positive redispatch of conventional power plants in 2030 can be replaced by the optimized use of bidirectional EVs.

**Index Terms:** congestion management; bidirectionally chargeable electric vehicles; European energy system; redispatch; vehicle-to-grid.

# I. Introduction

To achieve the European climate protection targets, the future European energy system will be strongly characterized by volatile renewable energies [1]. Since wind turbines and solar plants are placed on revenue-maximizing locations and not near the highest electricity demand like power plants usually are, the stress on the transmission grid increases. Consequently, the volumes of congestion management will most likely increase in the future. In Germany, congestion management initially included the curtailment of renewable energies and separately the provision of redispatch via conventional power plants. Redispatch refers to a short-term adjustment of a scheduled generation or consumption of an asset to lower congestions in the transmission grid. In 2019, NABEG 2.0 integrated combined heat and power plants and the curtailment of renewable energies into a uniform regulatory regime, so that the most effective and cost-efficient plants can be called upon for redispatch [2]. The 'Redispatch 2.0' introduced in 2021 standardizes this even more and also integrates smaller, remotely controllable assets and electricity storages of a power 100 kW or more [3] into congestion management. Bidirectionally chargeable electric vehicles (bidirectional EVs) represent an alternative option to provide redispatch by adjusting their scheduled charging and discharging power and could thus be integrated into the redispatch process as well. In Germany, the goal has been set to integrate 15 million electric vehicles by 2030 [4]. Increasingly, these electric vehicles will be equipped with bidirectional charging management technology [5]. Consequently, bidirectional electric vehicles can potentially have a large impact on the provision of ancillary services. Therefore, this paper analyzes the impact of bidirectional EVs integrated in the European redispatch process on the dispatch and thus emissions of conventional power plants.

Storages, and thus bidirectional EVs, reduce inter-temporal price volatility leading to smoothed electricity prices, while inter-regional price volatility may increase [6]. Regarding the provision of redispatch services, there are numerous papers that deal with stationary and mobile storages. Meyer-Huebner et al. show that preventive generator redispatch can be reduced by reserving positive or negative storage capacity by modeling a 5-bus test system with 192 timesteps [7]. However, they do not include bidirectional EVs and limit their investigations on a small-scale test system. Xiong et al. develop a two-stage model incorporating the day-head spot market and subsequent redispatch and find out that Power-to-Gas (PtG) assets providing redispatch services reduce the curtailment of renewable energies by 12% [8]. They model PtG assets like storages with time-depending constraints, but do not include mobile storages. Eickmann et al. also develop a two-stage model with a day-ahead market simulation followed by a redispatch evaluation [9]. They point out that pumped hydroelectric energy storages participating in the redispatch process reduce congestion management energy and costs. Müller et al. model time variant redispatch provided by virtual power plants that include biogas-combined heat and power plants, controllable loads and storage systems [10]. However, they do not include mobile storages and limit their optimization on Germany for 8h in 15 min time steps. Gutermuth et al. analyze the benefit of grid operator owned stationary storages by an hourly simulation

of the German transmission grid in 2030 showing a simplified iterative modeling approach to reduce complexity [11]. All aforementioned studies focus on stationary storages and limit the optimization to small-scale test systems, to only few timesteps or to simplified modeling. Since we want to evaluate the benefits of bidirectional EVs providing redispatch services in the large-scale European energy system, we set up a modeling of mobile storages, and analyze the complexity of the optimization with different parameterizations. There are also few studies that investigate bidirectional EVs integrated in the redispatch process. Staudt et al. model bidirectional EVs providing redispatch services in Germany but simplify the EV behavior and redispatch provision by a heuristic approach [12]. Thormann et al. model bidirectional EVs that provide redispatch focusing on the medium-voltage grid but simplify modeling by only comparing charging and discharging patterns with redispatch system needs [13]. We model the optimal operation of bidirectional EVs by realistic profiles and user behavior to show the influence of bidirectional EVs on the energy system.

### II. Modeling

# A. Two-stage market and transmission grid optimization in energy system model ISAaR

Congestion management in the energy system model ISAaR (Integrated Simulation Model for Unit Dispatch and Expansion with Regionalization) is modeled in a two-step approach. First, a multi-energy market optimization run is conducted. One market area is represented by one node. Energy transfers between market areas are allowed up to maximum net transfer capacities (NTCs). In the following step, the resulting dispatch of generation, load and storage instances is taken as a base for the transmission grid optimization run, in which the costs of congestion management (curtailment of renewable energies and provision of redispatch services) are minimized. Here, a node represents a transmission grid node. For a more detailed description of the multi-energy market optimization, we refer to Kigle et al. [14]. The transmission grid optimization run is previously described in [15] and [16]. In this publication, we extend the transmission grid optimization by integrating mobile and stationary storages with a focus on bidirectional EVs.

# B. Time-dependent modeling of bidirectional EVs in multi-energy system model ISAaR

Bidirectional EVs in the multi-energy system model ISAaR are modeled according to [17]. The most important constraint of a bidirectional EV is its time-coupled energy conservation as shown in (1). The state of the battery energy E at a time t is equal to the energy of the previous time step added by the charged energy  $P_i(t) \cdot \eta_i \cdot \Delta t$  and subtracted by the discharged energy  $\frac{P_o(t) \cdot \Delta t}{\eta_o}$  as well as the driving consumption  $E_{con}$ , where  $\eta_i$  and  $\eta_o$  refer to the fixed charging and discharging efficiency.

$$E(t) = E(t-1) + P_i(t) \cdot \eta_i \cdot \Delta t - \frac{P_o(t) \cdot \Delta t}{\eta_o} - E_{con}(t)$$
(1)

The charge and discharge power of the EV supply equipment (EVSE)  $P_i$  and  $P_o$  are constrained by a maximum power  $P_{i,max}$  and  $P_{o,max}$ . The battery energy is further constrained by a minimum state of charge (SoC) at a connection of the EV to the EVSE and a minimum SoC at the departure of the EV. These constraints are included in the profile of the minimum energy  $E_{min}$ , which is applied in addition to the constraining maximum energy  $E_{max}$  in (2).

$$E_{min} \le E(t) \le E_{max} \tag{2}$$

#### C. Modeling of the transmission grid optimization run

The transmission grid is modeled at a regional resolution, with the highest level of detail for Germany and higher aggregation of transmission grid nodes for more distant countries. Figure 1 shows the mapping of the transmission network. For example, in Spain and Portugal, all nodes are aggregated into one, Great Britain has 10 nodes and Germany is mapped with 460 nodes. In total, this results in 1,580 nodes, which leads to a high complexity of the optimization problem. The transmission grid data is provided by the static grid models described in [16] and extended by the German [18] and European [19] grid development plants. Highly aggregated parallel transmission lines are modeled as direct current (DC) lines, since the reactance of the lines would otherwise become very small [20], resulting in an unrealistic load flow.





Böing [15] describes the applied, linearized approximation of the non-linear alternating current (AC) load flow by applying a DC load flow based on PTDF (Power Transfer Distribution Factors). Assuming that there are no voltage drops (voltage amplitude is equal for all nodes), reactive power is neglected, line losses are neglected, and the voltage angle

differences are small, the simplification of a DC load flow is justified [21]. Since the reactive power consumption of the AC lines increases quadratically, modeling the AC load flow via a DC load flow is no longer permissible for high loads [22]. In addition, since line outages must be considered, the AC lines are limited to a maximum load of 70%, which represents the single outage contingency operational criterion (n-1) well [23]. The load flow of AC lines  $P_{line,AC}$  is determined by the injections and withdrawals of grid nodes  $P_{node}$  multiplied by the PTDF matrix according to (3). The PTDF matrix is formed by the incidence matrix A describing the grid topology and the diagonal matrix with line susceptances B according to (4). A detailed derivation of the equations can be found for example in [24], and [21].

$$P_{line,AC} = PTDF \cdot P_{node} \tag{3}$$

$$PTDF = (B \cdot A) \cdot (A^T \cdot B \cdot A)^{-1}$$
(4)

The converters at the end and at the beginning of a DC line can fully control its power flow [24]. Therefore, line flow from a node x to a node y can be modeled via a transport model according to (5):

$$P_{line,DC} = P_{x \to y} \tag{5}$$

# D. Modeling of congestion management via power plants and renewable energies

To model congestion management, a few constraints are introduced. Equation (6) shows the power balance based on electrical output of the market run, the optimized output, and the redispatch provisions of conventional power plants. The result of the marked optimization run  $P_{M,pp}$  for each power plant pp is set as a fixed bound. The electrical output ( $P_{o,pp}$ ) is optimized under consideration of positive ( $P_{p,pp}$ ) and negative ( $P_{n,pp}$ ) redispatch services. The lower and upper bounds of the redispatch services are shown in (7) and (8).

$$P_{o,pp}(t) - P_{p,pp}(t) + P_{n,pp}(t) = P_{M,pp}(t)$$
(6)

$$0 \le P_{n,pp}(t) \le P_{M,pp}(t) \tag{7}$$

$$0 \le P_{p,pp}(t) \le P_{o,max,pp}(t) - P_{M,pp}(t)$$
(8)

Variable renewable energy sources (vRES) are modeled without variable operational cost. Therefore, all vRES operate at maximum output in the market optimization run and no positive redispatch is possible through vRES. Thus, Equation (6) is shortened to (9), where  $P_{c,vRES}$  represents the curtailment of vRES,  $P_{M,vRES}$  the output of the market run and  $P_{o,vRES}$  the electrical output of the transmission grid run. Equation (10) restricts the curtailment of vRES.

$$P_{o,vRES}(t) + P_{c,vRES}(t) = P_{M,vRES}(t)$$
(9)

$$0 \le P_{c,vRES}(t) \le P_{M,vRES}(t) \tag{10}$$

# E. Modeling of congestion management via stationary and mobile storages

The modeling of congestion management with electricity storage systems is challenging due to their time-coupling constraints [15]. Modeled storage systems in ISAaR include pumped hydroelectric energy storages, stationary battery storages and bidirectional EVs.

The transmission grid optimization run is conducted parallel with several subproblems to save time and to keep the problem solvable. The whole optimization period of 8760 h for an annual simulation is therefore divided into equal time slices. At the first timestep of a time slice, the preceding storage level is unknown because there is no link between the parallel optimized time slices. Therefore, the storage level at the end of a time slice is fixed to the result of the market optimization run and the same value is handed over to the next time slice to maintain a coupling (of the time slices) as shown in Figure 2.



### **Figure 2:** Schematic modeling of the restricted storage fill level at the start and the end of a time slice

By changing the length of the time slice, the flexibility of the storage operation while providing congestion management is affected. Due to the fixed storage level at the end of each time slice, every change of operation from the market schedule is limited. Therefore, shorter time slices lead to a bigger limitation of flexibility of the storages. Thus, long time slices are recommended to avoid unrealistic restrictions on modeling.

To calculate the redispatch services of storage systems st, (6) is extended by the input power  $P_{i,st}$ , resulting in (11). The power balance of storage systems providing redispatch services differs from normal power plants, because of the bidirectional power flow. A reduction in charging power as well as an increase in discharging power causes positive redispatch. On the other hand, a reduction in discharging power as well as an increase in charging power causes negative redispatch. The upper bounds of the redispatch variables are shown in (12) and (13): Appendix A visualizes the flexibility limits of congestion management provision by storages.

$$-P_{i,st}(t) + P_{o,st}(t) - P_{p,st}(t) + P_{n,st}(t) = P_{M,st}(t)$$
(11)

$$0 \le P_{n,st}(t) \le P_{i,max,st} + P_{M,st}(t) \tag{12}$$

$$0 \le P_{p,st}(t) \le P_{o,max,st} - P_{M,st}(t) \tag{13}$$

#### F. Cost optimization of congestion management

The transmission grid run minimizes the total system costs *C* under consideration of congestion management. These costs *C* shown in (14) result from the specific costs  $f_{t,n,i}$  and the utilization of the optimization variables  $x_{t,n,i}$  for all time steps *t*, nodes *n* and instances *i*. Variable  $x_{t,n,i}$  include generation, consumption and import of all modeled energy carriers. Investment decisions in instances of any elements of the energy system model are fixed for the transmission grid run and thus not included in the system costs.

$$C = \min \sum_{t \in T} \sum_{n \in N} \sum_{i \in I_n} (f_{t,n,i} \cdot x_{t,n,i})$$
(14)

Costs of congestion management are also included in (14). Appendix B visualizes the applied order of instances providing congestion management. Costs for positive redispatch and revenues for negative redispatch services of power plants are set dependent on the specific operational costs of every individual plant. The operational costs get a surcharge to avoid unnecessary redispatch provisions. The revenues of curtailing vRES are set to zero. Due to this, it is ensured, that the power plant with the highest operational costs reduces its output first, while vRES is curtailed last. Costs of storages for positive redispatch are slightly lower.

### **III.** Results

#### A. Market run results

Our evaluation scenario for the year 2030 is based on the 'SolidEU' scenario of the eXtremOS research project [1]. The scenario is characterized by a joint European decarbonization strategy reaching a reduction in greenhouse gas (GHG) emissions of 55% by 2030 with respect to 1990 levels. For an updated modeling of the German climate targets, the German targeted high installed capacities of vRES in 2030 according to the coalition agreement are taken into account [25]. The scenario does not exogenously quantify the share of smart and bidirectional EVs. These shares are optimized endogenously as described in [17]. The optimization of the energy markets in 2030 leads to the generation and storage capacities for the EU in 2030 shown in Table 1 compared to the capacities in 2020. The installed generation capacities show a sharp increase in vRES capacities, while many coal-fired and nuclear power plants are shut down. Further, Table 1 shows the integrated mobile and stationary storage capacities. Mobile storages refer to bidirectional EVs and stationary storages include large-scale battery system and pumped hydroelectric energy storages participating in the electricity day-ahead market. Storage capacities increase significantly by 2030 to balance volatile generation from vRES. Installed power plant, renewable energies, and storage capacities per country and per transmission grid node are shown in Appendix C.

	Wind	PV	Nuclear	Coal	Gas	Mobile storage	Stat. storage
EU 2020	200	140	120	130	250	0	530
	GW	GW	GW	GW	GW	GWh	GWh
Scenario EU	520	910	90	50	260	790	710
2030	GW	GW	GW	GW	GW	GWh	GWh

Table 1:Generation and storage capacities resulting from the market run for the EU in<br/>2030 compared to 2020

#### B. Computation time analysis of transmission grid run

The computation time of the grid optimization run in the model ISAaR depends on many different parameters. The optimization problem of the computation time analysis is limited to a horizon of 672 hours and is based on the scenario described in 0. The varied parameters include the length of the time slices and the number of threads for each parallel run. In an additional evaluation case, 'no ST', the influence of storage systems is investigated by excluding storage systems from the congestion management. Table 2 shows the work units, optimization time and RAM usage for the considered evaluation scenarios. The hardware of the system and an explanation of the evaluation values is described in Appendix D.

The results in Table 2 indicate the following: As the size of the time slice increases, there is generally an increase in complexity accompanied by an increase in optimization time and in RAM usage per run. Excluding storages from the congestion management run slightly lowers complexity. An increased number of threads per run with fewer parallel runs decreases the work units for the time slice of 168 hours. Time slices of 336 and 672 hours are not computable on the specified hardware (Appendix D).

No. of	Pa	arameter va	riation:		optimization time	RAM usage per run (GB)	
analysis	time slice	parallel runs	threads per run	work units	(h)		
1	6	4	6	31,430	2	14	
2	48	4	6	154,010	6.75	40	
3	168	4	6	191,190	7.75	110	
4	168*	4	6	178,050	6.75	110	
5	168	2	12	114,600	7.75	110	
6	336	2	12	_**	-	265	
7	62	1	24	_***	-	-	

Table 2:Computation time analysis of different parameter sets and their impact on the<br/>optimization time

\* No storages

\*\* No results, since the optimization cannot be performed parallel due to a lack of random-access memory (RAM)

\*\*\* No results, since the model is too big to solve

Appendix D also discusses the differences in the results of the first 3 investigation scenarios. Since the flexibility of bidirectional EVs is very much limited for smaller time slices and the computation time for the 168h time slice is not significantly higher, we use a time slice length of 168 hours with two parallel runs for the simulations in the following.

# C. Added value of bidirectionally chargeable EVs providing redispatch services

In the following, the effects of bidirectional EVs on the congestion management process are shown based on two different transmission grid runs. In scenario Ref, we perform an optimization of the provision of congestion management services without bidirectional EVs. In the BCM scenario, bidirectional EVs (but no other storage units) are included in the congestion management process. Figure 3 shows the volumes of congestion management measures per country and for different technologies for the BCM run, Figure E1 (Appendix E) shows in comparison the congestion management of the run Ref. Displayed are those nine countries with the highest volumes of congestion management. In scenario Ref, there is a total volume of positive congestion management of 600 TWh compared to a total electricity production of just under 5,000 TWh. This indicates an overall congested grid situation due to the strong electrification and expansion of vRES. The results show that bidirectional EVs provide congestion management in many countries. Overall, 26 TWh or 4% less positive redispatch is provided by conventional power plants in the BCM scenario. Positive redispatch of backup capacities ('Slack') is even reduced by 13.5 TWh (16%). The curtailment of vRES decreases by 23 TWh or 5% and the provision of negative redispatch from conventional power plants decreases by 9 TWh in the BCM scenario, mainly by nuclear power plants. Thus, a total of 17 TWh less electricity is generated from conventional power plants leading to a reduction of emissions of 12 mio. t CO<sub>2</sub>.



**Figure 3:** Congestion management per country split up in different technologies for the BCM optimization run

The maps in Figures F1 and F2 (Appendix) show the location of the negative and positive congestion management, respectively. Additionally, the medium line usage of the modeled AC and DC lines is shown. The share of negative redispatch compared to the amount of curtailment is quite small. It is noticeable that curtailed vRES are mainly located in coastal areas in the north of Europe beside the aggregated grid nodes in the south of Italy. Positive
redispatch must be provided on the other side of the congestions, especially in areas with high demand.

The maps in Figure 4 and Appendix G show the positive and negative congestion management per grid node split up into different technologies for the *Ref* and the *BCM* scenario. The negative congestion management in the coastal areas of Europe is mainly caused by wind turbines. Bidirectional EVs provide more than the half of the negative redispatch, 120 TWh of 230 TWh in total. The positive redispatch of bidirectional EVs is slightly lower due to losses of additional charging and discharging. As bidirectional electric vehicles are decentralized distributed, they can contribute greatly to both negative and positive congestion management.



Figure 4: Positive congestion management per grid node split up in different technologies, run BCM

### IV. Discussion

Our results indicate that bidirectional EVs can play a large part in congestion management in the future energy system, leading to added value on many levels: Due to the decentralized distribution of bidirectional EVs, they can provide congestion management at numerous nodes in the transmission grid. This reduces the positive redispatch of conventional power plants by 4%. The use of backup power plants is even reduced by 16% leading to saved costs and  $CO_2$  emissions.

Our modeling of congestion management by bidirectional EVs has several limitations that must be considered. The length of the integrated optimized time slice was set to one week, in whose last time step the storage level is fixed. This limits the flexibility of bidirectional EVs. However, since perfect foresight does not exist in reality either, this is an acceptable limitation. Regarding the load flow calculation and the mapping of the electricity grid, several simplifications have been made. Voltage levels below the extra-high voltage level are not modeled and loads and generations are aggregated to the transmission grid nodes. The calculated load flow is a linearization of the AC load flow, resulting in deviations from the real load flow. However, since the grid optimization run already leads to a very high complexity due to the time coupling of the bidirectional EVs, these simplifications are

necessary to ensure computability. Furthermore, the correct regionalization of elements is associated with uncertainties. For example, the negative redispatch of a few nuclear power plants in France revealed their incorrect regionalization. Due to a too-low grid connection, their full power could not be fed into the system. The future grid connection points of wind offshore plants have also not yet been determined in some cases, so that problems arose here in feeding their energy into the system.

Overall, due to high congestion management, the modeled transmission grid does not appear to be designed for the installed capacities of wind and PV energy resulting from the market run, although planned grid expansion projects have been considered. Resulting congestion management volumes in Germany around 330 TWh are more than 10 times as much as real volumes in 2020 of around 23 TWh [26]. One possible explanation is the lack of modeled overhead line monitoring and high-temperature transmission lines, which underestimates the thermal capacities of the transmission grid [27]. Nonetheless, the 2030 grid simulations reveal major challenges in integrating renewables into the system. Here, bidirectional EVs can demonstrably support the provision of congestion management to reduce the curtailment of renewable energies and thus ultimately also bring about a reduction in CO<sub>2</sub> emissions.

## V. Conclusion

Due to the increasingly reduced capacities of conventional power plants, a future alternative provision of congestion management is of great importance. In this paper, we developed an approach that allows modeling of European congestion management with storage despite the high complexity of time-coupled conditions. Our results show that bidirectional electric vehicles (EVs) can take over a significant share of congestion management in the future energy system and thus save costs and CO<sub>2</sub> emissions. This gives stakeholders in the energy industry and politics the task of creating regulatory framework conditions to enable an integration of bidirectional (EVs) into the process of congestion management.

**Funding:** The described work is conducted within the project BCM by Forschungsgesellschaft für Energiewirtschaft mbH and funded by Bundesministerium für Wirtschaft und Klimaschutz (BMWK) under the funding code 01MV18004C.

## References

- [1] Guminski, Andrej et al.: eXtremOS Summary Report Modeling Kit and Scenarios for Pathways Towards a Climate Neutral Europe. Munich: FfE, 2021.
- [2] von Hammerstein, Christian et al.: Neue Redispatchregelung im NABEG 2.0. In: https://raue.com/aktuell/branchen/energie-rohstoffe-und-klimaschutz/erneuerbare-energien/neueredispatchregelung-im-nabeg-2-0/. (accessed on 2019-10-27); Berlin, DE: RAUE, 2019.

- [3] Fattler, Steffen: Economic and Environmental Assessment of Electric Vehicle Charging Strategies. Dissertation. Herausgegeben durch die TU München, geprüft von Wagner, Ulrich und Wietschel, Martin: München, 2021.
- [4] Germany's current climate action status. Berlin: Bundesministerium für Wirtschaft und Klimaschutz (BMWK), 2022.
- [5] Convenient, networked and sustainable: new solutions for charging electric Volkswagen models. Wolfsburg: Volkswagen, 2021.
- [6] Weibelzahl, Martin et al.: On the effects of storage facilities on optimal zonal pricing in electricity markets. In: Energy Policy Volume 113, February 2018, Pages 778-794. Aachen: RWTH Aachen University, Discrete Optimization & Advanced Analytics, 2018.
- [7] Meyer-Huebner, Nico et al.: N-1-Secure Optimal Generator Redispatch in Hybrid AC-DC Grids with Energy Storage. In: Innovative Smart Grid Technologies Conference Europe (ISGT-Europe) 2018 IEEE PES. Karlsruhe: Nico Meyer-Huebner, Michael Suriyah, Thomas Leibfried Institute of Electrical Energy Systems and High Voltage Technology, KIT, 2018.
- [8] Xiong, Bobby et al.: Spatial flexibility in redispatch: Supporting low carbon energy systems with Power-to-Gas. In: Applied Energy Volume 283, 1 February 2021, 116201. Trondheim: Department of Industrial Economics and Technology Management, 2021.
- [9] Eickmann, Jonas et al.: Optimizing Storages for Transmission System Operation. In: Energy Procedia 46 (2014) 13 – 21. Aachen: Institute of Power Systems and Power Economics (IAEW), RWTH Aachen, 2014.
- [10] Müller, Benjamin et al.: Coordinated Dynamic Use of Dispersed Flexibility to Maximize the Timevariant Aggregated Potential for Redispatch. In: IFAC-PapersOnLine Volume 53, Issue 2, 2020, Pages 12195-12200. Stuttgart: Institute for Combustion and Power Plant Technology (IFK), 2020.
- [11] Gutermuth, Georg et al.: Network operator owned storages as an option for congestion management. In: Innovative Smart Grid Technologies Europe (ISGT-Europe) 2020 IEEE PES. Ladenburg: ABB AG Corporate Research, 2020.
- **[12]** Staudt, Philipp et al.: Using vehicle-to-grid concepts to balance redispatch needs. In: Proceedings of the Ninth International Conference on Future Energy Systems e-Energy '18. Karlsruhe: Karlsruhe Institute of Technology, 2018.
- [13] Thormann, Bernd et al.: Evaluating the potential of future e-mobility use cases for providing grid ancilliary services. In: IEEE Xplore CIRED 2021 Conference. Leoben: Montanuniversitaet Leoben, Chair of Energy Network Technology, 2021.
- [14] Kigle, Stephan et al.: Greenhouse Gas Abatement in EUROPE—A Scenario-Based, Bottom-Up Analysis Showing the Effect of Deep Emission Mitigation on the European Energy System. In: Energies 2022, 15, 1334. Basel: MDPI, 2022. https://doi.org/10.3390/en15041334.
- **[15]** Felix, Böing: Cross-sector assessment of CO2 abatement measures and their impact on the transmission grid. Dissertation. Herausgegeben durch die TU München, geprüft von Wagner, Ulrich: München, 2020.
- [16] Böing, Felix; Murmann, Alexander; Pellinger, Dr. Christoph; Bruckmeier, Andreas; Kern, Timo; Mongin, Thibaud: Assessment of grid optimising measures for the German transmission grid using open source grid data in: Journal of Physics: Conference Series. Oldenburg: IOP Publishing, 2018
- [17] Kern, Timo et al.: Modeling and Evaluating Bidirectionally Chargeable Electric Vehicles in the Future European Energy System. In: Energy Reports ICACER 2022 conference proceedings. Amsterdam: FfE München, 2022.
- [18] Anhang zum Netzentwicklungsplan Strom 2030, Version 2019, zweiter Entwurf. Berlin, Dortmund, Bayreuth, Stuttgart: 50Hertz Transmission GmbH, 2020.
- [19] TYNDP 2020 Projects Sheets. Brussels: entso-e, 2020.
- [20] Andersson, Göran: Power System Analysis Power Flow Analysis Fault Analysis Power System Dynamics and Stability. Zürich: Eidgenössisch Technische Hochschule Zürich (ETH), 2011
- [21] Hagspiel, Simon et al.: Cost-optimal power system extension under flow-based market coupling in: science direct energy. Amsterdam: Elsevier, 2014
- [22] Brown, Tom; Schierhorn, Peter-Philipp; Tröster, Eckehard; Ackermann, Thomas: Optimising the European transmission system for 77% renewable electricity by 2030 in: IET Renewable Power Generation, 2016, Vol. 10, Iss. 1. Aalborg: IET Renewable Power Generation, 2015

- [23] Feix, Olivier; Obermann, Ruth; Strecker, Marius; Brötel, Angela: Netzentwicklungsplan Strom 2012 2. überarbeiteter Entwurf der Übertragungsnetzbetreiber. Bayreuth: Netzentwicklungsplan, 2012
- [24] Van Den Bergh, Kenneth; Delarue, Erik; D'haeseleer, William: DC power flow in unit commitment models in: TME Working Paper Energy and Environment. Belgium: KU Leuven Energy Institute, 2014
- [25] Koalitionsvertrag zwischen SPD, FDP und Grünen- 20. Legislaturperiode Mehr Fortschritt wagen -Bündnis für Freiheit, Gerechtigkeit und Nachhaltigkeit. Berlin: Koalitionsvertrag zwischen SPD, FDP und Grünen, 2021.
- [26] Bericht zum Redispatch nach Artikel 13 Verordnung (EU) 2019/943. Bonn: Bundesnetzagentur für Elektrizität, Gas, Telekommunikation, Post und Eisenbahnen, 2022.
- [27] Factsheet Freileitungsmonitoring. Bayreuth: TenneT, 2018.
- [28] Gurobi Optimization Product Brochure. Oregon, USA: Gurobi, 2022.

### Appendix

#### A. Flexibility limits of congestion management provision by storages





#### B. Applied order of congestion management provision



Figure B1: Applied order of congestion management provision

#### C. Installed capacities



Figure C1: Installed capacities of different technologies per grid node





#### D. Computation time analysis

The used computation server has the following specifications:

- operating system: Linux
- Solver: Gurobi Optimization [28]
- 2x AMD EPYC 7F52 16 Core
- 1008 GB RAM

The 'work unit' value in Table 2 is a unit that measures the work spent on an optimization problem. It is a deterministic unit thus the same amount of work is needed when optimizing the same problem with the same set of parameters and the same hardware. The optimization time is not deterministic, meaning it will take probably a different time to solve the same problem with the same set of parameters and the same hardware. But the optimization time is the point of interest for the modelers, since this time indicates how long it takes to obtain results. We do this analysis without any parallel running process on

the server, so all capacities can be used for solving the problem trying to reduce other influences on the optimization time. The shown optimization time displays the time from the start of an optimization till its end and is not the totaled time of every (partly) parallel conducted run.

The following Figure D1 shows the storage fill level of runs 1 to 3 in Table 2 in comparison with the market run, for one exemplary week at an exemplary grid node. It can be seen that the light blue line always has to reach the market result after every sixth hour and thus is not as flexible as the dark blue and red line, respectively. It is noticeable that the second and third transmission grid runs (time slices: 48 h, 168 h) are quite similar during the midhours of a 48 h time slice but differ in the hours around the end of one slice and the start of another (marked by dashed lines). This shows that long time slices enable more flexibility to the system to reach a cost optimal solution.



**Figure D1:** Comparison of storage fill levels from different parametrized optimization runs for an exemplary week at an exemplary grid node

# E. Congestion management per country split up in different technologies for the Ref run



Figure E1: Congestion management per country split up in different technologies for the Ref run

# F. Congestion management and mean line usage for the BCM optimization run



Figure F1: Negative congestion management and mean line usage for the BCM optimization run



Figure F2:Positive congestion management and mean line usage for the BCM optimization<br/>run

# G. Congestion management per type for the BCM and Ref optimization run



Figure G1: Negative congestion management per grid node split up in different technologies, run BCM



Figure G2: Negative congestion management per grid node split up in different technologies, run Ref



Figure G3: Positive congestion management per grid node split up in different technologies, run Ref

### Danksagung

Diese Dissertation wurde erst durch die kontinuierliche Unterstützung vieler Personen ermöglicht, die Teil meines beruflichen und privaten Weges sind.

An erster Stelle möchte ich mich bei meinem Doktorvater, Prof. Ulrich Wagner, bedanken für die Betreuung meiner Promotion und die Begleitung meines wissenschaftlichen Wegs. Durch Ihre kompetente, ruhige Art in den zahlreichen Gesprächen haben Sie mich sehr unterstützt und mich hinsichtlich der Relevanz meiner Arbeit bestärkt.

Mein Dank gilt außerdem Prof. Reinhard Haas für Ihr Interesse an den Themen meiner Promotion und den angenehmen Austausch im Vorfeld der Begutachtung sowie das Übernehmen der Zweitprüferschaft.

Weiterhin möchte ich insbesondere meinem Mentor Serafin von Roon danken, der mir zum einen ein unendliches Vertrauen entgegenbringt und meine Promotion mit kritischen Diskussionen unterstützt hat.

Mein besonderer Dank gilt zudem Prof. Wolfgang Mauch und den hier nicht namentlich genannten FfE-Kolleg:innen zur Schaffung eines tollen teamorientierten und fördernden Arbeitsumfelds, das es zudem erlaubt, kreativ, zielstrebig und eigenverantwortlich zu arbeiten.

Ohne die Zusammenarbeit mit dem großartigen FfE-Team wäre die Promotion nicht möglich gewesen. In besonderem Maße möchte ich hier Patrick Dossow nennen, der mir stets als fachlicher und privater Ansprechpartner zur Seite stand, mit dem das Arbeiten und Paper schreiben mehr Spaß als Arbeit bedeutet. Weiterhin konnte ich jederzeit auf den tiefen Zusammenhalt im gesamten ISAaR-Team zählen. Alex Murmann, Christian Wendlinger, Kirstin Ganz und Stephan Kigle: Mit euch Unlösbarkeiten zu beseitigen, ist eine Freude! Und auch meinen langjährigen Weggefährt:innen Andrej Guminski, Anika Neitz-Regett, Claudia Fiedler, Christoph Pellinger und Tobias Hübner möchte ich danken. Ob auf dem Beach-Court als Giraffen-Team oder am Schreibtisch: Euer Rat bedeutet mir viel!

Nicht zuletzt geht mein Dank an die zahlreichen Studierenden, die mit ihrer Arbeit, ihrem Einsatz und ihren spannenden Impulsen diese Promotion ermöglicht haben.

Abschließend möchte ich bei meiner Familie bedanken, die mich immer und überall in jeglichen Belangen unterstützt. Gleiches gilt für meine langjährigen Freunde (insbesondere DPÄ), auf die immer Verlass ist. Und dir, Julia, dafür, dass du mich über die spannenden Berge und auch teils holprigen Täler meiner Promotion begleitet hast, immer zu mir stehst, den Rücken freihältst und mit einfach Kraft gibst.